Eighth International Scientific Conference

ERAZ 2022



www.eraz-conference.com

SELECTED PAPERS













 8^{TH} INTERNATIONAL SCIENTIFIC CONFERENCE ERAZ 2022

KNOWLEDGE BASED SUSTAINABLE DEVELOPMENT Selected Papers

Online-Virtual (Prague, Czech Republic) May 26, 2022

International Scientific Conference ERAZ – Knowledge Based Sustainable Development

ISSN 2683-5568

Selected Papers (part of ERAZ conference collection)

Editor:

Vuk Bevanda 匝

PhD, Associate Professor, Faculty of Social Sciences, Belgrade, Serbia

Associate Editors:

Paula Heliodoro 🕩 PhD, Associate Professor, Polytechnic Institute of Setubal, Setubal, Portugal Marko Tomljanovic 回 PhD, Assistant Professor, Department of International Economics, University of Rijeka, Faculty of Economics and Business, Rijeka, Croatia

Organizational Committee:

Anton Vorina 匝 PhD. Associate Professor: Nikolina Vrceli 🕩 PhD candidate; Nevena Bevanda PhD student, Udekom Balkan, Serbia; Ivana Mirčević BSc, Udekom Balkan, Serbia; Uroš Mirčević Ing., Udekom Balkan, Serbia; Goran Stevanović BSc, Udekom Balkan, Serbia

Published by:

Association of Economists and Managers of the Balkans – UdEkoM Balkan Ustanicka 179/2 St. 11000 Belgrade, Serbia office@udekom.org.rs +381 62 8125 779

Conference partners:

Faculty of Economics and Business, Mediterranean University, Montenegro; University of National and World Economy - Sofia, Bulgaria; Faculty of Commercial and Business Studies - Celje, Slovenia; AMBIS University, Prague, Czech Republic Faculty of Applied Management, Economics and Finance – Belgrade, Serbia

Cover image by: Markéta Machová from Pixabay.

Print circulation: 100 copies

Printed by: SKRIPTA International, Belgrade, Serbia

Belgrade, 2023

ISBN 978-86-80194-61-5

ISSN 2683-5568

DOI: https://doi.org/10.31410/ERAZ.S.P.2022

Disclaimer: The author(s) of each paper appearing in this publication is/are solely responsible for the content thereof; the findings, interpretations and conclusions expressed in the papers are those of the authors and do not reflect the view of the editor, reviewers, scientific committee members, the publisher, conference partners or anyone else involved in creating, producing or delivering this publication.

E OS This work is licensed under a Creative Commons Attribution-NonCommercial 4.0 International License

CIP - Katalogizacija u publikaciji Narodna biblioteka Srbije, Beograd 330(082) 502.131.1(082)

INTERNATIONAL Scientific Conference Knowledge Based Sustainable Development ERAZ (8; 2022)

Selected Papers / The 8th International Scientific Conference ERAZ 2022, Knowledge Based Sustainable Development, Online-virtual (Prague, Czech Republic) May 26, 2022 ; [editor Vuk Bevanda]. - Belgrade : Association of Economists and Managers of the Balkans, 2023 (Belgrade : Skripta international). - XIV, 119 str. : ilustr. ; 29 cm. - (ERAZ Conference collection, ISSN 2683-5568)

Tiraž 100. - Bibliografija uz svaki rad.

ISBN 978-86-80194-61-5

a) Ekonomija -- Zbornici 6) Održivi razvoj -- Zbornici COBISS.SR-ID 115394313

Scientific (Program) Committee of the ERAZ conference:

- Fatmir Memai 匝 1. PhD, Full-Time Professor, Department of Statistics and Applied Informatics, Deputy Dean at Faculty of Economics, University of Tirana, Albania 2. Irma Shyle 🛄 PhD, Associate Professor, Polytechnic University of Tirana, Department of Production and Management, Tirana, Albania 3. Almir Alihodžić PhD, Associate Professor, University of Zenica, Faculty of Economics Zenica, Bosnia and Herzegovina 4. Tanja Schuschnig PhD, Associate Professor, Vice-Rector of Studies for Further Education, Department of Financial Management, Faculty of Management and Economics, University of Klagenfurt, Austria **Gudrun Fritz-Schmied** PhD, Associate Professor, Department of Financial Management, Faculty of Management and Economics, University of Klagenfurt, Austria Christian Tanushev 回 6. PhD, Full-Time Professor, University of National and World Economy, Sofia, Bulgaria
- 7. Tsvetana Stoyanova 🕩 PhD, Full-Time Professor, University of National and World Economy, Sofia, Bulgaria
- Diana Kopeva 匝 8. PhD, Full-Time Professor, University of National and World Economy, Sofia, Bulgaria
- 9. Silvia Trifonova 🕩 PhD, Full-Time Professor, University of National and World Economy, Sofia, Bulgaria
- 10. Georgi Zabunov 🕩
- PhD, Associate Professor, University of National and World Economy, Sofia, Bulgaria Hristina Oreshkova D
- PhD, Associate Professor, University of National and World Economy, Sofia, Bulgaria 12. Dragomir Stefanov 🛄
- PhD, Associate Professor, University of National and World Economy, Sofia, Bulgaria Svilen Kolev (
- PhD, Assistant Professor, University of National and World Economy, Sofia, Bulgaria 14. Yanko Hristozov 🗈
- PhD, Assistant Professor, University of National and World Economy, Sofia, Bulgaria 15. Suzana Marković 回
- PhD, Full-Time Professor, University of Rijeka, Faculty of Tourism and Hospitality Management, Opatija, Croatia
- 16. Alžbeta Kiráľová D

5.

- PhD, Associate Professor, AMBIS University, Prague, Czech Republic
- 17. Jiří Strouhal 回 PhD, Full-Time Professor, Department of Finance and Accounting, Skoda Auto University, Mladá Boleslav, Czech Republic
- 18. Aleš Pachmann 回 PhD, Associate Professor, AMBIS University, Prague, Czech Republic
- 19. Josef Polák 🛄 PhD, Associate Professor, AMBIS University, Prague, Czech Republic 20. Gregory Koronakos 回

PhD, Professor, University of Piraeus, Department of Informatics, Decision Support Systems Laboratory, Greece

21. Maria Lakatos

PhD, Associate Professor, Department of Finance and Accounting, Faculty of Economics and Business, University of János Neumann, Kecskemét, Hungary

- 22. Éva Erdélyi 🕩
 - PhD, Associate Professor, Faculty of Commerce, Hospitality and Tourism, Budapest Business School, University of Applied Sciences, Budapest, Hungary
- 23. Cheryl Venan Dias 💷
- PhD, Assistant Professor, Department of Tourism Studies, S.S. Dempo College of Commerce and Economics, Goa, India 24. Ubaldo Comite 🛄
- PhD, Associate Professor, Faculty of Economy, University Giustino Fortunato, Benevento, Italy
- 25. Annalisa Baldissera 👊
 - PhD, Associate Professor, University of Brescia, Business Economics, Brescia, Italy
- 26. Marco Taliento 💷
- PhD, Associate Professor, University of Foggia, Foggia, Italy
- 27. Milen Filipov 🕩 PhD, Assistant Professor, KIMEP University, Almaty, Kazakhstan 28. Dragoljub Janković

PhD, Associate Professor, Government of Montenegro, Adviser to the Deputy Prime Minister for Economic Policy and Financial System, Faculty of Business Studies, Mediterranean University, Podgorica, Montenegro

29. Marija Janković 🕩 PhD, Associate Professor, Dean, Faculty of Economics and Business, Mediterranean University, Podgorica, Montenegro

30. Jelena Žugić PhD, Associate Professor, Vice Dean for Academic Affairs, Finance and International Cooperation, Faculty of Economics and Business, Mediterranean University, Podgorica, Montenegro 31. Slobodanka Krivokapić PhD, Associate Professor, Faculty of Economics and Business, Mediterranean University, Podgorica, Montenegro 32. Sasho Kiosev 🕩 PhD, Full-Time Professor, Faculty of Economics, University "Ss. Cyril and Methodius", Skopje, North Macedonia 33. Karolina Olszewska 匝 PhD, Assistant Professor, Faculty of Social Sciences, Institute of International Studies, Wroclaw University of Economics and Business, Wroclaw, Poland 34. Andrzej Sztando 匝 PhD, Associate Professor, Department of Spatial Economy and Self-Governed Administration, Wroclaw University of Economics and Business, Wroclaw, Poland 35. Rui Dias 🕩 PhD, Associate Professor, Polytechnic Institute of Setúbal, School of Business Administration and Researcher at the University of Évora (CEFAGE), Setúbal, Portugal 36. Paulo Monteiro Alexandre 匝 PhD, Associate Professor, Polytechnic Institute of Setúbal, School of Business Administration, Setúbal, Portugal 37. Rocsana Manea Tonis 🛄 PhD, Associate Professor, National University of Physical Education and Sport, Bucharest, Romania 38. Zobova Ludmila L. 匝 PhD, Full-Time Professor, Institute of Economics, Kemerovo State University, Kemerovo, Russia 39. Rabab Ali Abumalloh PhD, Assistant Professor, Imam Abdulrahman Bin Faisal University, Al-Dammam, Saudi Arabia 40. Dejan Spasić 🕩 PhD, Full-Time Professor, Faculty of Economics, University of Niš, Serbia 41. Sanja Jelisavac Trošić 🕩 PhD, Senior Research Fellow, Institute of International Politics and Economics, Belgrade, Serbia 42. Dobrila Lukić PhD, Research Associate, "Jovan Cvijić" Geographical Institute of the Serbian Academy of Sciences and Arts, Belgrade, Serbia 43. Miodrag Brzaković 🕩 PhD, Full Time Professor, Faculty of Applied Management, Economy and Finance, Belgrade, Serbia 44. Tomislav Brzaković 回 PhD, Full Time Professor, Dean, Faculty of Applied Management, Economy and Finance, Belgrade, Serbia 45. Darjan Karabašević 回 PhD, Associate Professor, Vice Dean for Scientific Research, Faculty of Applied Management, Economy and Finance, Belgrade, Serbia 46. Pavle Brzaković 🕩 PhD, Assistant Professor, Faculty of Applied Management, Economy and Finance, Belgrade, Serbia 47. Pavle Radanov 🕩 PhD, Associate Professor, Faculty of Applied Management, Economy and Finance, Belgrade, Serbia 48. Tatjana Dolinšek 👊 PhD, Assistant Professor, Deputy Dean, Faculty of Commercial and Business Sciences, Celje, Slovenia 49. Anton Manfreda 💷 PhD, Assistant Professor, Faculty of Economics, University of Ljubljana, Slovenia 50. Eva Podovšovnik PhD, Associate Professor, University of Primorska, Faculty of Tourism Studies Turistica, Portorož, Slovenia 51. Ľubica Hurbánková 匝 PhD, Assistant Professor, Department of Statistics at the University of Economics in Bratislava, Slovakia 52. Dana Országhová 🕩 PhD, Associate Professor, Institute of Statistics, Operations Research and Mathematics, Faculty of Economics and Management, Slovak University of Agriculture in Nitra, Slovakia 53. María del Carmen Valls Martínez 匝 PhD, Professor, Department of Economics and Business, University of Almería, Spain 54. Vanesa Sánchez Ballesteros 匝 PhD, Professor, University of Jaén, Spain 55. Erhan İşcan 匝 PhD, Assistant Professor, Department of Economics, Cukurova University, Adana, Turkey 56. Nadiia Kobetska PhD, Full-Time Professor, Law Institute, Vasyl Stefanyk Precarpathian National University, Ukraine, Ivno-Frrankivsk, Ukraine 57. Lyudmila Batsenko 匝 PhD, Assistant Professor, Sumy National Agrarian University, Sumy, Ukraine 58. Ana Bogdanović 🕩 PhD, Associate Professor, Newcastle University Business School, Newcastle upon Tyne, United Kingdom



Contents

| Contents |
|---|
| Index of Authors |
| IndexIX |
| PrefaceXI |
| ERAZ 2022 Participants' AffiliationXIII |
| Risk Diversification in Central and Eastern European Capital Markets: Evidence from Russia's Invasion of Ukraine |
| Contribution of the Institutions of the Regional Political System |
| to the Preservation of the Level and Quality of Life |
| of the Population in the Context of the COVID-19 Pandemic (The Case Study of the Trans-Baikal Territory)11 |
| Yulia Matafonova Tatyana Gordeeva |
| COVID-19: A Story Told by Women Entrepreneurs about |
| How to Survive by Turning a Crisis into an Opportunity19 Tudor Tarbujaru Petruta Ionela Stanca |
| The Role of Women in Founding Top Startups Worldwide |
| Katia Giusepponi Ernesto Tavoletti |
| Financialization – Evidence from Dynamic Connectedness |
| among Agricultural Index Futures |
| Financial Literacy of People Gaining Economic Education |
| in the Context of Acquired Practical Experience |
| Preparation for the Career outside the Army: |
| The View of Soldiers of the Czech Armed Forces57Renata SkýpalováKristýna Binková |
| The Impact of Personality Traits |
| and Internal Locus of Control on Entrepreneurial Intention |
| Barriers to Effective Interpersonal Communication in Retail |
| Virtual vs. Live Conferences: |
| Measuring the Participants Memorable Conference Experience |
| Typology of Social Paradigm: |
| Literature Review of Tourist's Gastronomic Experience |



Index of Authors

С

D

E

J

Κ

Aleš Kozubík, 43

Catarina Revez, 1

Dora Rašan, 97; 107

Ernesto Tavoletti, 27

Josef Polák, 43

Katerina Fotova Čiković, 73 Katia Giusepponi, 27 Kristina Pilko, 107 Kristýna Binková, 57

М

Ν

Marina Laškarin Ažić, 97

Neeti Jain, 35 Nicole Horta, 1 Niti Nandini Chatnani, 35

o Ondřej Drahotský, 87

Ρ

Paula Heliodoro, 1 Paulo Alexandre, 1 Petra Kováčová, 87 Petruta Ionela Stanca, 19

R

Renata Skýpalová, 57 Rui Dias, 1

т

Tatyana Gordeeva, 11 Tudor Tarbujaru, 19

Y

Yulia Matafonova, 11

Ζ

Zuzana Kozubíková, 43

Index

Agricultural commodity markets, 35 Arbitrage, 1

Α

Behavioural intentions, 97

В

C

Central-Eastern European markets, 1 Challenges, 19 Communication partner, 87 Content analysis, 107 COVID-19, 11; 19 Crisis management, 19 Czech Armed Forces, 57

D

Decision making, 43

E

Economic education, 43 Employees, 87 Entrepreneurial intention, 73 Entrepreneurship, 73 Experience economy, 107

F

Finance, 43 Financial management, 43 Financial market, 43 Financial pressure, 19 Founding team, 27 Futures markets, 35

G

Gastronomic experience, 107

Independent t-test, 97 Internal locus of control, 73

L

1 .

Labor market, 57 Level and quality of life of the population, 11 Level of administration, 11 Literature review, 73

Μ

Memorable conference experience, 97 Method of dichotomous assessment, 11 Military organization, 57

0

Ρ

Organization, 87

Paradigm, 107 Personality traits, 73

R

Risk diversification, 1 Role of women in startups, 27 Russian-Ukrainian invasion, 1

S

Second career, 57 Social efficiency, 11 Sociology, 57 Soldiers, 57 Startup, 27 Sustainability of regional political systems, 11 Sustainable development of the region, 11

Transition, 57 TVP-VAR, 35 Virtual conference, 97 Women co-founders, 27 Women entrepreneurs, 19

Т





Preface

Economic development refers to enhancing economic activities in a society, resulting in positive changes in both the socio-economic structure and living standards. Sustainable economic development aims to eradicate poverty, inequality, and unemployment, ultimately promoting social inclusion and improving the overall quality of life. Consequently, analyzing this crucial issue requires a highly interdisciplinary approach.

The issue of sustainable economic development has gained recognition and attention from esteemed academic institutions in the Balkans region and the Czech Republic. Notably, University of National and World Economy – Sofia, Bulgaria; Faculty of Economics and Business, Mediterranean University – Podgorica, Montenegro; Faculty of Commercial and Business Studies – Celje, Slovenia; Faculty of Applied Management, Economics and Finance, Belgrade; Association of Economists and Managers of the Balkans along with the AMBIS University from Prague, Czech Republic have recognized the following issue and organized the 8th International Scientific Conference titled: *Knowledge Based Sustainable Development – ERAZ* 2022 online/virtually (Prague, Czech Republic) on May 26, 2022.



The primary aim of the conference was to facilitate the gathering of the academic community, including experts, scientists, engineers, researchers, students, and other interested parties, with the purpose of disseminating scientific knowledge and promoting personal and collective growth. To achieve this goal, the conference provided a platform for the presentation and publication of scientific papers, as well as interactive discussions and other forms of interpersonal exchange, which enabled participants to share their experiences and knowledge. The conference program was designed to cover the latest scientific developments in the following areas:

- Microeconomics and macroeconomics,
- Economic policy,
- International Economics and Trade,
- International Business,
- Economic diplomacy,
- Lobbying,
- Globalization,
- European business,
- Modern management and innovation,
- Business and Public Finance,
- Fiscal policy,
- Stock exchange and financial markets,
- Risk management,
- Insurance and reinsurance companies,
- Financial Management and Banking,
- Modern forms of entrepreneurship and investment,
- Investment Management,
- Enterprise and Learning,
- Women and Entrepreneurship,
- Corporate entrepreneurship,
- Agribusiness Strategy,
- Marketing and trade,
- Marketing services,
- Marketing of non-profit sector,

- Research in marketing,
- Marketing in education,
- Marketing in sport,
- Marketing in culture,
- Accounting and auditing,
- Quality management,
- Labor law,
- Business law,
- The role of the rule of law in the country's progress,
- Human rights and protection of minorities,
- Legal aspects of EU integration,
- Intellectual Property Law,
- The reform of corporate law in countries in transition,
- CEFTA,
- Ecology and energy,
- Renewable energy,
- Energetic efficiency,
- Information technology and business intelligence,
- The use and integration of new technologies,
- E-society and E-learning,
- Sustainable tourism,
- Hospitality

Within publications from the ERAZ 2022 conference:

- 11 double peer reviewed papers have been published in the International Scientific Conference ERAZ 2022 Knowledge Based Sustainable Development Selected Papers,
- 37 double peer reviewed papers have been published in the International Scientific Conference ERAZ 2022 Knowledge Based Sustainable Development **Conference Proceedings**,
- 70 abstracts have been published in the International Scientific Conference ERAZ 2022 Knowledge Based Sustainable Development **Book of Abstracts.**

ERAZ 2022 publications have nearly 600 pages.

The conference attracted the participation of over **100 researchers** from **20 different countries**, including Albania, Bosnia and Herzegovina, Bulgaria, Croatia, Czech Republic, Hungary, India, Indonesia, Italy, North Macedonia, Palestine, Portugal, Romania, Russia, Serbia, Slovakia, Spain, Taiwan, Turkey, and Ukraine. These researchers came from a diverse range of academic institutions, such as universities, eminent faculties, and scientific institutes, as well as from colleges, various ministries, local governments, public and private enterprises, multinational companies, and associations.



ERAZ 2022 Participants' Affiliation

Albania

- Aleksander Moisiu University, Faculty of Professional Studies, Durres
- Health Care Centre, Nr. 4, Shkoder
- National Environment Agency, Tirana
- University of Shkodra "Luigj Gurakuqi", Sheshi "Dugajt e Reja"

Bosnia and Herzegovina

• University of Zenica, Faculty of Economics, 72000 Zenica

Bulgaria

- Economic Research Institute at BAS, Sofia
- New Bulgarian University, 21 Montevideo Str., 1618 Sofia
- Technical University of Varna, 1 Studentska Str., Varna
- University of National and World Economy (UNWE), Faculty of Finance and Accounting; Blvd. "8-mi Dekemvri", 1700 Studentski Kompleks, Sofia

Croatia

- Institute for Development and International Relations (IRMO), Lj. F. Vukotinovića 2, 10000 Zagreb
- University North, Trg dr. Žarka Dolinara 1, 48000 Koprivnica
- University of Rijeka, Faculty of Tourism and Hospitality Management, Primorska 42, p.b. 97, 51410 Opatija

Czech Republic

- AMBIS University, Department of Economics and Management, Lindnerova 575/1, Prague
- Masaryk University, Faculty of Law, Veveří 70, 611 80 Brno
- Silesian University in Opava, School of Business Administration in Karvina; Univerzitní náměstí 1934/3, 733 40 Karviná
- University of Defense, Faculty of Military Leadership, Department of Resources Management, Kounicova 65, Brno

Hungary

- Hungarian University of Agriculture and Life Sciences, Guba Sándor u. 40, H-7400, Kaposvár
- University of Applied Sciences, Budapest Business School, Markó utca 29-31. H-1055 Budapest

India

- Indian Institute of Foreign Trade, B-21 Qutub Institutional Area, 110016 New Delhi
- TERI School of Advanced Studies; Vasant Kunj, New Delhi

Indonesia

• Muhammadiyah University of Yogyakarta, UMY Integrated Campus, Brawijaya Street, Kasihan, Bantul, Yogyakarta

Italy

- ASST Lecco, Via Eremo 9/11, Lecco
- Polytechnic University of Marche, Department of Management, Piazzale Martelli 8, 60121, Ancona
- University of Brescia, Department of Law, Via San Faustino 41, Brescia
- University of Macerata Department of Education, Cultural Heritage and Tourism, P.le Bertelli 1, c.da Vallebona, 62100 Macerata
- University of Macerata Department of Political Sciences, Communication and International Relations, Piazza Strambi 1, 62100 Macerata
- University of Salento; Department of History, Society and Human Studies - Via di Valesio - 73100 Lecce

North Macedonia

- Komercijalna banka ad Skopje, ul. Orce Nikolov, No. 3, Skopje
- · Law company Cukic & Markov, Skopje
- Ss Cyril and Methodius University in Skopje, Faculty of Economics Skopje, bul. Goce Delcev 9v, Skopje
- University American College Skopje; Treta makedonska brigada nr. 60, 1000 Skopje
- University St. Kliment Ohridski Bitola

Palestine

• Al-Quds Open University, Northern Gaza

Portugal

- Polytechnic Institute of Guarda, Guarda
- School of Business and Administration, Polytechnic Institute of Setúbal, Setúbal
- University of Évora, CEFAGE-UE, IIFA

Romania

- National Institute of Research Development and Testing in Electrical Engineering, Craiova
- Romanian-American University, Bucharest
- Technical University of Civil Engineering Bucharest; Bd. Lacul Tei no. 122-124, s.2, 020707, Bucharest
- University of Agronomic Sciences and Veterinary Medicine of Bucharest, 59 Mărăști Blvd, Bucharest
- University of Craiova, Craiova

Russia

• Trans-Baikal State University, Chita

Serbia

- Faculty of Social Sciences, Belgrade
- National Assembly of the Republic of Serbia, Trg Nikole Pašića 13, Belgrade
- University of Belgrade, Faculty of Agriculture, Nemanjina 6, 11081 Belgrade
- Western Serbia Academy of Applied Studies, Valjevo Department, Vuka Karadžića 3a, 14000 Valjevo

Slovakia

- Alexander Dubček University in Trenčín, Faculty of Social and Economic Relations, Študentská 3, 91150 Trenčín
- Bratislava University of Economics and Management, Furdekova ul. 16, 851 04 Bratislava
- Comenius University, Faculty of Management, Odbojarov 10, 825 05 Bratislava
- Slovak University of Agriculture in Nitra, Faculty of Economics and Management, Tr. A. Hlinku 2, 949 76 Nitra
- University of Žilina, Faculty of Management and Informatics, Dpt. of Macro and Microeconomics, Univerzitná 8215/1, 010 26 Žilina
- University of Žilina, Faculty of Management and Informatics, Dpt. of Mathematical Methods and Operations Research, Univerzitná 8215/1, 010 26 Žilina

Spain

- Autonomous University of Madrid, Ciudad Universitaria de Cantoblanco Ctra. de Colmenar Viejo, Madrid
- UCAM Catholic University of Murcia, Campus de los Jerónimos, Guadalupe 30107, Murcia
- University of Almería, Mediterranean Research Center on Economics and Sustainable Development; La Cañada de San Urbano s/n - 04120 - Almería
- University of Extremadura, Faculty of Economics and Business, Av de Elvas s/n, 06006 Badajoz
- University of Jaén, Department of Civil, Financial and Tax law
- University of Málaga, Avda. Cervantes, 2. 29071 Málaga

Taiwan

• National Taipei University, New Taipei

Turkey

• Yaşar University, Department of Logistics Management, Bornova

Ukraine

• Taras Shevchenko National University of Kyiv, Anton Tsedik Str., 20, 03057 Kyiv



Risk Diversification in Central and Eastern European Capital Markets: Evidence from Russia's Invasion of Ukraine

Rui Dias¹ Nicole Horta² Catarina Revez³ Paulo Alexandre⁴ Paula Heliodoro⁵

Keywords:

Russian-Ukrainian invasion; Central-Eastern European markets; Arbitrage; Risk diversification



Received: July 1, 2022 Accepted: August 26, 2022 Published: December 30, 2022

Abstract: Following the Revolution of Dignity in Ukraine in 2014, Russia annexed Crimea, while separatist forces supported by the Russian government seized part of the Donbas region in south-eastern Ukraine. Since the beginning of 2021, a build-up of Russian military presence has occurred along the Russia-Ukraine border. The United States and other countries have accused Russia of planning an invasion of Ukraine. On February 24th, Putin announced a "special military operation," supposedly to "demilitarize" and "denazify" Ukraine. In light of these events, the global economy and consequently the financial markets had significant structural breaks; based on these facts, this paper aims to analyze the synchronizations between the capital markets of Austria (Austrian Traded), Budapest (BUX), Bulaaria (SE SOFIX), Croatia (CROBEX), Russia (MOEX), Czech Republic (Prague SE PX), Romania (BET), Slovakia (SAX 16), and Slovenia (SBI TOP), in the period from January 2nd, 2017 to May 6th, 2022. To perform this analysis and to get more robust results we divided the sample into two sub-periods: The first from January 2nd, 2017, to December 31st, 2019, with the second sub-period called capital markets stress comprising the time lapse from January 1st, 2020, to May 6th, 2022. In order to answer the research question, we aim to find out, whether the Russian invasion of Ukraine accentuated interdependencies in Central/Eastern European financial markets. The time series do not show normal distributions, with the Russian market showing the highest risk; we find that the markets broke down significantly, mostly in March 2022 arising from instability in the global economy. The results obtained suggest very significant levels of integration during the stress period in the capital markets analyzed, and we see that during the quiet period the Slovakian market tends to be highly integrated (8 out of 8 possible), while the Slovenian market shows no integration with its regional peers, which shows that we are dealing with a segmented market. These findings suggest that markets tend toward integration in periods of extreme volatility, calling into question the implementation of efficient portfolio diversification strategies.

1. INTRODUCTION

Financial instability is a very important factor for society since a financial crisis or a stock market crash can directly or indirectly affect the level of the economic well-being of a country's inhabitants. If a certain stock market is strongly linked to the stock market of another country, the financial stability of the former depends in part on the financial stability of the latter. For this reason, a close or strong connection between markets increases the levels of vulnerability

⁵ School of Business and Administration, Polytechnic Institute of Setúbal, Portugal



¹ School of Business and Administration, Polytechnic Institute of Setúbal, Portugal; CEFAGE-UE, IIFA, University of Évora, Portugal.

² School of Business and Administration, Polytechnic Institute of Setúbal, Portugal

³ School of Business and Administration, Polytechnic Institute of Setúbal, Portugal

⁴ School of Business and Administration, Polytechnic Institute of Setúbal, Portugal

to external shocks and, as a result, influences the economic conditions and welfare levels of the respective countries. Thus, the occurrence of integration between markets can have significant implications for international risk diversification (Bagão et al., 2020; R. Dias et al., 2020; Dias et al., 2021b, 2021a; Dias et al., 2020, 2021, 2022; Pardal, P., Dias, R., Šuleř, P., Teixeira, N., and Krulický, 2020; Pardal et al., 2021; Silva et al., 2020).

When integrating equity markets, investors generally seek to include assets in their portfolios that hold lower levels of correlation to promote effective diversification strategies. In the same context, Grubel (1968) and Levy and Sarnat (1970) argue that investing in international stock markets is substantiated by the fact that the correlation between assets is lower than that examined in domestic assets. Therefore, the low correlation between international stock markets is a key factor for portfolio diversification.

This paper aims to analyze financial integration in the capital markets of Austria (Austrian Traded), Budapest (BUX), Bulgaria (SE SOFIX), Croatia (CROBEX), Russia (MOEX), Czech Republic (Prague SE PX), Romania (BET), Slovakia (SAX 16), and Slovenia (SBI TOP) during the period from January 2nd, 2017, to May 6th, 2022. The results obtained suggest very significant levels of integration, and this evidence decreases the chances of portfolio diversification in the long run, except for the Slovenian market.

This research adds relevant contributions to the literature, namely to the study of risk diversification in Central and Eastern European markets in the context of the Russian invasion of Ukraine; to the best of our knowledge, this is the first study that analyzes in isolation these financial markets during this time-lapse. In terms of structure, this paper is organized into 5 sections. Section 1 is represented by the current introduction. Section 2 presents a Literature Review of articles on financial market integration. Section 3 describes the data and methodology. Section 4 contains the results. Finally, Section 5 presents the general conclusions of the paper.

2. LITERATURE REVIEW

Understanding international linkages between financial markets in periods of the financial crisis is relevant for investors, fund managers, and academics in different aspects, including portfolio diversification (Lee, 2017).

Voronkova (2004) analyzed the level of financial integration between Central European markets and developed markets; after accounting for structural changes, the author shows that Central European capital markets exhibit significant levels of integration, suggesting that the hypothesis of portfolio diversification may be called into question. Authors Lucey and Voronkova (2008), on the other hand, analyzed the integration of financial markets before and after the 1998 crisis, with cointegration tests with regime shifting, suggesting that markets are partially integrated.

Syllignakis and Kouretas (2011) highlight that the Central European capital markets (Czech Republic, Hungary, Poland, Slovenia, and Slovakia), and the German and US stock markets are integrated, while the Estonian and Romanian markets are segmented.

Özer, Kamışlı and Kamışlı (2016) show hybrid results by failing to highlight comovement between the market of Germany, Austria, Czech Republic, Croatia, Lithuania, and Greece. These findings have important implications for international investors, portfolio managers and

policymakers. Authors Moagar-Poladian, Clichici and Stanciu (2019), meanwhile, highlight that Central and Eastern European capital markets show a significant level of integration during the 2010 financial crisis.

Liu, Manzoor, Wang, Zhang and Manzoor (2020) studied the impact of the 2020 global pandemic on 21 capital markets, suggesting the presence of significant structural breaks. Zeren and Hizarci (2020), analyzed the effects of the Covid-19 epidemic on equity markets over the period January 23, 2020, and March 13, 2020. The authors' evidence levels of causality between the number of deaths from the global outbreak and financial markets. It was understood that the global outbreak cases have cointegrating relationships with the SSE, KOSPI and IBEX35 markets, but not with the FTSE MIB, CAC 40, DAX 30 markets.

In more recent studies, author Hung (2022) analyzed the integration in the capital markets of Croatia, Czech Republic, Hungary, Poland, and Romania using the multivariate dynamics equicorrelation - generalized autoregressive conditional heteroskedasticity (DECOGARCH) model and the spillover index. The results show that the average return equicorrelation between CEE and world stock indices is positive; this evidence undermines the benefits of portfolio diversification in these regional markets.

In summary, this work aims to contribute by providing information to investors and regulators in Central and Eastern European capital markets, where individual and institutional investors seek diversification benefits, as well as to help promote the implementation of policies that contribute to the efficiency of these markets.

3. METHODOLOGY AND DATA

3.1. Data

Data for the closing prices of the capital markets in Austria (Austrian Traded), Budapest (BUX), Bulgaria (SE SOFIX), Croatia (CROBEX), Russia (MOEX), Czech Republic (Prague SE PX), Romania (BET), Slovakia (SAX 16), and Slovenia (SBI TOP) for the period from January 2nd, 2017, to May 6th, 2022, were analyzed.

In order to perform this analysis and obtain more robust results, we have divided the sample into two sub-periods: the quiet one from January 2nd, 2017, to December 31st, 2019, and the second one called capital markets stress, which includes the super-period from January 1, 2020, to May 6, 2022. To mitigate distortions in exchange rates, we have maintained the stock indices in local currency.

3.2. Methodology

The research was developed in several stages. The sample was characterized using descriptive statistics, the adherence test of Jarque and Bera (1980), and quantile plots. To assess the stationarity of time series we will use the ADF (Dickey and Fuller, 1981), PP (Perron and Phillips, 1988) tests in the panel with Fisher and Choi transformations. To answer the research question, we will use the methodology of Gregory and Hansen (1996), considering we have examined a very turbulent period in the financial markets. Additionally, the reason why standard cointegration tests such as Engle and Granger (1987) and Johansen (1988) are not appropriate for testing cointegration with regime change is that such tests assume that the cointegration vector is time-invariant. 8th International Scientific Conference ERAZ 2022 Selected Papers

4. **RESULTS**

Figure 1 shows the evolution, in levels, of the 9 capital markets under analysis, namely, the stock market indices of Austria (Austrian Traded), Budapest (BUX), Bulgaria (SE SOFIX), Croatia (CROBEX), Russia (MOEX), Czech Republic (Prague SE PX), Romania (BET), Slovakia (SAX 16) and Slovenia (SBI TOP). Through graphical observation of the period from January 2nd, 2017, to May 6th, 2022, we see the existence of significant structure breaks, arising from the 2020 pandemic crisis as well as during the Russian invasion of Ukraine. These findings are in line with the evidence suggested by authors Vasco et al. (2021), Dias et al. (2022), Zebende et al. (2022) that show significant structure breakdowns in international capital markets.



Figure 1. Evolution, in levels, of the 9 financial markets, in the period from January 1st, 2017, to May 6th, 2022 Source: Own elaboration

Figure 2 shows the evolution of the return of the 9 stock markets under analysis and shows extreme volatility during the first and second quarters of 2020, as well as in 2022, derived from a sharp drop in the index prices due to the evolution of the global pandemic (Covid-19) and in 2022 due to the military operation carried out by Russia on Ukrainian territory.

In table 1 we can observe the descriptive statistics and verify that the average returns are positive, while the Russian market presents the sharpest standard deviation (0.17251). When we check the distribution of the data, we see that the time series present negative skewness, namely the IMOEX index (-8.837122), while considering the kurtosis, the values are different from 3, i.e., the skewness and kurtosis coefficients are statistically different from those with a normal distribution. These findings are validated by the adherence test of Jarque and Beta (1980), where the null hypothesis that the data follow a normal distribution is rejected at a significance level of 1%. Risk Diversification in Central and Eastern European Capital Markets: Evidence from Russia's Invasion of Ukraine



Figure 2. Evolution of the return of the 9 capital markets, in the period from January 1st, 2017, to May 6th, 2022 Source: Own elaboration

In summary, the Russian market shows much higher than average levels of volatility which was expected due to the Russian invasion of Ukraine in February 2022 and the instability caused in the global economy.

| | Mean | Median | Std. Dev. | Skewness | Kurtosis | Jarque-Bera | Probability | Observations |
|----------|----------|----------|-----------|-----------|----------|-------------|-------------|--------------|
| BUX | 0.000201 | 0.000223 | 0.013073 | -1.655406 | 18.63376 | 14843.73 | 0.000000 | 1395 |
| SOFIX | 2.28E-05 | 0.000000 | 0.008050 | -2.722058 | 39.15623 | 77707.97 | 0.000000 | 1395 |
| CROBEX | 4.81E-05 | 0.000100 | 0.008220 | -3.632165 | 49.42244 | 128329.1 | 0.000000 | 1395 |
| MOEX | 5.32E-05 | 0.000298 | 0.017251 | -8.837122 | 230.9672 | 3038857. | 0.000000 | 1395 |
| PRAGUE | 0.000255 | 0.000259 | 0.009319 | -1.365256 | 19.44129 | 16145.49 | 0.000000 | 1395 |
| BET | 0.000404 | 0.000411 | 0.010737 | -1.878556 | 26.88813 | 33989.09 | 0.000000 | 1395 |
| SAX16 | 0.000131 | 0.000000 | 0.009444 | -0.043912 | 13.64513 | 6587.103 | 0.000000 | 1395 |
| SBITOP | 0.000338 | 0.000116 | 0.008565 | -2.049224 | 26.18284 | 32215.27 | 0.000000 | 1395 |
| AUSTRIAN | 0.000123 | 0.000230 | 0.013821 | -1.347490 | 22.11028 | 21649.56 | 0.000000 | 1395 |

Table 1. Descriptive statistics, in returns, of the 9 capital marketsover the period January 1st, 2017, to May 6th, 2022

Source: Own elaboration

Through the graphic observation of quantiles illustrated in Figure 3, one can also infer the normality of the time series data under analysis. Comparing the data dispersion relative to the normal distribution line, it appears that none of the series is completely overlapping, and there is some skewness. Nevertheless, we find that the time series tend towards an approximately normal distribution, given that the number of observations is sufficiently large.

8th International Scientific Conference ERAZ 2022 Selected Papers



in the period from January 2nd, 2017, to May 6th, 2022

Source: Own elaboration

Table 2 shows the ADF and PP stationarity tests with Fisher and Choi transformations for the 9 capital markets under analysis, namely, the stock indices of Austria (Austrian Traded), Budapest (BUX), Bulgaria (SE SOFIX), Croatia (CROBEX), Russia (MOEX), Czech Republic (PragueSE PX), Romania (BET), Slovakia (SAX 16) and Slovenia (SBI TOP), for the complete sample period. The tests suggest that the time series are stochastic only in the first differences, that is, we are facing a white noise that allows us to validate the robustness of the results.

Table 2. Stationarity tests, in return, of ADF, with Fisher and Choi transformation, concerningthe 9 capital markets for the period from January 2nd, 2017, to May 6th, 2022

| Null Hypothesis: Unit root (individual unit root process) | | | | | | |
|---|-----------|---------|--|--|--|--|
| Method | Statistic | Prob.** | | | | |
| ADF - Fisher Chi-square | 1295.75 | 0.0000 | | | | |
| ADF - Choi Z-stat | -33.2900 | 0.0000 | | | | |

**Note: Probabilities for Fisher tests are computed using an asymptotic Chi-square distribution. All other tests assume asymptotic normality.

Source: Own elaboration

| Null Hypothesis: Unit root (individual unit root process) | | | | | |
|---|----------|--------|--|--|--|
| Method Statistic Prob.** | | | | | |
| PP - Fisher Chi-square | 605.291 | 0.0000 | | | |
| PP - Choi Z-stat | -22.5453 | 0.0000 | | | |

Table 3. Stationarity tests, in return, of PP, with Fisher and Choi transformation, concerning the 9 capital markets, in the period from January 2nd, 2017, to May 6th, 2022

**Note: Probabilities for Fisher tests are computed using an asymptotic Chi-square distribution. All other tests assume asymptotic normality.

Source: Own elaboration

In Table 4 it is possible to analyze the integrations between the pairs of financial markets under analysis during the two sub-periods: Quiet and during the period marked by the 2020 pandemic and more recently by Russia's invasion of Ukraine in 2022. During the quiet period we verified the presence of 15 integrated market pairs (out of 72 possible), with the Slovakian market registering the highest level of integration (8 out of 8 possible). Slovenia, on the other hand, has no integrations with its regional peers, which shows that this is a market that is segmented. When we analyze the sub period of stress in capital markets, we see that the level of integration rose sharply (from 15 to 33); the Slovak market continues to maintain, among the analyzed markets, the highest number of integrations, registering 7 out of 8 possible, followed by the Russian and Romanian markets with 5 integrations (out of 8 possible). The stock indices of Budapest and Prague showed 4 integrations, while the markets of Austria and Slovenia with 3 integrations. Bulgaria and Croatia maintained their level of integration (1 out of 8 possible). These findings are in line with the evidence suggested by Dias et al. (2020), Dias et al. (2021), Dias and Carvalho (2021), and Dias et al. (2022) that in periods of extreme volatility markets tend towards integration thus calling into question the hypothesis of efficient portfolio diversification.

| markets under undrysis; in the Quiet and erisis periods | | | | | | | |
|---|--------------|------------------------|--------------|--|--|--|--|
| Markets | Quiet | Crisis | Evolution | | | | |
| BUDAPEST BUX | 1/8 possible | 3/8 possible | ↑ | | | | |
| BULGARIA SE SOFIX | 1/8 possible | 4/8 possible | 1 | | | | |
| CROATIA CROBEX | 1/8 possible | 1/8 possible | = | | | | |
| MOEX RUSSIA | 1/8 possible | 1/8 possible | = | | | | |
| PRAGUE SE PX | 0/8 possible | 5/8 possible | 1 | | | | |
| ROMANIA BET | 1/8 possible | 4/8 possible | \uparrow | | | | |
| SLOVAKIA SAX 16 | 2/8 possible | 5/8 possible | 1 | | | | |
| SLOVENIAN BLUE | 0/2 magnihia | $\frac{2}{8}$ maggible | ^ | | | | |
| CHIP SBI TOP | 0/8 possible | 3/8 possible | 1 | | | | |
| AUSTRIAN TRADED | 8/8 possible | 7/8 possible | \downarrow | | | | |

Table 4. Summary of the results obtained in the Gregory-Hansen tests, in the financialmarkets under analysis, in the Quiet and Crisis periods

Source: Own elaboration

5. CONCLUSION

This paper analyzed financial integration in the capital markets of Austria (Austrian Traded), Budapest (BUX), Bulgaria (SE SOFIX), Croatia (CROBEX), Russia (MOEX), Czech Republic (Prague SE PX), Romania (BET), Slovakia (SAX 16), and Slovenia (SBI TOP) during the period from January 2nd, 2017, to May 6th, 2022. In order to perform this analysis and gauge more robust results we partitioned the sample into two sub-periods: quiet from January 2nd, 2017, to December 31st, 2019, and the second called capital markets stress that comprises the super-period from January 1st, 2020 to May 6th, 2022. In order to answer the research question, we intend to find out, whether the Russian invasion of Ukraine accentuate interdependencies in Central/ Eastern European financial markets. The results show that during the Quiet period there were 15 integrated market pairs (out of 72 possible), with the Slovak market registering the highest level of integration (8 out of 8 possible). Slovenia, on the other hand, has no integrations with its regional peers, which shows that we are facing a market that is segmented. When we analyze the sub period of stress in capital markets, we see that the level of integration rose sharply (from 15 to 33); the Slovak market continues to maintain, among the analyzed markets, the highest number of integrations, registering 7 in 8 possible, followed by Russia and Romania with 5 integrations (in 8 possible). The stock indices of Budapest and Prague showed 4 integrations, while the markets of Austria and Slovenia with 3 integrations. Bulgaria and Croatia maintained their level of integration (1 in 8 possible).

The general conclusion to be retained and supported by the results obtained through the econometric model tests, is that the global pandemic of 2020 and the Russian invasion of Ukraine in 2022 have a significant impact on the memory properties of the Central and Eastern European financial markets. We find that the level of financial integration is very significant in these markets during this uncertainty in the global economy; so, these findings call into question the implementation of efficient portfolio diversification strategies. In conclusion, we believe that this evidence is relevant for policymakers and investors concerning regional development policies and portfolio diversification strategies in Central and Eastern European financial markets.

REFERENCES

- Bagão, M., Dias, R., Heliodoro, P., & Alexandre, P. (2020). the Impact of Covid-19 on European Financial Markets: an Empirical Analysis. 6th LIMEN Conference Proceedings (Part of LIMEN Conference Collection), 6(July), 1–11. DOI: 10.31410/limen.2020.1
- Dias, R. T., & Carvalho, L. (2021). Foreign Exchange Market Shocks in the Context of the Global Pandemic (COVID-19). May, 359–373. DOI: 10.4018/978-1-7998-6643-5.ch020
- Dias, R. T., Pardal, P., Santos, H., & Vasco, C. (2021a). COVID-19 Pandemic and Its Influence on Safe Havens. June, 289–303. DOI: 10.4018/978-1-7998-6926-9.ch016
- Dias, R. T., Pardal, P., Santos, H., & Vasco, C. (2021b). *Testing the Random Walk Hypothesis for Real Exchange Rates. June*, 304–322. DOI: 10.4018/978-1-7998-6926-9.ch017
- Dias, R., Teixeira, N., Machova, V., Pardal, P., Horak, J., & Vochozka, M. (2020). Random walks and market efficiency tests: Evidence on US, Chinese and European capital markets within the context of the global Covid-19 pandemic. *Oeconomia Copernicana*, 11(4). DOI: 10.24136/OC.2020.024
- Dias, Rui, Pardal, P., Teixeira, N., & Machová, V. (2020). Financial Market Integration of ASE-AN-5 with China. *Littera Scripta*, 13(1). DOI: 10.36708/littera_scripta2020/1/4
- Dias, Rui, Pereira, J. M., & Carvalho, L. C. (2022). Are African Stock Markets Efficient? A Comparative Analysis Between Six African Markets, the UK, Japan and the USA in the Period of the Pandemic. *Naše Gospodarstvo/Our Economy*, 68(1), 35–51. DOI: 10.2478/ ngoe-2022-0004
- Dias, Rui, Santos, H., Alexandre, P., Heliodoro, P., & Vasco, C. (2021). Random Walks and Market Efficiency Tests: Evidence for Us and African Capital Markets. 5th EMAN Selected Papers (Part of EMAN Conference Collection), October, 17–29. DOI: 10.31410/ eman.s.p.2021.17
- Dickey, D., & Fuller, W. (1981). Likelihood ratio statistics for autoregressive time series with a unit root. *Econometrica*, 49(4), 1057–1072. DOI: 10.2307/1912517

- Engle, R. F., & Granger, C. W. J. (1987). Co-Integration and Error Correction: Representation, Estimation, and Testing. *Econometrica*, 55(2), 251. DOI: 10.2307/1913236
- Gregory, A. W., & Hansen, B. E. (1996). Residual-based tests for cointegration in models with regime shifts. *Journal of Econometrics*, 70(1), 99–126. DOI: 10.1016/0304-4076(69)41685-7
- Grubel, H. G. (1968). Internationally Diversified Portfolios: Welfare Gains and Capital Flows. *American Economic Review.*, 58(5), p.1299. 16p. DOI: 10.1126/science.151.3712.867-a
- Hung, N. T. (2022). Return equicorrelation and dynamic spillovers between Central and Eastern European, and World stock markets, 2010–2019. *Regional Statistics*, *12*(1). DOI: 10.15196/ RS120108
- Jarque, C. M., & Bera, A. K. (1980). Efficient tests for normality, homoscedasticity and serial independence of regression residuals. *Economics Letters*, 6(3), 255–259. DOI: 10.1016/0165-1765(80)90024-5
- Johansen, S. (1988). Statistical Analysis of Cointegrated Vectors. *Journal of Economic Dynamics and Control*, 12(2–3), 231–254.
- Lee, E.-J. (2017). Intra- and inter-regional portfolio diversification strategies under regional market integration: Evidence from U.S. global banks. *International Review of Financial Analysis*. DOI: 10.1016/j.irfa.2017.08.001
- Levy, H., & Sarnat, M. (1970). International diversification of investment portfolios. *The American Economic Review*, 60(4), 668–675. DOI: 10.2307/1818410
- Liu, H., Manzoor, A., Wang, C., Zhang, L., & Manzoor, Z. (2020). The COVID-19 outbreak and affected countries stock markets response. *International Journal of Environmental Research and Public Health*, 17(8). DOI: 10.3390/ijerph17082800
- Lucey, B. M., & Voronkova, S. (2008). Russian equity market linkages before and after the 1998 crisis: Evidence from stochastic and regime-switching cointegration tests. *Journal of International Money and Finance*, 27(8), 1303–1324. DOI: 10.1016/j.jimonfin.2008.07.004
- Moagar-Poladian, S., Clichici, D., & Stanciu, C. V. (2019). The comovement of exchange rates and stock markets in Central and Eastern Europe. *Sustainability (Switzerland)*. DOI: 10.3390/su11143985
- Özer, M., Kamışlı, M., & Kamışlı, S. (2016). The Analysis of Volatility Spillovers between the German and Central and Eastern European (CEE) Stock Markets by Using Frequency Domain Causality Test. In *Europe and Asia: Economic Integration Prospects*.
- Pardal, P., Dias, R., Šuleř, P., Teixeira, N., & Krulický, T. (2020). Integration in Central European capital markets in the context of the global COVID-19 pandemic. 15(4). DOI: 10.24136/ eq.2020.027
- Pardal, P., Dias, R. T., Santos, H., & Vasco, C. (2021). Central European Banking Sector Integration and Shocks During the Global Pandemic (COVID-19). June, 272–288. DOI: 10.4018/978-1-7998-6926-9.ch015
- Perron, P., & Phillips, P. C. B. (1988). Testing for a Unit Root in a Time Series Regression. *Biometrika*, 2(75), 335–346. DOI: 10.1080/07350015.1992.10509923
- Silva, R., Dias, R., Heliodoro, P., & Alexandre, P. (2020). Risk Diversification in ASEAN-5 Financial Markets: an Empirical Analysis in the Context of the Global Pandemic (Covid-19). 6th LIMEN Selected Papers (Part of LIMEN Conference Collection), 6(July), 15–26. DOI: 10.31410/limen.s.p.2020.15
- Syllignakis, M., & Kouretas, G. P. (2011). Long and Short-Run Linkages in CEE Stock Markets: Implications for Portfolio Diversification and Stock Market Integration. SSRN Electronic Journal. DOI: 10.2139/ssrn.910507
- Vasco, C., Pardal, P., & Dias, R. T. (2021). *Do the Stock Market Indices Follow a Random Walk? May*, 389–410. DOI: 10.4018/978-1-7998-6643-5.ch022

- Voronkova, S. (2004). Equity market integration in Central European emerging markets: A cointegration analysis with shifting regimes. *International Review of Financial Analysis*, 13(5 SPEC.ISS.), 633–647. DOI: 10.1016/j.irfa.2004.02.017
- Zebende, G. F., Santos Dias, R. M. T., & de Águiar, L. C. (2022). Stock market efficiency: An intraday case of study about the G-20 group. *Heliyon*, 8(1), e08808. DOI: 10.1016/j.heli-yon.2022.e08808
- Zeren, F., & Hizarci, a. (2020). The Impact of Covid-19 Coronavirus on Stock Markets: Evidence From Selected Countries. *Muhasebe ve Finans İncelemeleri Dergisi*. DOI: 10.32951/ mufider.706159



Contribution of the Institutions of the Regional Political System to the Preservation of the Level and Quality of Life of the Population in the Context of the COVID-19 Pandemic (The Case Study of the Trans-Baikal Territory)

Yulia Matafonova¹ D Tatyana Gordeeva² D

Received: July 2, 2022 Accepted: August 22, 2022 Published: December 30, 2022

Keywords:

Level and quality of life of the population; Social efficiency; Level of administration; Sustainability of regional political systems; Sustainable development of the region; Method of dichotomous assessment; COVID-19

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-Non-Commercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission. **Abstract:** The study of the activities of regional authorities to preserve the level and quality of life of the population in the context of the COVID-19 pandemic acquires scientific value due to the special conditions in which the regional community and the regional political institutions were placed.

The article analyzes the impact of the conditions of the COVID-19 pandemic on the activities of the institutions of the regional political system to preserve the level and quality of life of the population within a particular subject of the Russian Federation – the Trans-Baikal Territory. As the main research method, a dichotomous assessment of the data obtained as a result of an expert survey in the fall of 2021 is used. The experts were the heads and specialists of the public authorities of the region and local authorities.

The contribution of the institutions of the regional political system to the preservation of the level and quality of life of the population is considered through the categories of "social efficiency" and "level of administration". As a result of the study, we defined the pandemic has a negative impact on the functioning of the institutions of the regional political system in the field of maintaining the level and quality of life of the population. The revealed trends do not allow us to consider the regional political system sufficiently effective and capable of providing the basic conditions for the sustainable development of the region.

1. INTRODUCTION

The conditions of the COVID-19 pandemic have become a test for both population and regional political systems. According to experts and researchers, the difficulties faced by regional authorities are diverse. Restrictions imposed during the pandemic caused many negative socio-economic consequences (reduced income, loss of work, restriction in the sphere of entrepreneurial activity) that in its turn led to a decrease in the level and quality of life. Most often, a reduction of availability of general education (forced expenses of families to provide schoolchildren with computer equipment or means of communication) (Osipov, 2020, p. 35), a shortage of social and medical services (Mau, 2020, p. 146), a negative impact of isolation on psychological health, etc. are seen by Russian researchers as social risks caused by the pandemic. The works of foreign scientists include the similar negative consequences associated with the violation of the standard conditions of social functioning with changes in social and psychological circumstances (Qiu, 2020, p. 1). On the other hand, representatives of social and political sciences note that the conditions of the pandemic contributed to checking the power institutions' ability to make timely decisions aimed at minimizing the crisis consequences and meeting the vital needs of citizens (Zhade, 2021, p. 736).



¹ Trans-Baikal State University, Chita, Russia

² Trans-Baikal State University, Chita, Russia

The level and quality of life are not only complex socio-economic indicators measured in statistical data but also indicators determined through the population's perception and evaluation of its living conditions. While conducting a study of a level and quality of life, it is necessary to take into account not only objective factors (the dynamics of the money income level and expenditures of population, the subsistence level, health status, availability of material resources, work, etc.) but also subjective factors related to the perception of opportunities by individuals and the degree of satisfaction with the living conditions. Therefore, the activities of public authorities (including the regional level) should be evaluated from the point of view of their existence justification (execution of the functions assigned to them) and implementation of certain activities. In this article, we consider the area associated with the preservation of the level and quality of life of the population within the region (the Trans-Baikal Territory) in the context of the pandemic.

The purpose of the article is to identify the effectiveness of the activities of the regional authorities in the Trans–Baikal Territory in the sphere of preserving the level and quality of life of the population in the context of the pandemic, using a dichotomous assessment of the expert survey data.

The study hypothesized that the conditions of the pandemic have a negative impact on the functioning of regional political systems and lead to a decrease in the effectiveness of the functions assigned to them.

2. LITERATURE REVIEW

A significant amount of modern research is devoted to the study of the dynamics of the level and quality of life and the functioning of authorities (including the regional level) in the conditions of uncertainty that have developed under the influence of the COVID-19 pandemic. M.V. Shakleina, M.I. Volkova, K.I. Shaklein, S.R. Yakiro (Shakleina, 2020, p. 137-138), O. Kozlova focuses on the problems of measuring the level and quality of life and the need for a combination of objective and subjective indicators. A., Gladkova T. V., Makarova M. N., Tukhtarova E. Kh. (Kozlova, 2015, p. 281-282). F.D. Algahtani, S.-u.-N. Hassan, B. Alsaif, R. Zrieq revealed the negative impact of the pandemic on predictors of the level and quality of life-based on regression analysis of data. The authors concluded that in the crisis conditions under consideration, the population needs public and institutional support (Algahtani, 2021, p. 10).

The assessment of the impact of the COVID-19 pandemic on social and political processes at the level of separate regions is also widely presented in research papers. Dorothee Allain-Dupre, Isabelle Chatry, Antoine Kornprobst and Maria-Varinia Michalun underline the asymmetry of the impact of the pandemic crisis on the regions, the focus on its territorial dimension and emphasize the special role of the authorities of subnational territorial units (regions, constituent units of the federation) in taking measures to control negative consequences and maintaining the functioning of the health care system, social services, and regional economy (OECD, 2020, November 10). Spatial and managerial aspects of the pandemic are presented in the article by D. McCoy. Professor David McCoy writes about the need to develop a managerial strategy, where he identifies 3 key aspects of the activities of authorities: 1) Dealing with the disease; 2) Managing the health system; 3) Planning for the social and economic impacts of the crisis (McCoy, 2020). The priority of managing the consequences of the COVID-19 pandemic at the level of political and territorial entities is noted by Caroline Chang, Scott Moore and Ali Wyne. They point out that at the national level, the United States did not really cope well with the pandemic, but at the state level, there are numerous examples of managerial practices to follow (Chang, 2020).

Contribution of the Institutions of the Regional Political System to the Preservation of the Level and Quality of Life of the Population in the Context of the COVID-19 Pandemic (The Case Study of the Trans-Baikal Territory)

3. METHODOLOGY

3.1. The Empirical Base of the Study

The collection of empirical data was carried out using:

- the online survey on the Online Test Pad platform https://onlinetestpad.com/hoy6ugd7pkzxm. The online survey was conducted in two stages: the first stage – from October to November 2020, and the second stage – from June to August 2021. 385 respondents took part in the survey;
- in the expert survey conducted in November December 2021 the number of experts was 79 persons: 10 of which were representatives of the executive regional authorities, 7 – legislative, and 62 – representatives of local authorities.

The answers of the respondents and experts were processed using the SPSS statistics program.

3.2. The Method of Dichotomous Assessment

This method of expert assessment is based on determining the evaluation of the indicator taking into account the priority (of direction, scope, tasks, results). It must be recognized that the application of this method introduces certain simplifications into the formulation of a research problem. Nevertheless, the dichotomous approach allows us to obtain necessary data about the problem, its condition, initial estimates and results to test the hypothesis and develop further actions.

The analysis is based on the construction of a goal tree – a hierarchical breakdown of the main goal – the estimated indicator – the contribution of the regional political system institutions to sub–goals - indicators that determine the top assessment. They, in turn, are also divided into two components, etc. (Fig. 1).



Figure 1. Hierarchical breakdown of the main goal – the estimated indicator Source: Own research

In order to obtain an aggregated assessment of the contribution of the regional political system institutions to the preservation of the level and quality of life of the population, we took the directions of "social efficiency" and "level of administration" as the second level elements. To evaluate each of them, we defined the elements of the third level.

To evaluate social effectiveness, we take two variables: "standard of living" and "quality of life". The validity of using these indicators as variables within the scope of the task and the

capabilities of the assessment methodology used is fully justified, since the indicators of the level and quality of life are comparative in their initial content, and in terms of the number of components and assessment methods (the use of quantitative and qualitative assessments) they are different and the result of their assessment is not always unidirectional.

The standard of living is a "numerical indicator" of the authorities' activity, and the quality of life is an integral representation of subjective indicators, consisting of assessments of satisfaction with various spheres of life, including emotional assessments and objective ones – the degree and conditions of satisfaction with a wide range of material and non-material (spiritual) needs.

To assess the indicator "level of administration", we identified the variables "decision-making discretion" and "timeliness" which were evaluated as the average value of expert responses.

Splitting into two elements for each indicator at all levels, except the top one, simplified the process of evaluation – convolution of estimates-indicators from the bottom to the top.

The dichotomous approach allowed us to build assessment matrices based on the priorities chosen by experts. When constructing the matrices for the evaluation of individual indicators, we took into account the monotonicity rule (the left estimate cannot be greater than the right one, and the value of the estimates increases from the lower left corner to the right).

With the help of the prepared evaluation matrices, the averaged expert assessments obtained made it possible to gradually evaluate the lower rows of the indicator elements at first, then the indicators of the next level, and as a result – to obtain an aggregated assessment.

4. RESULTS

As a result, we have formed three matrices to assess the priority and the "result" of the selected indicators. The first one is to assess the social efficiency of the institutions of the regional political system. The second matrix is to assess the level of administration. The third matrix is integral – it reflects the contribution of the institutions of the regional political system in the chosen directions, and shows the impact on the established variables.

For example, let's consider the conditions for the formation of one of the matrices – the third one, which determines the final integral assessment (the rest are done by analogy, taking into account the indicators and the priority). To assess the contribution of the regional political system institutions, we will consider ensuring social efficiency as a priority.

While filling in the matrix, we will take into account the following provisions:

- 1) with an unsatisfactory assessment of both indicators, the contribution of the institutions of the regional political system is assessed as unsatisfactory;
- 2) with an increase in social efficiency ratings and lower assessments of the level of administration, priority is given to the efficiency, i.e. with good assessments of the level of social efficiency and an unsatisfactory/satisfactory assessment of the level of administration, the assessment of the "contribution" is satisfactory or good;
- similarly, with an increase in assessments of the level of administration and lower assessments of social efficiency, priority is given to this indicator and the "contribution" as a whole is assessed only as satisfactory with a negative trend;

Contribution of the Institutions of the Regional Political System to the Preservation of the Level and Quality of Life of the Population in the Context of the COVID-19 Pandemic (The Case Study of the Trans-Baikal Territory)

4) even with a high assessment of the level of administration, but a satisfactory and good assessment of social efficiency, power institutions of the regional political system will not receive an excellent assessment.

To assess the lower-level variables, we took answers to the questionnaires of the surveys conducted: among the population of the Trans-Baikal Territory and an expert survey, where state and municipal employees participated as experts.

So, we assessed the "quality of life" through the average responses to questions about the evaluation of the social sphere development (2,8548) and the security evaluation (2,8197). Interpreting these data taking into account the scale of responses (they shift to "unsatisfactory"), it is impossible, in our opinion, to talk about an ordinary "satisfactory" assessment. We will consider it as a trend towards a decrease in the quality of life.

To assess the "standard of living", we took answers to questions about changes in the level of income (50% of respondents said that income has not changed significantly, 27.8% indicated a decrease in income, 5% said that income has increased slightly since there was support from the state) and an evaluation of economic policy (the average estimate is 2,931). In this case, we can designate the state as a "stable" one.

The economic component – the standard of living – is easier to regulate and ensure than the quality of life, which, as has already been mentioned, is more subjective.

The intersection of the lower-level estimates gives us the value of the indicator for social efficiency. If we choose the standard of living as a priority, then the social efficiency of the institutions of the regional political system can be estimated at 2.884 (satisfactory) (Table 1.).

| Great | 4 | | 4 | 5 | 5 |
|------------------------------------|----------------|---------|--------------|------|-------|
| Good | 3 | | 4 | 4 | 4 |
| Satisfactory | 3 | | 3 | 3 | 4 |
| 2.931 | 3 | (2.884) | | | |
| Unsatisfactory | 2 | 2 | 2 | 3 | 3 |
| Standard of living/Quality of life | Unsatisfactory | 2.837 | Satisfactory | Good | Great |

| Table 1. Matrix with the results of assessing the social efficiency |
|---|
| of the regional political system institutions of the Trans-Baikal Territory |

Source: Own research

Assessing the level of administration – in fact, the effectiveness of the "official" activities of the regional authorities, we considered two variables that determine this indicator in the context of the research task: discretion in making decisions and timeliness.

In the case of crisis situations, in particular pandemics, in our opinion, decision-making time becomes a priority. The direct result depends on the timeliness of decisions and their implementation: the number of infected, sick people, and the level of provision with the necessary funds. Of course, decision-making discretion of regional authorities in general, and especially in the context of a pandemic, is of priority importance, since the situations in separate regions are different and measures should be taken according to the development of events with respect to regional specifics.

To assess this variable, we took the average value of experts' answers to the questions about decision-making discretion in crisis conditions by regional authorities (3,608) and the effectiveness of legislative practice during the pandemic (3,515). And the average result of the answers to the question about the timeliness of making necessary decisions became a direct assessment of this variable (3,972). Both indicators are quite close to the "good" rating. Nevertheless, in crisis situations, according to experts, decision-making time has a priority. Taking this fact into account, we assign "good" (3,744) (Table 2.) to the indicator of "level of administration".

| regional political system in the Trans-Barkar ferritory | | | | | | | |
|---|----------------|--------------|--------------|------|-------|--|--|
| Great | 4 | 4 | | 4 | 5 | | |
| Good | 3 | 3 | | 4 | 5 | | |
| 3.515 | 3 | | Good (3.744) | | | | |
| Satisfactory | 2 | 3 | | 3 | 4 | | |
| Unsatisfactory | 2 | 3 | 3 | 3 | 3 | | |
| Discretion/Timeliness | Unsatisfactory | Satisfactory | 3.972 | Good | Great | | |

| Table 2. Matrix with the results of the assessment of the level of administration for the |
|---|
| regional political system in the Trans-Baikal Territory |

Source: Own research

The final stage of assessing the contribution of the regional political system institutions (the regional authorities) is to combine the two results obtained: a satisfactory assessment of social efficiency and a good assessment of the level of administration. It is the choice of priority that is of particular importance here. The question of priority, as a scientific or research task, is debatable. In practical terms, the further development of the situation and possible adjustments depend on the priority chosen by the experts (Table 3.).

Table 3. Integral matrix for assessing the contribution of the regional political system institutions of the Trans-Baikal Territory to the studied indicators (directions)

| montations of the france Dunkar ferritory to the station materials (differences) | | | | | | | |
|--|----------------|-------|--------------|------|-------|--|--|
| Great | 4 | | 4 | 5 | 5 | | |
| Good | 3 | | 4 | 4 | 4 | | |
| Satisfactory | 3 | | 3 | 3 | 4 | | |
| 2.884 | | 3 | | | | | |
| Unsatisfactory | 2 | | 2 | 3 | 3 | | |
| Social efficiency/Level of administration | Unsatisfactory | 3.744 | Satisfactory | Good | Great | | |

Source: Own research

As a result, we state that taking into account the priority of social efficiency in relation to the level of administration, the contribution of the regional political system institutions of the Trans-Baikal Territory to the preservation of the level and quality of life of the population can only be assessed satisfactorily with a negative trend. The data obtained do not allow us to consider the regional political system as sufficiently effective one in the sphere of providing the basic conditions for the sustainable development of the region.

5. FUTURE RESEARCH DIRECTIONS

Discuss future and emerging trends. The results of this study allow us to talk about the negative impact of the COVID-19 pandemic conditions on both the level and quality of life of the population within the study region – the Trans-Baikal Territory and on the functioning of regional authorities. Despite the fact that the institutional framework of political systems is more resistant to various kinds of disturbing influences, the analysis of the results of the surveys conducted

using the dichotomous method allowed us to establish a decrease in the level of social efficiency of the authorities in the conditions of Coronavirus infection and to assess the degree of fulfillment of the functions assigned to the institutions of the regional political system as "satisfactory". Such a trend of reducing the social efficiency of regional authorities in a crisis situation is natural and ordinary not only for the subjects of the Russian Federation but also for the political-territorial entities of many other states.

Further research will be aimed at detailing the results obtained and establishing a correlation between the indicators of the level and quality of life of the population within the region and the stability of the regional political system.

6. CONCLUSION

The dichotomous analysis used in this article as the main method, despite some simplifications, allowed us to identify trends in the influence of the COVID-19 pandemic conditions on the institutions of the regional political system within the Trans-Baikal Territory and assess their contribution to maintaining the level and quality of life of the population.

The article is based on empirical data obtained through a survey of the population of the Trans-Baikal Territory and an expert survey. Therefore, the obtained results, first of all, reflect the subjective component of the categories "standard and quality of life" – they show the perception of socio-political situation in the region by the population and experts, their assessment of the contribution of the authorities to maintaining the level and quality of life. In this article, we did not set the task of correlating the indicators of official statistics with the results of the conducted surveys. It is a guideline for future studies.

ACKNOWLEDGMENT

The research was conducted on the basis of the laboratory of "Regional Management and Policy Studies" of the Trans-Baikal State University. We express our gratitude to our colleagues and the administration of the Trans-Baikal State University.

REFERENCES

- Algahtani, F. D., Hassan, S.-u.-N., Alsaif, B. & Zrieq, R. (2021). Assessment of the Quality of Life during COVID-19 Pandemic: A Cross-Sectional Survey from the Kingdom of Saudi Arabia. *International Journal of Environmental Research and Public Health*, 18(3), 847. DOI: 10.3390/ijerph18030847
- Chang, C., Moore, Sc. & Wyne, A. (In Press) Federalism in the Time of Coronavirus: A Comparative US Advantage. Despite poor performance at the national level, state and local authorities in the United States have performed admirably, *The Diplomat*, 2020, May 29. https://thediplomat.com/2020/05/federalism-in-the-time-of-coronavirus-a-comparative-us-advantage/ (Retrieved: 10.05.2022).
- Kozlova, O. A., Gladkova, T. V., Makarova, M. N. & Tukhtarova, E. Kh. (2015). Methodological Approach to Measure the Quality of Life of the Region's Population. *R-Economic*, 2, 280-290.
 DOI 10.1502(1000) 2015 2.011

DOI:10.15826/recon.2015.2.011

- Mau, V. A., Idrisov, G. I., Kuzminov, Ya. I., Radygin, A. D., Sadovnichiy, V. A., Sinelnikov-Murylev, S. G. (Ed.). (2020) Society and the Pandemic: Experience and Lessons from the Fight against COVID-19 in Russia, Moscow.
- McCoy (2020), "What exactly is the government's coronavirus strategy?", *Queen Mary University of London*, https://www.qmul.ac.uk/media/news/2020/pr/what-exactly-is-the-governments-coronavirus-strategy.html. (Retrieved: 10.05.2022).
- OECD (2020, November 10). *The territorial impact of COVID-19: Managing the crisis across levels of government.* https://www.oecd.org/coronavirus/policy-responses/the-territorial-impact-of-covid-19-managing-the-crisis-across-levels-of-government-d3e314e1/ (Retrieved: 10.05.2022).
- Osipov, G. V. (Ed.)., Levashov, V. K. (Ed.). (2020). Russian society and the State in a pandemic: socio-political situation and demographic development of the Russian Federation in 2020: Collective monograph, Moscow: ITD "PERSPEKTIVA". DOI: 10.38085/978-5-905790-48-5-2020-1-532
- Qiu, J., Shen, B., Zhao, M., Wang, Z., Xie, B. & Xu, Y. (2020). A nationwide survey of psychological distress among Chinese people in the COVID-19 epidemic: Implications and policy recommendations. *General Psychiatry*, 33(2):e100213. DOI: 10.1136/gpsych-2020-100213
- Shakleina, M. V., Volkova, M. I., Shaklein, K. I. & Yakiro S.R. (2020). Theoretical and methodological problems of measuring social comfort: results of empirical analysis based on Russian data. *Economic and Social Changes: Facts, Trends, Forecast, 13*(5), 135–152. DOI: 10.15838/esc.2020.5.71.8
- Zhade, Z., Shadzhe, A., Kukva, E., Lyausheva, S. & Ilyinova, N. (2021). Regional Power and Society in the Context of the COVID-19 Pandemic: Indicators of Integration, *Advances in Social Science, Education and Humanities Research*. Paper presented at 8^h International Scientific and Practical Conference "Current problems of social and labour relations" (ISPC-CPSLR 2020), January, 2021, volume 527 (pp. 736-741). Atlantis Press. DOI: 10.2991/assehr.k.210322.208



COVID-19: A Story Told by Women Entrepreneurs about How to Survive by Turning a Crisis into an Opportunity

Tudor Tarbujaru¹ ⁽¹⁾ Petruta Ionela Stanca² ⁽²⁾

Received: June 27, 2022 Accepted: August 30, 2022 Published: December 30, 2022

Keywords: Women entrepreneurs; COVID-19; Challenges; Crisis management; Financial pressure

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-Non-Commercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission. Abstract: The Covid-19 epidemic has generated an environment full of confusion and instability among global economies. It has put enormous strain on the governments and the economies, both on an international and national scale. No single day goes by without news about the consequences of the Covid-19 pandemic's devastating economic, health, and emotional well-being. The intensity of its burnt is unprecedented since it is a humanitarian crisis that has generated shock to the global economic system. The crisis had a significant influence on economic activities all over the world, but on the other hand, it boosted the presence of women as entrepreneurs, resulting in a shift in the demographic features of business and economic processes. This research investigated how women entrepreneurs had to deal with all of this during Covid-19. The study exemplifies the difficulties that a female entrepreneur faces as a result of the crisis. Furthermore, it sheds light on the strategies used by female entrepreneurs to deal with the impacts of the crisis in order to safequard their businesses. The focus of the paper is also on the associated obstacles, challenges, and potential solutions for the emergence of female entrepreneurs in this pandemic era. We provide a paradigm to aid in making sense of how gender and entrepreneurship research is currently framed and positioned. The possibilities in disguise and future potentials that will alter women's choices and talents following the Covid-19 outbreak have been briefly examined. Women entrepreneurs have the potential to convert the Covid-19 critical point into an even better opportunity. This study is unique in that it will serve as the foundation for many future studies that will strive to investigate further the status of difficulties and opportunities for women entrepreneurs to manage and sustain their entrepreneurial endeavors effectively.

1. INTRODUCTION

When holistically considered, Covid-19 has presented a real challenge to both established and small businesses. This pandemic was a total outlier, and it came at a point when it was unprecedented, and so were the consequences. The impact of the Covid-19 pandemic rocked all sectors of society, but the economic systems were the most affected due to the forced lockdowns and social restrictions that hampered the coordination of the usual flow of business activities (Mo et al., 2020). Ultimately most businesses were brought to their knees as they could not sustainably keep the workforce while operating with limited customers as most people went indoors in fear of contracting the virus. However, some like Amazon and other e-commerce companies recorded a historically high increase over the same period (Bhadury, 2017).

This underscores the point the pandemic was a redefining point in history in which the roles and positions of businesses in the global market were recalibrated. This had the same impact on women's run businesses which had to structure new mechanisms and a fresh framework for dealing with the downturns of the pandemic. Therefore, the pandemic was a real redefining

¹ University of Craiova, Craiova, Romania

² National Institute of Research Development and Testing in Electrical Engineering, Craiova, Romania



moment for all businesses. Within these critical moments, big exchanges occur, favoring the ones that adopt suitable strategies for combating the challenge. Concerning the latest Covid-19 cases, it marked a period in which businesses were tried and tested. This was also a great opportunity for the women-led business, which goes the same opportunity alongside other businesses to fight and thrive.

Women's entrepreneurship is a vehemently debated academic issue, with scholars vying to provide more to the increasing body of knowledge. To many economic players, particularly entrepreneurs, the burden of lockdown on significant sectors of society and economic and social life has come as a shock. The worldwide lockdown caused by the epidemic, which forced people all around the world to stay home, also revealed the creative and inventive spirit. With the epidemic causing financial pressure in every home, women have been forced to discover new ways of making extra money. These women have worked hard to fulfill themselves, not only through marriage and motherhood but also by starting businesses and earning money to support themselves and their families, with the most difficult constraining factor being the dual role in home and business. Uncertainty produces worry and fear, and, at times, denial. This research aims to emphasize what these developments signify for female entrepreneurs and their current position in a changing society by pinpointing specific significant changes that society has gone through over the last decades.

Women have responded to the Covid-19 pandemic in diverse ways. The strategies that they applied in surviving through the pandemic are major determinants of how they are likely to respond in the ensuing years hence the maturity that has been realized in women's business. The challenges of the obstacles presented by women have been a real test of the strength of problem-solving and the innovative levels among women-owned businesses (Kareska, 2016). Indeed, these businesses have come of age regarding the coping mechanisms they applied. This paper shall comprehensively look at the strategies that most of these businesses applied during the pandemic period, thus laying the ground for determining the maturity levels and the competence of these businesses in countering future uncertainty.

2. LITERATURE REVIEW

This section shall be related to a review of the available literature related to women's struggle in business over the Covid-19 period. It shall, therefore, give a detailed account of the methods and approaches that were applied and that could have been applied, and those that can be applied in addressing future uncertainty.

According to Ogundana et al. (2021), women operate in an unfriendly business environment which is majorly characterized by corruption, lawlessness; incoherent policies, lack of government transparency, poor infrastructure, disproportionate taxes, lack of legislation for supporting property ownership rights among women and poor infrastructure among other factors which are detrimental to business strength and stability (Cabeza-García et al., 2018). This presents an unfriendly working environment for women to operate on, especially in times of crisis when fast and sweeping decisions are essential for keeping the businesses pertinent to the presented situation (Bakare & Lawal, 2020). Moreover, this provided limited chances for navigating through the problem, and therefore, survival depends on the strength of the operators' underlying strategy in coping with the challenge (Ogundana et al., 2021). In parallel, considering the women have the responsibility of providing care to the children contributed to the challenges they encountered in remaining relevant to the situation of the day (Syedda, 2018).
Therefore, the striking differences in the operating environment where developing countries have palpable institutional voids and sociocultural values that undermine the well-being of businesses contribute to the operational differences when compared to their counterparts in developed countries who have well-structured policies and a better operational environment (Cherrier et al., 2018). Even though various studies have been conducted on the factors that determine the growth and survival of women-operated businesses, most of these have lacked a theoretical framework; thus, it has been challenging to arrive at a reliable conclusion (Hussain et al., 2019). Therefore, the lack of theoretical conceptualization resulting from inadequate research represents some of the hurdles experienced in understanding the real scope of women's challenges. Moreover, it presents researchers as they are no basis for the formulation of a hypothesis. Therefore, the lack of prior studies examining the plight of women is a major contributor to the current lack of policy for supporting women's presence in business.

However, this research has been able to identify some of the factors that determine women's success and presence in business. Some of these include money access and usage within their premises (Stefan et al., 2021). Therefore, the exploration of the diverse methods of capital usage lays the ground for identifying the strengths and weaknesses of businesses in developing countries and how the innate factors in the aboriginal ground set the pace for the trajectory of their businesses (Barrachina et al., 2021). However, there is still a major research gap in determining how women utilize their capital regarding the stated challenges related to the lack of well-established policies and institutional voids. Furthermore, education is among the leading determinants of the performance of women's businesses. This study indicates that women's ability to operate effectively in businesses increases up the curve depending on their levels of education, with the better-educated ones possessing better capabilities for dealing with problem-solving challenges (Grandy et al., 2020). Additionally, the study unravels the impact of the market as a determinant of the strength of women's entrepreneurship. Finally, it explores the impact of motherhood and other factors in the macro-environment and their role in determining business success.

Dzisi (2021) explores the progress of women in social entrepreneurship through the lenses of women's involvement in the construction industry in Ghana. The study found that innovative enterprises are becoming quite common and attracting more people in the modern world. This provides the basis for examining women's prevalence and participation in such opportunities. This study reveals that there is an increasing number of women entrepreneurs in social entrepreneurship. It also reveals that socio-entrepreneurship is an integral and important component of society as it assists in generating fresh and innovative ideas to solve societal problems (Dzisi, 2021). The author finds out that social enterprises provide men with better incomes which give them a leg up in home decisions making, thus influencing the affairs within their homes. Therefore, this creates a vicious cycle in gender matters; thus, men have a higher predilection for deciding whether women join social entrepreneurship. This research has identified that women's involvement can be useful in the diversification of income while adding to the available innovation in the market (Rao & Giri, 2021). Therefore, this may assist in minimizing the poverty levels, thus establishing more stable communities. In this study which specifically focused on the construction industry, it was realized that women's absence or low participation in entrepreneurship had been influenced by low technical skills, shortage of adequate funding for underpinning entrepreneurship activities, and the presence of social and cultural limitations (Scur et al., 2021).

Even though this section has explored the available literature evidence pertaining to women's involvement in entrepreneurship, there is limited theoretical literature related to the study topic, which has hampered the realization of comprehensive research related to the topic. Therefore, further research is needed to lay the ground for more research related to the subject. 8th International Scientific Conference ERAZ 2022 Selected Papers

3. METHODOLOGY

This paper utilized the available literature pertaining to women's involvement in entrepreneurship activities. Some of the selected materials are the ones that were based on empirical research, while the others are literature reviews of the available research works. These materials were selected from credible sources, thus allowing for the drawing of relevant conclusions. Moreover, the selected studies are from different years in history, which was useful in the assessment of the evolution and advancement that has been achieved in relation to the subject. Similar, a review of secondary research was useful in this study in allowing the incorporation of numerous scholarly perspectives, thus identifying the common points between the research. In parallel, this was useful in gaining high-quality research within a short time frame, thus paving the way for reducing the research duration.

4. **RESULTS**

Innumerable obstacles in today's culture stand in the way of the success of enterprises run by women. According to the research that has been done, very few policies and programs are designed to support the presence of women in businesses. This issue is more prevalent in developing economies due to institutional voids, poor governance, corruption, and limited legislation that allow women to own properties. In addition, because women are expected to fulfill their responsibilities as mothers, they require a healthy work-life balance if they are to continue to be productive members of the workforce. As a consequence of this, women faced a greater number of obstacles in the face of the Covid-19 pandemic (Mahat et al., 2021; Boiral et al., 2021). This was an additional deficiency to the one that already existed, which further restricted their ability to carry out their roles. As a result, they were forced to resort to a variety of coping mechanisms in order to survive the dangers posed by the epidemic, which put them in a stronger position to prevail over the obstacles that were placed in front of them (Strielkowski et al., 2016). In addition, women face additional barriers that prevent them from effectively participating in the operations of businesses. These barriers include things like a lack of access to financial resources, an inadequate grasp of the critical technical skills required for successfully operating a business and sociocultural beliefs that discourage the participation of women in activities that result in financial gain (Kabukuru & Afande, 2016). In addition to this, males play a significant role in the financial responsibilities of their families and make up the majority of the population in these countries. As a result, they have a propensity to exercise control over significant decisions, such as determining whether or not their wives should join in the business.

5. **DISCUSSION**

According to current research findings, women were at a significant disadvantage when attempting to deal with the challenges posed by the recent epidemic. It's possible that this was because our culture has a limited number of institutional frameworks for dealing with women's issues in business. As a result, they were required to devise methodical coping techniques to tackle the challenges posed by the pandemic while continuing to fulfill their responsibilities to society (Risley, 2020). Women who lived in countries with lower levels of economic development carried a greater share of the load. In these countries, women frequently lacked access to essential resources, such as financial backing to support their academic enterprises. Despite the fact that women are faced with difficult obligations and many obstacles that prevent them from participating, their presence in the workforce is strong. In addition to this, women have been actively participating in the creation of jobs and the expansion of the economy business (Sangem, 2020). Studies have indicated that the participation of women in business is a key factor in overall economic growth (Melkani et al., 2021). Even though there is just a little amount of research available on women's roles in the expansion of corporations, their contributions to economic growth can nevertheless be felt. In most cases, a lower participation rate for women in the labor force results in a slower rate of economic growth since fewer individuals are put to work. As a result, women play a dual role in the growth of the economy through engagement in activities that create money and the supply of job opportunities to the population (Burrell, 2021).

As a result, of the global economic crisis that began in 2008, there was a corresponding increase in the number of unemployed people, which suggested that women play an essential role in providing work opportunities for a sizeable section of the population. Women's participation has been greatly endangered due to the pandemic that has wreaked havoc on the world's economy. Consequently, ingenuity has been sparked, which some people have harnessed to enable themselves to withstand the existential threat created by the pandemic. As a direct consequence of this, the spirit of entrepreneurship has been reignited, which has led to an increase in the number of new opportunities and ideas on the market.

In order to succeed in the business world, one must engage in appropriate and thorough planning in order to respond appropriately to the Covid-19 problem. As a result, female business owners responded to the pandemic by coming up with additional ways to generate cash by capitalizing on the chances the scenario gave (Afshan et al., 2021; Matharu et al., 2021). In the same way that Toyota and Tesla reorganized in order to manufacture ventilators after widespread movement hindrances inhibited the movement of people and led to lower demand for vehicles, some women were innovative and embarked on providing the essentials required to prevent the disease from spreading. Some people could only scrape together enough money to start manufacturing and selling facemasks, detergents, and sanitizers to support themselves (Mustafa et al., 2021). This demonstrates the importance of women's actions to increase their chances of surviving the pandemic's effects (Ayatakshi-Endow & Steele, 2021). The implementation of these policies has enabled women to amass significant financial resources.

Women have a number of options available to them to help them triumph over the challenges they face. This can be accomplished by giving them access to business management programs that will help them achieve the competence that will allow them to function more effectively in the environment in which they are currently operating. In addition, there is a multitude of funding programs available, some of which are specifically designed to support the entrepreneurial endeavors of women. In addition, women are able to prevail over such challenges if they adopt a collaborative strategy that involves coordinating their efforts to build larger firms that are eligible for funding and that make it possible for employees to share responsibilities. Because of this, they will be able to more easily attain the goals they have set for themselves, thereby overcoming the obstacles that currently exist in their firm. In addition, working together can be helpful in the creation of an environment in which they can share their experience, thereby boosting their capability to run businesses successfully. Parallel to this; they can easily be registered as differentiated corporates when operating together, so addressing the sociocultural problems that hinder the development of women-led firms. Most notably, the 21st century has experienced massive technical developments; as a result, the current technologies carry a huge potential for streamlining commercial operations through effective financial management, stock tracking, and progress predicting enterprises (Intawong et al., 2021). Therefore, all of these components

8th International Scientific Conference ERAZ 2022 Selected Papers

can be skillfully put together to realize a conducive work atmosphere that can allow businesses to leverage the available opportunities in the market to take advantage of the supply gaps afforded by the pandemic. When travel is limited, one of the choices that can be made is to offer delivery services directly to customers' homes. Alternately, persons knowledgeable in nursing can offer home care and mobile medical services to those suffering throughout the Covid-19 period. This can be done in order to assist those who are affected by the disease.

6. CONCLUSION

Tersely, the preceding sections have provided an elaborate and thoughtful analysis of women's coping strategies over the Covid-19 period. Moreover, it has highlighted the research gaps and the leading issues that affect women's performance and participation in entrepreneurship, especially in developing countries. It has been established that women in developed countries have the upper hand regarding their counterparts in developing countries, especially in Africa, where institutional gaps and poor policy formulations for supporting women's involvement in business activities prevail. In parallel, it has been discovered that women's success in entrepreneurship ventures hinges on various factors ranging from sociocultural beliefs, access to capital, and the level of education, among other less significant elements. All these contribute to determining women's involvement as they shape the market, thus shifting it in favor of some players. Moreover, corruption and the absence of policies are among the other contributors to market inequalities. Therefore, the paper has reflected the true position and state of women's participation in business concerning the Covid-19 pandemic. This has laid the basis for pinning down the possible remedies and coping mechanisms that can be useful in enabling women to deal with future uncertainty more decisively.

REFERENCES

- Afshan, G., Shahid, S., & Tunio, M. N. (2021). Learning experiences of women entrepreneurs amidst COVID-19. *International Journal of Gender and Entrepreneurship*; ahead-of-print(ahead-of-print):25.
- Ayatakshi-Endow, S., & Steele, J. (2021). Striving for balance: women entrepreneurs in Brazil, their multiple gendered roles and Covid-19. *International Journal of Gender and Entrepreneurship*.
- Bakare, A., & Lawal, Y. (2020). Women In Entrepreneurship Lessons From Covid-19 In Lagos State.
- Barrachina Fernández, M., García-Centeno, M. D. C., & Calderón Patier, C. (2021). Women sustainable entrepreneurship: Review and research agenda. *Sustainability*, 13(21), 12047. DOI: 10.3390/su132112047
- Bhadury, A. (2017). E-Commerce: A silent revolution in women empowerment. *International Journal in Management & Social Science*, 5(3), 18-26. DOI: 10.5281/zenodo.569530
- Boiral, O., Brotherton, M. C., Rivaud, L., & Guillaumie, L. (2021). Organizations' management of the COVID-19 pandemic: a scoping review of business articles. *Sustainability*, 13(7), 3993. DOI: 10.3390/su13073993
- Burrell, D. N. (2021). Managerial adaptability and business strategic change in age of COV-ID-19. *PSU Research Review*, (ahead-of-print).
- Cabeza-García, L., Del Brio, E. B., & Oscanoa-Victorio, M. L. (2018). Gender factors and inclusive economic growth: The silent revolution. *Sustainability*, 10(1), 121. DOI: 10.3390/ su10010121

- Cherrier, H., Goswami, P., & Ray, S. (2018). Social entrepreneurship: Creating value in the context of institutional complexity. *Journal of Business Research*, *86*, 245-258. DOI: 10.1016/j.jbusres.2017.10.056
- Dzisi, S. (2021). Women Social Entrepreneurship In Ghana And The Sustainable Development Goals.
- Grandy, G., Cukier, W., & Gagnon, S. (2020). (In) visibility in the margins: COVID-19, women entrepreneurs and the need for inclusive recovery. Gender in Management: An International Journal.
- Hussain, N., HAQUE, A. U., & Baloch, A. (2019). Management theories: The contribution of contemporary management theorists in tackling contemporary management challenges. *Yaşar Üniversitesi E-Dergisi*, 14, 156-169. DOI: 10.19168/jyasar.635061
- Intawong, K., Olson, D., & Chariyalertsak, S. (2021). Application technology to fight the COVID-19 pandemic: Lessons learned in Thailand. *Biochemical and biophysical re*search communications, 534, 830-836. DOI: 10.1016/j.bbrc.2020.10.097
- Kabukuru, A., & Afande, F. O. (2016). Analysis of challenges faced by women entrepreneurs in accessing finance in Kenya (A case of beauty sector in Nairobi central business district). Journal of Poverty, Investment and Development, 24(8).
- Kareska, K. (2016). Challenges in modern management and modern business that Macedonian organizations face in gaining competitive advantage. *Journal of Economics*, 1(2).
- Mahat, N., Mahat, I. R. B., & Mustafa, M. S. A. (2021). Covid-19 pandemic: issues and challenges among women entrepreneurs in Malaysia. *International Journal of Academic Research in Business and Social Sciences*, 11(6), 231-239. DOI: 10.6007/IJARBSS/ v11-i6/10114
- Matharu, S. K., & Juneja, D. (2021). Factors impacting resilience of women entrepreneurs in India in the face of COVID-19. Vision, 09722629211043299.
- Melkani, E., Singh, K., & Mehta, M. (2021). Productivity of rural women through entrepreneurship intervention. *International Journal of Education and Management Studies*, 11(1), 7-12.
- Mo, G., Cukier, W., Atputharajah, A., Boase, M. I., & Hon, H. (2020). Differential impacts during COVID-19 in Canada: A look at diverse individuals and their businesses. *Canadian Public Policy*, 46(S3), S261-S271. DOI: 10.3138/cpp.2020-072
- Mustafa, F., Khursheed, A., Fatima, M. and Rao, M. (2021), "Exploring the impact of COV-ID-19 pandemic on women entrepreneurs in Pakistan", *International Journal of Gender and Entrepreneurship*, Vol. 13 No. 2, pp. 187-203. DOI: 10.1108/IJGE-09-2020-0149
- Ogundana, O. M., Simba, A., Dana, L. P., & Liguori, E. (2021). Women entrepreneurship in developing economies: A gender-based growth model. *Journal of Small Business Management*, 59(sup1), S42-S72. DOI: 10.1080/00472778.2021.1938098
- Rao, S., & Giri, V. N. (2021). Impact of Innovative Culture on Organizational Effectiveness: The Ediating M Role of Social Media. *Journal of Engineering Education Transformations*, 35(2), 29-45. DOI: 10.16920/jeet/2021/v35i2/22069
- Risley, C. (2020). Maintaining performance and employee engagement during the COVID-19 pandemic. *Journal of Library Administration*, 60(6), 653-659. DOI: 10.1080/01930826.2020.1773716
- Sangem, M. (2020). Challenges for Women Entrepreneurs in the Wake of COVID-19 Pandemic. *Journal of Interdisciplinary Cycle Research*, 12(11), 279-284.
- Scur, D., Sadun, R., Van Reenen, J., Lemos, R., & Bloom, N. (2021). The World Management Survey at 18: lessons and the way forward. Oxford Review of Economic Policy, 37(2), 231-258. DOI: 10.1093/oxrep/grab009

- Stefan, D., Vasile, V., Oltean, A., Comes, C. A., Stefan, A. B., Ciucan-Rusu, L., ... & Timus, M. (2021). Women entrepreneurship and sustainable business development: Key findings from a SWOT–AHP analysis. *Sustainability*, *13*(9), 5298. DOI: 10.3390/su13095298
- Strielkowski, W., Shishkin, A., & Galanov, V. (2016). Modern management: beyond traditional managerial practices. *Polish Journal of Management Studies*, 14.
- Syedda, M. S. (2018). Exploring the strategies and challenges to empower female entrepreneurs via business support and ICT in the UK: a resource-based theory perspective (Doctoral dissertation, Cardiff Metropolitan University).



The Role of Women in Founding Top Startups Worldwide

Katia Giusepponi¹ D Ernesto Tavoletti²

Received: June 14, 2022 Accepted: July 22, 2022 Published: December 30, 2022

Keywords: Startup; Women co-founders; Founding team; Role of women in startups

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-Non-Commercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission. **Abstract:** The work investigates the role of women as co-founders of globally successful startups, particularly their impact on the company's mission and vision. The authors' initial research on entrepreneurial and organizational profiles in top startups worldwide underlined that the presence of women as co-founders of globally successful startups is very limited (Giusepponi, Tavoletti & Tosi, 2019). The study presented here highlights that a female presence in the founding team supports the sharing of the vision and the effectiveness of the mission, and these elements are reflected in the organizational culture, specifically in the sense of corporate belonging and cohesion. This study is based on secondary sources derived from advanced analyses in terms of connections between variables within the startups studied. From an advancement perspective, the research could be deepened by studying other impact profiles and, considering that it is here limited to 109 top global startups, extended to other entrepreneurial realities.

1. INTRODUCTION

Women are increasingly playing a leading role in international business but the public debate among practitioners is focused on careers inside MNEs, work-life balance in large organizations, and the percentage of women in top hierarchical positions. There is not the same attention to the role of women in top startups worldwide and their entrepreneurial contribution to the founding and growth of those global players: how frequently are women part of the founding team? What is the benefit of their role and contribution?

Gender studies in business have focused mainly on the presence of women as CEOs and corporate officers in MNEs and well-established leading Fortune 500 companies (Catalyst, 2019a, 2019b). While academic research on women's entrepreneurship is extensive (Jennings & Brush, 2013; Cardella et al., 2020), limited attention is paid to the role of women in founding teams of globally successful startups and international new ventures. The words "startups" and "ventures" are entirely and significantly missing from the World Economic Forum Global Gender Gap Report (2018, 2021) despite the growing interest of the international business academic community in international new ventures (Vaccarini et al., 2017).

Our recently published research on entrepreneurial and organizational profiles in top startups worldwide fills that gap (Giusepponi, Tavoletti & Tosi, 2019). This study reveals that the presence of women as co-founders of globally successful startups is as limited as in top positions of MNEs and leading Fortune 500 companies. We underline here that, despite being rare, the presence of women has a positive impact on vision, mission, and organizational cohesion.

² University of Macerata – Department of Political Sciences, Communication and International Relations, Piazza Strambi 1, 62100 Macerata, Italy



¹ University of Macerata – Department of Education, Cultural Heritage and Tourism, P.le Bertelli 1, c.da Vallebona, 62100 Macerata, Italy

2. RESEARCH DESIGN AND METHODS

The research covered 109 startups valued at 1.1 billion US dollars or more according to the list available at The Billion Dollar Startup Club (Austin, Canipe & Slobin - Dow Jones Venture Source and The Wall Street Journal; access: 2017).

The list only includes privately held companies that have raised funds in the past four years with at least one venture capital firm among their investors (Ibidem). The list includes famous names such as Uber, Lyft, Airbnb, Dropbox, BlaBlaCar, Qualtrics, Pinterest, and many others. Distinguishing the gender of the co-founders was possible for all but 2 of the 109 companies surveyed. Among the 107 valid elements, 97 companies were found to have only male co-founders and only 10 with at least one female co-founder. The startups with at least one woman co-founder are located in California (6), New York (1), Singapore (1), China (1), and Germany (1).

The considered startups were analyzed for what concerns their strategic perspective, with attention to different profiles, particularly focusing on the *level of shared vision*, the *effectiveness of mission* and the *organizational culture as a sense of belonging*. These profiles were observed and evaluated based on online information and resources: companies' websites, the Dow Jones Venture Source, The Wall Street Journal, LinkedIn, Crunchbase Pro, Bloomberg, Glassdoor, AngelList, Quora, interviews released by the founders, and other corporate actors. However, the assessment of the organizational sense of belonging has been made based on what emerged from the statements of employees and other corporate actors. These statements were drawn mainly from online sources other than companies' websites.

| SCORE | 0 | 1 | 2 | 3 | 4 | 5 |
|---------|-----------------------------------|----------|----------|----------|----------|----------------------------------|
| MEANING | (worst performances) highly | widely | slightly | slightly | widely | (best performances) highly |
| | | negative | | | positive | |

 Table 1. Rating scale

Source: Own elaboration

The evaluation was carried out using a 0 to 5 rating scale as shown in table 1. This even (vs odd) scale was chosen because it requires analysts to choose between a positive or negative intermediate value (that is between "slightly positive" and "slightly negative" in our case), avoiding the risk of non-choice potentially linked to a unique central value. The evaluation was based on a process of rigorous comparison. Within the sample, 0 was given for the worst performances and 5 for the best; the other scores were attributed following a comparative and differential approach. Two researchers cross-checked their independent evaluations until a consensus had been reached to assure homogeneous and consistent evaluations (Giusepponi, Tavoletti & Tosi, 2019). The assessment was done when the sub-groups of startups with women co-founders had not been identified, and gender issues were not the main purpose of the research.

3. FINDINGS

Out of the total 234 co-founders (men and women), 11 women (4.70%) and 223 men (95.30%) have been observed. The figure concerning women is dramatically low, especially considering that we are studying top startups worldwide; however, these numbers are in line with the 5% of women (25) currently holding CEO positions at SP 500 companies (Catalyst, 2019b) and the

4.7% Fortune 500 board leadership position with women (Catalyst, 2019a). As the low figure refers to co-founders, it cannot be explained by a preexisting masculine organizational culture as for MNEs or Fortune 500 companies. The reason for the very low number of female co-founders has to be searched in the business environment outside corporations, in startups success drivers, investors' preferences, and gender-differentiated motivations for launching a startup.

The *level of shared vision* concerns the strength of the corporate vision as stated in companies' websites, strategic lines, and interviews with the founders and other actors. The *effectiveness of the mission* regards the incisiveness, characterization, and strength in terms of the orientation of the actions shown by the mission as declared and disseminated online. The *sense of belonging* is observed here as a reason and result of individual commitment and involvement in the organization and studied as a fundamental factor of organizational cohesion and culture.

| | Level of shared vision (average) | Effectiveness of mission (average) | Organizational culture as a sense of belonging (average) |
|------------------------------------|-------------------------------------|---------------------------------------|--|
| Startups with women co-founders | 4.22 | 4.67 | 3.50 |
| Startups without women co-founders | 3.04 | 3.86 | 2.87 |
| All startups considered | 3.14 | 3.93 | 2.89 |

Table 2. Level of shared vision and effectiveness of mission: mean values

Source: Own elaboration

Based on our results, 10 companies with women co-founders expressed an average *level of shared vision* of 4.22 (vs average values of 3.14 for all the startups, and 3.04 for startups with only men as founders). They also expressed average *effectiveness of the mission* of 4.67 (vs average values of 3.93 for all the startups, and 3.86 for startups with only men as founders). In terms of a *sense of belonging*, they expressed an average of 3.50 (vs average values of 2.89 for all the startups, and 2.87 for startups with only men as founders) (table 2).

| | number of women co-founders | level of shared vision | effectiveness of mission | organizational culture as a sense of belonging |
|--|--------------------------------|---------------------------|-----------------------------|--|
| number of women co-founders | 1 | 0,22** | 0,20** | 0,17* |
| level of shared vision | 0,22** | 1 | 0,54*** | 0,45*** |
| effectiveness of mission | 0,20** | 0,54*** | 1 | 0,17* |
| organizational culture as a sense of belonging | 0,17* | 0,45*** | 0,17* | 1 |

Table 3. Pearson correlation coefficient (ρ)

Significance: * P<0.10; ** P<0.05; *** P<0.01

Source: Own elaboration

The number of women co-founders can be observed in positive correlation both with the *lev-el of shared vision* ($\rho = 0.22$) and with the *effectiveness of the mission* ($\rho = 0.20$). Therefore, a positive role of women co-founders emerges in the strategic contextualization and finalization. Furthermore, their contribution to strategic design and communication, in terms of sharing and focusing, also suggests possible positive links with organizational cohesion. In fact, our data reveal that the number of female co-founders positively correlates with an organizational culture that emphasizes a sense of belonging to the startup – highlighting a statistically significant Pearson correlation coefficient, equal to 0.17 (table 3). This result is of critical importance for

practitioners and managers who are committed to supporting and advising a founding team, as there is evidence that the presence of women adds value to the organization, with special reference to cohesion, shared vision, and effectiveness of the mission.

Furthermore, the fact that the strategic profiles of the mission ($\rho = 0.17$) and, above all, of the vision ($\rho = 0.45$) are in positive correlation with the sense of belonging (table 3) was also observed. It is thus possible to conclude that a female presence in the founding team supports the *level of shared vision* and the *effectiveness of mission*, and these elements are reflected in the organizational culture, specifically in the sense of corporate belonging and cohesion (figure 1).



Figure 1. Women co-founder, strategic orientation and sense of belonging Source: Own elaboration

The implications for managerial practice cannot be underestimated: the involvement of women in the founding team is beneficial for developing corporate belonging and cohesion. Some representative examples of female co-founders of globally successful startups are provided below.

Nichole Mustard, who went from being a trainee manager at Pizza Hut to being a co-founder of Credit Karma (a \$4 billion startup making consumer credit scores easier for people to see and monitor), is representative of a contribution to organizational cohesion. She underlines the importance of motivation in achieving challenging objectives, and she highlights how having not only financial perspectives is important for sustaining risks and constant changes: «"I love people," [...] "there were people I wanted to be with," she said of her cofounders. "To do this in a way that's really actually helping consumers and not just making money, that feels great."» (Bort, 2018).

Adi Tatarko, co-founder of Houzz (a startup that helps people remodel their homes based on her individual experiences), is an example of determination in building a shared vision, including letting people go when they do not fit the company's vision and goals: «"When someone is not a good fit, it's better to let go sooner rather than later. [...] *It's all about the people. The people that help us build Houzz all over the world day-to-day and the people that are part of the Houzz community*"» (Krishna, 2017).

Amy Pressman, co-founder of Medallia (a startup leader in customer experience management software), is an example of effectiveness of mission for her ability to promote a culture of direct connection between customers and employees: «If you want to help employees thrive, don't focus on perks, being "hot," or buzzwords. Instead, connect employees to the impact of their work. [...] *In my experience, a great way to help employees understand their impact is to connect them to customers.*» (Weiner, 2018).

Tan Hooi Ling, co-founder of Grab (Southeast Asia's mobile technology company that connects consumers to drivers and businesses), is representative of the importance of focusing on strategic orientation through a continuous problem-solving approach and a constant drive towards enhancement: «As many Grab employees would attest, it is not uncharacteristic of Ling to jump into "problem-solving" mode. At the company she oversees strategy, people operations and new business, spending much of her work day in meetings to analyze how best to take the company forward» (Soo, 2019).

Jessica Schultz, co-founder of HelloFresh (a global recipe kit delivery service), underlines the importance of a «big vision», beyond the fear of failure, because it is «a key to continuous improvement» and «you are only learning when you don't see what lies ahead» (Northzone, 2019).

4. DISCUSSION AND CONCLUSIONS

By revealing that the number of women co-founders in globally successful startups is very similar to the number of CEOs in SP 500 companies, our research removes any easy-escape explanation to be found in the organizational culture of big businesses, confirming that the obstacles women face in their careers are embedded in business environments external to business organizations, as well as in social ecosystems worldwide (Berger & Kuckertz, 2016; Bullough et al., 2022).

The work positions itself in the management literature that provides evidence about the enriching role of women in startups (Berger & Kuckertz, 2016), their limited presence (Kanze, 2018), and the obstacles they face in getting funding (Kuschel et al., 2017). It contributes to this available literature by focusing on the presence of women in the founding team of the 109 most successful startups globally; the positive role of women in strategic design, communication, and the consequences for the development of organizational culture with a sense of belonging and cohesion. Therefore, it integrates and explains the reported evidence that, despite being a minority, «businesses founded by women ultimately deliver higher revenue—more than twice as much per dollar invested—than those founded by men, making women-owned companies better investments for financial backers» (Abouzahr et al., 2018). It also provides a different picture of the top successful global startups concerning previous regional studies that reported slightly less strong skill of strategic vision among women, compared to men (Kuschel, Labra & Díaz, 2018).

The results have strong implications for practitioners and managers because, by revealing that the presence of women in global startups is as low as in MNEs and Fortune 500 companies, they remove the widespread idea that gender discrimination is limited to hierarchy and big corporations' masculine culture. On the opposite, the limited presence of women in the founding teams of globally successful new ventures shows that the issue extends to markets and business environments. No one is exempted from taking affirmative action, not even practitioners with limited responsibility. That is because we reveal that the presence of women is beneficial for cohesion, corporate culture and sense of belonging, shared vision and effectiveness of mission in a very critical area for world innovation and future prosperity: the founding of the most globally successful startups.

As our sample is limited to the 109 most globally successful startups, we hope that our framework and analysis will be theoretically replicated, as well as regarding the specific industry and national settings to better understand which skills more frequently observed in women are particularly relevant to these perspectives. Furthermore, improving the analysis through interviews directly addressed to founding teams and their associates would be useful for more robust research.

REFERENCES

Abouzahr, K., Taplett, F. B., Krentz, M., & Harthorne, J. (2018). Why Women-Owned Startups Are a Better Bet. *BBG The Boston Consulting Group*.

AngelList, at https://angel.co.

- Berger, E. S., & Kuckertz, A. (2016). Female entrepreneurship in startup ecosystems worldwide. *Journal of Business Research*. 69(11), 5163-5168. Available at SSRN: https://ssrn. com/abstract=2779709
- Bloomberg, at https://www.bloomberg.com.
- Bort, J. (2018). How this woman went from a Pizza Hut employee to a founder of a \$4 billion startup. *Business Insider*. 12 August, https://www.businessinsider.com/credit-karmanichole-mustard-interview-2018-7?IR=T (Accessed: July 28, 2019).
- Bullough, A., Guelich, U., Manolova, T. S., & Schjoedt, L. (2022). Women's entrepreneurship and culture: gender role expectations and identities, societal culture, and the entrepreneurial environment. *Small Business Economics*, 58(2), 985-996. DOI: 10.1007/ s11187-020-00429-6
- Cardella, G. M., Hernández-Sánchez, B. R., & Sánchez-García, J. C. (2020). Women entrepreneurship: A systematic review to outline the boundaries of scientific literature. *Frontiers in Psychology*, 11, 1557. DOI: 10.3389/fpsyg.2020.01557
- Catalyst (2019a). Missing Pieces: The 2018 Board Diversity Census of Women and Minorities on Fortune 500 Boards. https://www.catalyst.org/research/missing-pieces-reportthe-2018-board-diversity-census-of-women-and-minorities-on-fortune-500-boards/ (Accessed: July 28, 2019).
- Catalyst (2019b). *Women CEOs of the S&P 500*. https://www.catalyst.org/research/women-ce-os-of-the-sp-500/ (Accessed: July 28, 2019).
- Crunchbase Pro, at https://www.crunchbase.com.
- Dow Jones Venture Source and The Wall Street Journal, *The Billion Dollar Startup Club* (by Scott Austin, Chris Canipe & Sarah Slobin), at http://graphics.wsj.com/billion-dollar-club/ (Accessed on May 2017).
- Giusepponi, K., Tavoletti, E., & Tosi, V. (2019). Top startups worldwide. An investigation on entrepreneurial and organisational profiles. Milano: Franco Angeli.
- Glassdoor, at https://www.glassdoor.com.
- Jennings, J. E., & Brush, C. G. (2013). Research on women entrepreneurs: challenges to (and from) the broader entrepreneurship literature? *The Academy of Management Annals*, 7(1), 663-715. DOI: 10.1080/19416520.2013.782190
- Kanze, D., Huang, L., Conley, M. A., & Higgins, E. T. (2018). We ask men to win and women not to lose: Closing the gender gap in startup funding. *Academy of Management Journal*, 61(2), 586-614. DOI: 10.5465/amj.2016.1215
- Krishna, V. (2017). Find the right talent, but better to let go of people sooner if they are not the right fit: Houzz founder Adi Tatarko. *Your Story*. March, https://yourstory.com/2017/03/adi-tatarko-houzz (Accessed: July 28, 2019).
- Kuschel, K., Labra, J.P., & Díaz, G. (2018) Women-Led Startups and Their Contribution to Job Creation. In: Presse A., Terzidis O. (eds) Technology Entrepreneurship. FGF Studies in Small Business and Entrepreneurship. Springer, Cham.
- Kuschel, K., Lepeley, M., Espinosa, F., & Gutiérrez, S. (2017). Funding challenges of Latin American women start-up founders in the technology industry. *Cross Cultural & Strategic Management*, 24(2), 310-331. DOI: 10.1108/CCSM-03-2016-0072
- LinkedIn, at https://www.linkedin.com.

- Northzone (2019). *Jessica Schultz*. https://northzone.com/team/member/jessica-schultz/ (Accessed: December 20, 2019).
- Quora, at https://www.quora.com.
- Soo, Z. (2019). Grab co-founder Tan Hooi Ling has had to work hard to embrace the spotlight. South China Morning Post, https://www.scmp.com/tech/enterprises/article/3013316/grabco-founder-tan-hooi-ling-has-had-work-hard-embrace-spotlight (Accessed: December 20, 2019).
- Vaccarini, K., Spigarelli, F., Tavoletti, E., & Latteman, C. (2017). Cultural Distance in International Ventures Exploring Perceptions of European and Chinese Managers. Basingstoke: Palgrave.
- Weiner, Y. (2018). An entrepreneur is someone who gets motivated by the word 'no'. *Thrive Global*. October. https://thriveglobal.com/stories/an-entrepreneur-is-someone-who-gets-motivated-by-the-word-no-with-amy-pressman-co-founder-of-medallia/ (Accessed: July 28, 2019).
- World Economic Forum (2018). The Global Gender Gap Report 2018. Switzerland: World Economic Forum. http://www3.weforum.org/docs/WEF_GGGR_2018.pdf (Accessed: July 28, 2019).
- World Economic Forum (2021). *Global Gender Gap Report 2021*. Switzerland: World Economic Forum. https://www.weforum.org/reports/global-gender-gap-report-2021/ (Accessed: April 30, 2022).

AUTHORS' CONTRIBUTIONS

KG and ET jointly undertook the research project, carried out the study and wrote all sections of the manuscript.



Financialization – Evidence from Dynamic Connectedness among Agricultural Index Futures

Neeti Jain¹ ^D Niti Nandini Chatnani² ^D

Received: July 3, 2022 Accepted: August 31, 2022 Published: December 30, 2022

Keywords: Agricultural commodity markets; TVP-VAR; Futures markets

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-Non-Commercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission. **Abstract:** The introduction of index futures was a landmark event for global commodity markets. It has been blamed by regulators and academicians for its role in food price surges from time to time. This paper examines the price discovery and volatility spillover relationship among agricultural index futures globally. Results from the study reveal that index futures play a dominant role in contributing to price discovery. The price leadership of the futures market, although found to be strong, is diminished in the presence of stringent regulatory trading curbs that were put in place as a response to the crisis.

Furthermore, an improved Diebold & Yilmaz method based on TVP-VAR-SV model was used to analyze dynamic connectedness between the index and standalone contracts of agriculture commodity markets. The results show that the impacts on the net spillover of various indices are different. However, the evidence fails to support the argument that volatility is induced due to spillovers among the indices.

1. INTRODUCTION

Over the last decades, the market environment of commodity trading has undergone vast changes. With trade barriers diminishing, the need for modern risk management and measurement has become crucial. The spillovers among markets can be measured through connectedness relating to systemic risk and systematic risk. Understanding the systemic risks helps to determine the contribution of financial markets to the overall financial system and acts as an early warning system.

Bulk commodities mostly listed on organized futures markets, form an important part of the industry chain and transmit superimposed demand. Financialization is the transformation of a segmented market into an asset class for portfolio investors that has distorted commodity pricing in the futures market. The financialization of commodities is linked to the market developments in the mid-2000s period. Financialization of the futures market has been linked to diminishing hedging mechanisms and increasing speculative behavior. In times of geo-political tensions, there is an increase in such exuberant behavior. Agricultural commodities have a vital strategic influence on most countries. The consequence of high agricultural prices is more dramatic for emerging and developing economies characterized by major expenditure spent on food consumption.

2. LITERATURE REVIEW

Using Markowitz's (1952) portfolio setting, Bodie and Rosansky (1980), Jensen et al. (2000) and Edwards and Liew (2013) concluded that the presence of commodities enhances the portfolio

² Indian Institute of Foreign Trade, B-21 Qutub Institutional Area New Delhi 110016, India



¹ Indian Institute of Foreign Trade, B-21 Qutub Institutional Area New Delhi 110016, India

return, while Cao et al. contested that the inclusion of commodities is not advisable and evidenced that their contribution is a placebo effect (Bodie & V.I., 1980; Cao et al., 2012; Edwards & Liew, 2013; Jensen et al., 2000).

Post-2003, the increase in open interest of commodity futures markets has increased cross-linkages among markets as hedge funds and commodity index funds are used for risk management reasons. Sharp downward movements in equity markets have led to financial investors liquidating their stakes in commodity markets in order to raise cash for margin calls or safeguard investments (Büyüksahin et al., 2010). Events of the global financial crisis of 2008 and the subsequent commodities super-cycle that lasted up to 2011 provide an opportunity to study such scenarios (Cargill, 2021; Clapp & Helleiner, 2012; Singleton, 2014). Academicians and practitioners have pressed for substantial allocations to commodities as an asset class for the purpose of diversification and return generation (Carter et al., 2017; le Pen & Sévi, 2018; Naeem et al., 2021).

Basak & Pavlova (2016) conclude that the effects of financialization are only on storable commodity prices (Basak & Pavlova, 2016). In the presence of institutions, the prices of storable commodities and their inventories are higher than in the benchmark economy and the effects are more pronounced for commodities that are part of an index. Irwin and Sanders (2012) observe largely unrelated regressions and the limited result of Granger causality from index trading to returns (Irwin & Sanders, 2012).

Asymmetry between the costs of going long and short restricts the ability of some investors to trade pessimistically on new information. However, such behavior has limited evidence of inducing volatility. Boyd et al. (2016) find consistent evidence, documenting that herding behavior by speculators across 32 futures markets was limited in scope and serves to stabilize markets—speculators herd by buying (selling) into falling (rising) markets (Boyd et al., 2016). Index investments having agricultural contracts as their underlying asset are more adaptive to informational disclosures relating to global supply and demand forecasts (Balcilar et al., 2021)

Index investments having agricultural contracts as their underlying asset are more adaptive to informational disclosures relating to global supply and demand forecasts. The introduction of commodity index funds is targeted as a driver of creating global liquidity imbalances. The question stands - Has speculative trading in agricultural futures markets led to increased volatility, leading to a global food security crisis?

There is a greater need of understanding the factors that lead to speculation and volatility. The price discovery function of agricultural products is ambiguous in the current literature. The paper attempts to discuss how futures prices adjust to demand-driven changes in commodity prices and by gradual entry and exit of index investments. The ambiguity of what classifies as a shift of demand and supply and what as bubble formation remains to be detangled.

3. METHODOLOGY

TVP-VAR model enables us to capture a possible time-varying nature of underlying structure in the economy flexibly and robustly. All parameters in the VAR specification are assumed to follow the first-order random walk process, thus allowing both temporary and permanent shifts in the parameters (Balcilar et al., 2017). Dependence among commodity price co-movements helps to extract information related to risk management & hedging. With comparatively fewer parameters compared to other econometric models and simplicity of comprehending solutions, the method is suitable to broaden our understanding of financialization through the lens of agricultural futures indices.

The connectedness approach introduced by Diebold and Yilmaz is used to explore complex nonlinear transmission mechanisms in networks across areas of finance and economics (Diebold et al., 2009; Diebold & Yilmaz, 2012, 2014). It allows the comparison of different impacts that a shock in one variable has on the ability of another variable to forecast error variance while considering the feedback loops. The joint spillover index enables a natural depiction of spillover along with measuring the goodness-of-fit measure.

The stochastic volatility model is a generalized version of Black-Scholes model to allow analysis of stochastic volatility, assuming that there exists the price of the underlying factor, two risk factors and a price volatility factor (Xu & Gong, 2022). The risk neutral state can be denoted as:

$$D S = (r - \delta) S dt + \sqrt{\nu}S d z_1$$

And

$$Dv = \kappa \text{ (m-v) } dt + \sigma p \sqrt{vS} dz_1 + \sigma \sqrt{1 - p^2} \sqrt{v} dz_2$$

where

K = Rate of mean reverting,

m = Long run mean.

Mean reversion is an assumption of the model. Volatility process can be associated positively or negatively correlated to the price discovery process, dependent on the sign p denotes. Previous studies have concluded that indices or equity securities exhibit negative correlation between prices and price volatility while the converse holds true for commodities.

The arbitrage equation for V (S,v,t) is:

$$rv = V_t + (r - \delta) SV_S + = \kappa (m - v) V_v + \frac{1}{2}vS^2V_{ss} + \sigma pv SV_{sv} + \frac{1}{2}\sigma^2(1 - p^2)vV_{vt}$$

On conversion of the prices into logarithm scale, s = ln(S). Applying Ito's Lemma, the equation is represented as:

 $Ds = (r - \delta) dt + \sqrt{\nu} dz_1$

Arbitrage equation for V (S,v,t) in the reduced form can be expressed as:

$$rV = V_t + (r - \delta) V_S + = \kappa (m - v) V_v + \frac{1}{2} v V_{ss} + \sigma p v V_{sv} + \frac{1}{2} \sigma^2 (1 - p^2) v V_{vv}$$

The underlying state process that is expressed in form of s and v is a form of affine diffusion. Stochastic Volatility models refer to the stochastic and time varying specification of the variance evolution. In particular, the assumption of following AR(1) process is made for log-variance.

8th International Scientific Conference ERAZ 2022 Selected Papers

4. DATA

The selection criteria for the data were covering agricultural indices that were frequently traded, had agricultural futures contracts majorly as underlying assets and covered key commodity markets. The data was extracted from Bloomberg Terminal and covers the period from 17 February 2009 to 20 May 2022.

Daily index futures data from Bloomberg Commodities Sub Index Australia (BCOMAU), ICE Futures Europe - Index Future Contract on MSCI BRIC Index in USD (MBRI), S&P GSCI Agriculture Enhanced Euro Index (SGCUEGE) and Bloomberg Commodities Sub Index US (BCOMAG) are considered. BCOMAU is represented as AUS, MBRI as BRIC, SGCUEGE as EURO and BCOMAG as US based on the geographical location or group of countries and their underlying markets.

The descriptive analysis is presented in Table 1. All the series had significantly leptokurtic distributions indicating that the tails were fatter than a normal distribution. This supported the Jarque-Bera normality tests of the distributions not being distributed normally. Hence, the squared returns were used to employ TVP-VAR-SV approach.

5. **RESULTS**

The section covers the dynamic and average connectedness measures. The average value of the Total Connectedness Index predicting the entire dataset is depicted across time in Figure 2. It is useful to evaluate the response of TCI to various economic factors prevailing during the period. During 2020, vast disruptions were experienced by commodities across the board. Table 2 illustrates the joint spillover index that enables comparisons across the dataset. NPT represents Net Pairwise Total that evaluates interrelation among the indices over the evolution of time and their potential roles. It should be noted that for the indices under study, a positive value indicates the net transmitting role while negative values refer to the net receiving role.

| | Table 1. Descriptive Statistics | | | | | |
|-------------|---------------------------------|----------|----------|----------|--|--|
| | EURO | BRIC | AUS | US | | |
| Mean | 500.0232 | 84.7479 | 99.47631 | 60.53626 | | |
| Median | 473.6545 | 79.30025 | 100.2998 | 57.30335 | | |
| Maximum | 843.387 | 163.8554 | 146.1163 | 97.6683 | | |
| Minimum | 212.023 | 55.9502 | 70.9493 | 34.1523 | | |
| Std. Dev. | 108.2023 | 16.20589 | 14.32732 | 15.7856 | | |
| Skewness | 0.626714 | 1.49555 | 0.31073 | 0.402672 | | |
| Kurtosis | 3.175581 | 5.978568 | 2.422268 | 2.235013 | | |
| Jarque-Bera | 214.3889 | 2384.715 | 96.35812 | 165.1214 | | |
| Sum | 1606075 | 272210.2 | 319517.9 | 194442.5 | | |

 Table 1. Descriptive Statistics

Source: Own research

It can be seen that the US index BCOMAG (US) and European index MBRI (Euro) assume the role of both net transmitters and receivers at the same time; however, the effect of spillover does not appear to be de-stabilizing. From the impulse response table, figure 2, the time varying behavior of active portfolio management is undermined as the volatility having a contagion effect is underscored.



Figure 1. Historical decomposition structural VAR Table Source: Own research

The MBRI index (BRIC economies) and BCOMAU (Australian index), in the stated order exercise influence in term spillover effect. An argument for the same is that during times of increased market uncertainty and risk, there is an effect on government interventions that leads to increasing volatility connectedness of agricultural markets. In the case of rising markets, the volumes of investors diversifying or switching their investments from bonds and equity to commodity markets increase, leading to high volatility. Another argument relates to the lead-lag relationship between information transmission and price discovery. As developed markets such as US and Eurozone have wider participation from institutional and retail investors in their indices, and their agricultural futures markets are deep, these markets are dominant in determining the price vis-à-vis the developing markets.

| | Table | 2. Dynamic Nei | | able | |
|---------|--------|----------------|--------|--------|-------------|
| | Aus | BRIC | Euro | US | FROM |
| Aus | 63.44 | 2.24 | 17.14 | 17.19 | 36.56 |
| BRIC | 4.48 | 81.17 | 5.09 | 9.27 | 18.83 |
| Euro | 9.89 | 2.37 | 53.65 | 34.09 | 46.35 |
| US | 10.85 | 3.3 | 35.65 | 50.21 | 49.79 |
| ТО | 25.22 | 7.9 | 57.87 | 60.54 | 151.53 |
| Inc.Own | 88.66 | 89.07 | 111.52 | 110.75 | cTCI/TCI |
| NET | -11.34 | -10.93 | 11.52 | 10.75 | 50.51/37.88 |
| NPT | 1 | 0 | 3 | 2 | |

Table 2. Dynamic Net Connectedness Table

Source: Own research

8th International Scientific Conference ERAZ 2022 Selected Papers



6. CONCLUSION

The volatility spillover based on information transmission is marginally high in American & European Markets. It indicates the price discovering role for these markets. There is no strong evidence of financialization having a direct impact on the 4 indices.

We can envisage agricultural commodities indices have an informational transmission in the long run so that the results of the correlation test provide that at least one co-integrating equation exists between two variables, that is, two global indices. The patterns are essential in order to capture the movement of the volatility trend. It can be straightforwardly envisaged that the trend of responses occurred by different exogenous shocks in different periods. The impulse response function suggests that there is no significant effect of shocks on the global indices.

The findings are crucial for portfolio and risk managers aspiring to design an optimal risk-return strategy. The connectedness effect can act as an early warning system for potential spillover. Understanding the key factors driving spillovers in agricultural commodities can help determine the future macro-economic environment as well. This will help in maximizing the welfare objectives for the investors.

REFERENCES

- Balcilar, M., Cakan, E., & Gupta, R. (2017). Does US news impact Asian emerging markets? Evidence from nonparametric causality-in-quantiles test. North American Journal of Economics and Finance, 41, 32–43. DOI: 10.1016/j.najef.2017.03.009
- Balcilar, M., Gabauer, D., & Umar, Z. (2021). Crude Oil futures contracts and commodity markets: New evidence from a TVP-VAR extended joint connectedness approach. *Resources Policy*, 73. DOI: 10.1016/j.resourpol.2021.102219

- Basak, S., & Pavlova, A. (2016). A Model of Financialization of Commodities. *Journal of Finance*, *71*(4), 1511–1556. DOI: 10.1111/jofi.12408
- Bodie, Z., & V.I., R. (1980). Risk and Return in Commodity Futures. 36(3), 27-31.
- Boyd, N. E., Büyükşahin, B., Haigh, M. S., & Harris, J. H. (2016). The Prevalence, Sources, and Effects of Herding. *Journal of Futures Markets*, *36*(7), 671–694. DOI: 10.1002/fut.21756
- Büyüksahin, B., Haigh, M. S., & Robe, M. A. (2010). Commodities and equities: ever a "market of one"? *The Journal of Alternative Investments, Winter*, 76–95. www.iijournals.com
- Cao, G., Xu, L., & Cao, J. (2012). Multifractal detrended cross-correlations between the Chinese exchange market and stock market. *Physica A: Statistical Mechanics and Its Applications*, 391(20), 4855–4866. DOI: 10.1016/j.physa.2012.05.035
- Cargill. (2021). Corn and Soybeans: The Start of a New Supercycle in Agriculture?
- Carter, D. A., Rogers, D. A., Simkins, B. J., & Treanor, S. D. (2017). A review of the literature on commodity risk management. In *Journal of Commodity Markets*. DOI: 10.1016/j. jcomm.2017.08.002
- Clapp, J., & Helleiner, E. (2012). Troubled futures? The global food crisis and the politics of agricultural derivatives regulation. *Review of International Political Economy*, 19(2). DOI: 10.1080/09692290.2010.514528
- Diebold, F. X., & Yilmaz, K. (2012). Better to give than to receive: Predictive directional measurement of volatility spillovers. *International Journal of Forecasting*, 28(1). DOI: 10.1016/j. ijforecast.2011.02.006
- Diebold, F. X., & Yilmaz, K. (2014). On the network topology of variance decompositions: Measuring the connectedness of financial firms. *Journal of Econometrics*, *182*(1), 119–134. DOI: 10.1016/j.jeconom.2014.04.012
- Diebold, F. X., Yilmaz, K., Binder, M., Dominguez, K., Frankel, J., Giavazzi, F., Leeper, E., Reichlin, L., & West, K. (2009). Measuring Financial Asset Return and Volatility Spillover, with Application to Global Equity Markets. *The Economic Journal*, 119, 158–171. https://academic.oup.com/ej/article/119/534/158/5089555
- Edwards, F. R., & Liew, J. (2013). Managed Commodity Funds. 9(November 2012), 38-47.
- Irwin, S. H., & Sanders, D. R. (2012). Financialization and Structural Change in Commodity Futures Markets. *Journal of Agricultural and Applied Economics*. DOI: 10.1017/s1074070800000481
- Jensen, G. R., Johnson, R. R., & Mercer, J. M. (2000). Efficient Use of Commodity Futures in Diversified Portfolios. 20(5), 489–506.
- le Pen, Y., & Sévi, B. (2018). Futures trading and the excess Co-movement of commodity prices. *Review of Finance*, *22*(1), 381–418. DOI: 10.1093/rof/rfx039
- Naeem, M. A., Farid, S., Nor, S. M., & Shahzad, S. J. H. (2021). Spillover and drivers of uncertainty among oil and commodity markets. *Mathematics*, 9(4), 1–23. DOI: 10.3390/math9040441
- Singleton, K. J. (2014). Investor flows and the 2008 boom/bust in oil prices. *Management Science*, 60(2). DOI: 10.1287/mnsc.2013.1756
- Xu, J., & Gong, X. (2022). Geopolitical risk and dynamic connectedness between commodity markets.



Financial Literacy of People Gaining Economic Education in the Context of Acquired Practical Experience

Josef Polák¹ (b) Zuzana Kozubíková² (b) Aleš Kozubík³ (b)

Received: June 27, 2022 Accepted: July 20, 2022 Published: December 30, 2022

Keywords: Decision making; Economic education; Finance; Financial management; Financial market

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-Non-Commercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission.

Abstract: Financial literacy affects numerous economic decisions of people on a daily basis, as well as the strategic financial management of corporations. In several previous studies, the authors confirmed a significant difference in the level of financial literacy between full-time and combined students. Based on these results, they hypothesized that practical experience has a positive effect on the level of financial literacy, but this needed to be verified. Therefore, a new survey was conducted with an extended version of the questionnaire with questions specifically related to the practical use of financial products and services. Students of economics who have access to education that supports the growth of financial literacy were interviewed. However, the distribution of the examined groups was not determined by age, a form of study and other common aspects, but the divider was in form of the experience gained, which provided a unique approach using an adequate statistical set of respondents. The results from primary research showed differences in knowledge and the way of solving tasks in the monitored groups of respondents. Using statistical analysis methods and hypothesis tests, the authors' hypothesis was confirmed and it was shown that students using a wider range of services performed better. It provides usable interdisciplinary outputs for HRM, applied psychology, pedagogy and financial management. The level of financial literacy was tested by the personal finance index which is an innovative metric that analyses knowledge from the eight functional areas of financial literacy. Given that the research sample was drawn from a group of people studying financial management and entrepreneurship subjects, it was also confirmed that theoretical knowledge alone is not sufficient to significantly enhance financial literacy. Thus, the irreplaceability of practical experience for enhancing the level of financial literacy was demonstrated.

1. INTRODUCTION

Financial literacy is one of the crucial skills of a citizen of modern society in the 21st century. We encounter the need to make the right financial decisions that can affect our entire future life almost every day. Whether it's securing our financial situation in old age, making the right decision to finance our own home or providing financial security in times of illness, it is always essential to make the right decision at an early stage. Therefore, it is extremely important to strengthen the financial literacy of young people as they are facing these life decisions but with minimal experience.

In our previous research (Kozubik, Kozubíková, & Polák, 2019; Polák, Kozubíková, & Kozubík, 2018), we compared the level of financial literacy of students in the face-to-face and distance

³ University of Žilina, Faculty of Managements and Informatics, Dpt. of Mathematical Methods and Operations Research, Univerzitná 8215/1, 010 26 Žilina, Slovakia



¹ College of Regional Development and Banking Institute – AMBIS, Department of Economics and Management, Lindnerova 575/1, 180 00 Praha 8, Czech Republic

² University of Žilina, Faculty of Managements and Informatics, Dpt. of Macro and Microeconomics, Univerzitná 8215/1, 010 26 Žilina, Slovakia

learning forms of study. It turned out that distance learning students achieved a higher success rate. Since these are students who have a permanent employment relationship, we concluded that their enhanced practical experience is the factor that positively influenced their results. In order to confirm this conjecture, it was considered necessary to conduct another survey, this time focusing on real practical experience.

In this follow-up survey, we focused on the relationship between practical experience with financial market products and the level of financial literacy. In doing so, we abstracted from the form of study and focused only on actual practical experience. In this spirit, we also formulated the research hypothesis.

H1: Students with more practical experience with financial market tools and services achieve higher levels of financial literacy.

2. LITERATURE REVIEW

There are a number of studies studying financial literacy in various social groups. Some of them are united by the claim that a higher level of financial literacy of respondents subsequently improved their financial decisions which seems to be logical. Van Rooij *et al.* (2011) measure financial literacy and examine its relationship to stock market participation in their research. They found that most respondents have basic financial knowledge and are familiar with concepts such as interest-bearing, inflation and the time value of money. The situation has changed for more complex concepts. Many respondents did not know the difference between bonds and stocks, the relationship between bond prices and interest rates, and the basics of risk diversification. Based on their findings, van Rooij *et al.* (2011) state that financial literacy influences financial decisions. People with lower financial literacy and low-wealth investors then invested less in stocks and often hold the wrong portfolio. The situation in which respondents move to a system where they have to decide how much to save for retirement and how to invest their retirement wealth is pushing them to change their approach.

According to the findings of Sun *et al.* (2021) financial literacy has a significant positive impact on the choice of wealth management products, risky financial assets and the total amount of retirement financial assets. On the other hand, it has a negative impact on the choice of bank savings. Another negative impact of financial literacy is on the proportion of savings but Sun *et al.* (2021) found also a significant positive impact on the proportion of wealth management products and risky financial assets. In the context of risk perception, research by Riepe *et al.* (2022) identified a lower risk aversion for entrepreneurs with lower financial literacy. The difference disappears for entrepreneurs with higher financial literacy.

A higher level of financial literacy seems to be an advantage. However, Zhu (2020) identified certain disadvantages in research focused on adolescents. Adolescents who are too confident in their financial literacy are more likely to engage in risky financial behavior and show a higher level of financial autonomy. Zhu (2020) also identified this type of behavior in adults. Excessive self-confidence can therefore weaken the necessary prudence in decision-making. Kawamura *et al.* (2021) even state that people with high financial literacy tend to take risks and over-indebtedness. These people also take naive financial attitudes and become bold and reckless, but on the other hand, they are more responsible in retirement planning. Gerth *et al.* (2021) also point to the negative connection between excessive self-confidence and financial literacy

in their article examining the contribution of behavioral characteristics to the financial literacy of UAE residents.

The development of financial literacy can also be viewed from the point of view of the cultural, customary and social environment. Razen *et al.* (2021) focused on the level of financial literacy of adolescents in Austria. They state that the level of financial literacy is positively associated with patience, the male gender, and the educational level of the father. At the same time, they found that risky behaviors such as smoking and gambling are positively associated with various measures of risk tolerance and negatively associated with patience. Among other things, their results underline the importance of financial education as a family effort.

Bottazzi & Lusardi (2021) claim that the gender differences in the financial literacy of young people in Italy are large. They identify this fact in all regions of the state, especially in the south and the islands. The role of parental background and especially the role of mothers were identified as a key role in the development of financial knowledge. Bottazzi & Lusardi (2021) state that the social and cultural background of boys and girls and the historical context of the locality are important for gender differences in financial literacy. The authors identified favorable conditions for the transformation of the role of women in society in the original medieval commercial centers and the nuclear family structure.

Okamoto & Komamura (2021) also state the positive effect of financial literacy. In their research, they focused on seniors living in Japan. They identified an increase in financial literacy by age 60. Subsequently, this literacy decreased among the respondents, but self-confidence in decision-making remained high (so-called overconfidence bias). With declining financial literacy, the authors identified a decline in cognitive abilities, beliefs about the benefits of accumulated experience, and stereotypes about better financial literacy of men. The research also identified that men's financial decisions are less thoughtful. However, lower financial literacy for women may be associated with access to adequate education and the amount of financial assets. In any case, as women live to be older, they must have a financial plan in place. Socio-cultural aspects associated with financial literacy are evident in the results of this research.

Kim *et al.* (2021) identified a decline in financial literacy among the senior population in the United States but they also found in this group that more financially educated individuals sought the services of counsellors, which can be described as responsible behavior. However, this behavior does not preclude subsequent inappropriate financial decisions.

Fong *et al.* (2021) identified an understanding of interest rate composition and inflation in a sample of pensioners in Singapore, but only half of the respondents understood the issue of risk diversification. The authors also identified significant problems with the appropriate portfolio allocation, which would correspond to the age of the investors. Respondents with higher financial literacy had a higher tendency to repay credit cards on time, hold stock, and follow an age-appropriate investment glide path in their research.

A number of studies demonstrate the importance of financial literacy and its benefits and risks to respondents living in given social groups. In this context, it is interesting to mention the study by the author's Park & Martin (2021), which states that financial behavior cannot be judged solely on the basis of the influence of financial factors. They state that this is a complex area of how the financial environment interacts with consumers' psychological factors and influences their

decisions. They, therefore, emphasize the need for clear and comprehensible financial services for consumers with low financial literacy. Although increasing financial literacy is important, many financial services are relatively complex and require a more advanced understanding of the issue.

Just as financial aspects affect many aspects, people with higher financial literacy affect their surroundings. This statement can be broken down by the results of several studies. The first is a study from Bangladesh. The variables such as professions, income levels, knowledge of depositing and withdrawing money and knowledge of interest rates have a major impact on overall access to finance and, together with education programs on financial literacy, improve the financial integration of rural regions in developing countries and they help alleviate poverty reduction and prosperity development (Hasan, Le, & Hoque, 2021). However, if we look at a market in which financial literacy is already developed, it is possible to document the study of Jin *et al.* (2021), which examines the relationship between financial literacy and transparency of bank financial reporting on a sample of US banks. They state that the financial literacy of banking institutions' customers can influence the behavior of bank managers in terms of the mechanics of the loan loss provisioning and their opportunistic actions. Their article also states that financially literate customers represent more stable sources of financing, enabling more predictable creation of provisions for credit losses. This situation subsequently contributes to more lasting profits.

In a Swedish study by Anderson & Robinson (2021), the behavior of financially literate and environmentally conscious financial market users can be found. The study states that households with environmental preferences generally do not make green financial decisions due to financial disengagement and information constraints associated with green investment decisions. Households with strong pro-environmental behavior and beliefs are not financially involved and are generally not interested in financial matters. However, if we look at active investors making decisions in their pension portfolios, environmentally oriented investors are more likely to buy mutual funds with pro-environmental names. Within the available information, they give priority to environmental preferences. The availability and manner of providing relevant investment information remain an issue. In this example, it is also possible to identify the impact of financial literacy of environmentally-based individuals on investment decisions and thus on their surroundings.

Fujiki's (2022) research from Japan showed that financial literacy helped to ensure the required social distance during a pandemic through the ability to use non-face-to-face financial services and cashless payments.

The documented examples show that higher financial literacy affects the behavior and decision-making of financial market participants, but at the same time, financially literate participants affect their surroundings and the financial market.

Quality education has its positive effects on the development of society, but it is necessary to observe new phenomena that create further challenges. At the same time, it is necessary to implement new scientific knowledge at the political level and into practice. Petrů, Kramoliš, & Stuchlik (2020) state, among other conditions, that if the state or educational institutions could create truly effective teaching support, the reluctance of many business owners to educate and implement changes in their business could be alleviated. This statement follows their research on small and medium-sized enterprises in the Czech Republic.

3. METHODOLOGY

Methodologically, our research, like all statistical investigations, had two phases. The first phase involved data collection, and the second phase involved data evaluation using methods of mathematical statistics.

A standard procedure for collecting data on financial literacy is a questionnaire survey. Our survey was no exception to this traditional approach. We used the personal finance index (shortly P-FIN index) introduced in Lusardi, Yakoboski, & Oggero (2017) as an instrument to measure financial literacy. We adapted the composition of our questionnaire to cover all eight functional areas of financial literacy defined in the P-FIN index. Specifically, the functional areas of the P-FIN index are determined as:

- earnings, determinants of wages and income,
- consuming, budgeting and spending,
- saving, comprehension of the accumulation factors,
- investing, understanding the types and risks of investments,
- borrowing and debt management,
- risk management, comprehension of the uncertain outcomes,
- insurance and the understanding of coverages,
- accessing and working with information sources.

To test the respondents' abilities in each functional area of financial literacy, we designed the theoretical part of the questionnaire. This part contained 30 questions with multi-choice option responses. Of the possible choices, only one was always correct, and one choice was of the "I don't know" type. We then interpret the P-FIN index as the percentage success rate of the whole test.

In the second part of the questionnaire, we investigated respondents' practical experience with banking and insurance market products. Among banking services, we were interested in both deposit and loan products. Since the use of personal accounts has become standard, we considered respondents who used at least two deposit or investment products to be more experienced. Here we were particularly interested in term deposits, savings accounts, mutual funds, and also building savings. In the area of credit products, we have included mortgage loans, consumer loans, student loans, as well as loans from non-bank entities and instalment sales. Finally, from insurance products, we surveyed experience in life insurance and non-life insurance, in particular liability insurance, car insurance, and property insurance. In the case of credit and insurance products, we divided respondents into two groups based only on whether or not they use a product.

In the second phase, we further analyzed the data obtained from the questionnaire survey using statistical methods. The main tool was the statistical hypothesis testing technique. First, we had to check whether our research samples could be considered samples from a normal distribution. We used the Shapiro-Wilk test to verify normality. The result of this test then influenced the decision to select the appropriate test for comparing results across groups. If the assumption of normality of distribution is rejected, the Kruskal-Walis test should be used, otherwise, the Welsch t-test can be applied.

All numerical calculations were performed in the specialized statistical environment R. All mentioned tests are implemented as functions in this programming language. For this reason, we do not provide a detailed explanation of the test methods used.

4. **RESULTS**

We performed our survey at two universities in the Czech Republic and Slovakia. Specifically, the University of Žilina in Žilina and AMBIS in Prague. The research targeted students in management fields of study. We distributed 750 questionnaires among the students as part of the survey. After removing incompletely filled answer sheets and maliciously completed answer sheets (all questions answered in with "I don't know" responses), we obtained 652 questionnaires. The corresponding response rate is 86.93%.

| | Experience with products | | | |
|------------|--------------------------|-------|-----------|--|
| | Deposits | Loans | Insurance | |
| Count | 342 | 28 | 134 | |
| Percentage | 52.45% | 4.30% | 20.55% | |

 Table 1. Counts and percentages of respondents according to their experience with financial products

Source: Own elaboration

It showed that 342 respondents use multiple deposit products, while 310 use only a current account. Thus, two numerically balanced groups emerged. There was a significant imbalance when disaggregated by the use of credit products. Here there were only 28 cases that have practical experience with loans. Despite this disproportion, we have compared their test scores here as well. The use of insurance products was reported by 134 respondents, about one-fifth. These counts are summarized in Table 1.

Table 2. Descriptive statistics of the P-Fin scores according to the experience types of the respondents

| | Min. | 1st Quartile | Median | Mean | 3rd Quartile | Max |
|-------------------------|------|--------------|--------|------|--------------|------|
| No deposit experience | 0.10 | 0.40 | 0.47 | 0.48 | 0.56 | 0.86 |
| Deposit experienced | 0.17 | 0.47 | 0.53 | 0.56 | 0.67 | 0.97 |
| No loan experience | 0.10 | 0.40 | 0.53 | 0.52 | 0.60 | 0.97 |
| Loan experienced | 0.20 | 0.52 | 0.60 | 0.62 | 0.74 | 0.93 |
| No insurance experience | 0.10 | 0.40 | 0.50 | 0.50 | 0.57 | 0.90 |
| Insurance experienced | 0.27 | 0.50 | 0.60 | 0.62 | 0.75 | 0.97 |

Source: Own elaboration

Based on their experience with different types of financial market products, we divided respondents into experienced and inexperienced groups. The basic statistical characteristics of the resulting P-FIN index values are summarized in Table 2. In all indicators, there is a shift towards better results for more experienced respondents. The only exception is the maximum value when broken down by loan experience.

Before proceeding to test the hypothesis of equality of the mean values of the P-FIN index, it is necessary to verify that the results are normally distributed. For this purpose, we performed the Shapiro-Wilk test. To save some space, we report only a summary that in all tests the p-value of the test ranged from $2.2 \cdot 10^{-16}$ to 0.002. This means that we can reject the hypothesis of a normal distribution with high confidence, and we need to use the Kruskal-Wallis test to test for equality of means. Tables 3-5 present the results of tests for equality of average success rates among respondents, broken down by product experience. As we can see from the tables, in all three cases we can reject the hypothesis at a high confidence level. Thus, we can conclude that more experience with financial market products and services leads to higher levels of financial literacy.

| Table 3. Results of the Kruskal-Wallis test of equality of the mean percentages | 5 |
|---|---|
| for the respondents experienced and inexperienced with the credit products | |

| Credit experience | Mean percentage | chi-squared | p-value | |
|-------------------|-----------------|-------------|-------------|--|
| Inexperienced | 48.47 % | 41.66 | 1.083.10-10 | |
| Experienced | 56.08 % | 41.66 | | |

Source: Own elaboration

Table 4. Results of the Kruskal-Wallis test of equality of the mean percentages for the respondents experienced and inexperienced with the loan products

| Credit experience | Mean percentage | chi-squared | p-value |
|-------------------|-----------------|-------------|---------|
| Inexperienced | 52.03 % | 10.007 | 0.002 |
| Experienced | 62.14 % | 10.007 | 0.002 |

Source: Own elaboration

Table 5. Results of the Kruskal-Wallis test of equality of the mean percentages for the respondents experienced and inexperienced with the insurance products

| Credit experience | Mean percentage | chi-squared | p-value | |
|-------------------|-----------------|-------------|-------------------------|--|
| Inexperienced | 50.01 % | 60.095 | 9.041·10 ⁻¹⁵ | |
| Experienced | 61.94 % | 00.095 | 9.041.10 | |

Source: Own elaboration

A more detailed look at the structural differences in financial literacy levels can be obtained by testing the average scores achieved in the individual functional areas. Table 6 summarizes the results of the Kruskal-Wallis test for equality of means for each functional area according to the experience of using deposit and savings products. As can be observed from this table, respondents with more experience with deposit and savings products achieve higher levels of financial literacy in all its functional areas. This result is confirmed with a confidence level exceeding 95% in all cases.

Table 6. Results of the Kruskal-Wallis test of equality of the mean percentages in all functional areas for the respondents experienced and inexperienced with the credit products

| Functional area | Credit experience | Mean percentage | chi-squared | p-value |
|---------------------|-------------------|-----------------|-------------|-------------------------|
| Earnings and income | Inexperienced | 46.85% | 30.416 | 3.486.10-8 |
| | Experienced | 57.97% | | 5.400 10 |
| Consuming, and | Inexperienced | 63.63% | 5.3177 | 0.02111 |
| spending | Experienced | 68.05% | 3.31// | 0.02111 |
| Savings | Inexperienced | 56.61% | 8.6659 | 0.003242 |
| | Experienced | 63.74% | 8.0039 | 0.005242 |
| T | Inexperienced | 37.10% | 7.864 | 0.005043 |
| Investing | Experienced | 42.76% | /.804 | |
| Debt management | Inexperienced | 62.50% | 14.993 | 0.000108 |
| Debt management | Experienced | 70.18% | 14.995 | |
| Dials management | Inexperienced | 31.53% | 6.2859 | 0.01017 |
| Risk management | Experienced | 36.11% | 0.2839 | 0.01217 |
| Incurance | Inexperienced | 40.73% | 42.827 | 5.981·10 ⁻¹¹ |
| Insurance | Experienced | 54.09% | 42.827 | 5.981.10 |
| Information sources | Inexperienced | 49.19% | 4.531 | 0.03329 |
| accessing | Experienced | 55.41% | 4.551 | 0.03529 |

Source: Own elaboration

| Functional area | Credit experience | Mean percentage | chi-squared | p-value |
|---------------------|-------------------|-----------------|-------------|----------|
| Earnings and income | Inexperienced | 52.20% | 4.8131 | 0.02824 |
| | Experienced | 63.39% | | |
| Consuming, and | Inexperienced | 65.70% | 1.5061 | 0.2197 |
| spending | Experienced | 71.43% | | |
| Savings | Inexperienced | 60.10% | 1.323 | 0.2501 |
| | Experienced | 66.07% | | |
| Investing | Inexperienced | 39.78% | 2.1137 | 0.146 |
| | Experienced | 46.43% | | |
| Debt management | Inexperienced | 66.11% | - 3.9357 | 0.04727 |
| | Experienced | 75.89% | | |
| Risk management | Inexperienced | 33.29% | 10.94 | 0.000941 |
| | Experienced | 36.11% | | |
| Insurance | Inexperienced | 48.21% | - 14.831 | 0.000118 |
| | Experienced | 66.96% | | |
| Information sources | Inexperienced | 52.32% | 0.171961 | 0.6784 |
| accessing | Experienced | 55.36% | | |

 Table 7. Results of the Kruskal-Wallis test of equality of the mean percentages in all

 functional areas for the respondents experienced and inexperienced with the debt instruments

Source: Own elaboration

The results in the same structure for respondents grouped by their level of experience with debt instruments are shown in Table 7. Here the situation is not as clear-cut as in the previous case. The hypothesis that the average level of financial literacy is the same can be rejected in only half of the functional areas. Specifically, these are earnings, determinants of wages and income, borrowing and debt management, risk management, comprehension of the uncertain outcomes, insurance and the understanding of coverages.

Finally, Table 8 presents the results according to experience with insurance products. It is clear from the results presented in the table that only in the area of access to information resources the difference in results cannot be considered significant. In all other areas, the null hypothesis can be rejected with a confidence level exceeding 99%.

Table 8. Results of the Kruskal-Wallis test of equality of the mean percentages in all functional areas for the respondents experienced and inexperienced with the insurance

| Functional area | Credit experience | Mean percentage | chi-squared | p-value |
|---------------------|-------------------|-----------------|-------------|-------------|
| Earnings and income | Inexperienced | 48.99% | - 52.915 | 3.483.10-13 |
| | Experienced | 66.98% | | |
| Consuming, and | Inexperienced | 64.62% | 8.3138 | 0.003934 |
| spending | Experienced | 71.08% | | |
| Savings | Inexperienced | 58.49% | 9.3412 | 0.002241 |
| | Experienced | 67.53% | | |
| Investing | Inexperienced | 38.12% | - 13.898 | 0.000193 |
| | Experienced | 47.57% | | |
| Debt management | Inexperienced | 64.53% | - 16.766 | 4.229.10-5 |
| | Experienced | 74.25% | | |
| Risk management | Inexperienced | 31.56% | - 23.672 | 1.142.10-6 |
| | Experienced | 43.10% | | |
| Insurance | Inexperienced | 42.86% | - 86.611 | 2.2.10-16 |
| | Experienced | 66.60% | | |
| Information sources | Inexperienced | 51.83% | 0.74021 | 0.3896 |
| accessing | Experienced | 54.85% | | |

5. DISCUSSION

The results of our statistical analysis of the data clearly confirmed a higher percentage of students with a greater range of practical experience with financial sector products and services. This difference was reflected not only in the overall level of financial literacy as measured by the P-FIN index but also in the individual functional areas of financial literacy. This result is not very conclusive due to the small size of the group of respondents having experience with debt instruments. This is due to the low age of the respondents who do not yet have the need to incur debt, for example, to secure their own housing, etc. The fact that they do not tend to finance consumption through debt is also positive. It is also surprising that not a single respondent mentioned the use of student loans. In all cases, these results confirm the validity of our research hypothesis. Based on the findings of our research, we can conduct the following comprehensive discussion.

According to a number of studies (Fong *et al.*, 2021; Kim *et al.*, 2021; van Rooij *et al.*, 2011), financial literacy has a positive effect on responsible financial decision-making. In this research, we approached a sample of respondents by dividing students according to their experience with financial market products. This created a group of respondents with basic practical knowledge and a group of students with advanced knowledge of financial products (inexperienced and experienced group). Full-time and part-time students were mixed. However, both groups have access to the same quality management and economic education.

Based on the findings of this research, it can be agreed that financial literacy has a positive effect on financial decision-making, but at the same time we have identified a significantly higher level of financial literacy among a group of students who have practical experience in the financial market. This group of students was put into a decision-making process, where they had to decide what source of financing they would use, how they would invest money, how they would ensure their lives and health, and how they would lead their company financially and managerially.

In such a situation, the individual is often internally motivated to make the right decision. It is a stronger motivation than simply attending school and gaining a positive assessment. The personal involvement of the individual is relatively high here because, in the case of decision-making in personal finances, a bad decision can be associated with a threat to basic human needs. At the same time, one can also perceive the fact that a financially literate individual subsequently influences his / her surroundings due to his / her acquired knowledge, skills and preferences, as stated e.g. by Hasan, Le, & Hoque (2021) and Jin *et al.* (2021).

The authors of this article identified a similar way of human behavior and decision-making in their research as Zhu (2020), Kawamura *et al.* (2021) and Okamoto & Komamura (2021). Respondents from our research also answered questions that seemed easy in the wrong way. They did not pay them the necessary attention and approached them irresponsibly. Fong *et al.* (2021) and Kim *et al.* (2021) agree that higher financial literacy leads to more responsible financial behavior, but Kim *et al.* (2021) add that this behavior does not preclude subsequent financial decisions. These findings also correspond to the results of our research. Obviously, higher financial literacy will not solve all the problems that modern society faces. Park & Martin (2021) also state that financial behavior cannot be assessed solely based on the influence of financial factors, but the psychological factors of consumers must be taken into account.

Therefore, in the implementation of research, it is important to perceive the broader context of aspects affecting financial literacy. Based on our results, we state the importance of internal motivation of respondents themselves, for whom their life situation drove them to measurably improve financial literacy above the level of respondents at the same level of education, who have not yet needed to use the acquired knowledge in their lives and gain deeper knowledge individually. Other researchers (eg Bottazzi & Lusardi, 2021; Okamoto & Komamura, 2021; Razen *et al.*, 2021) also examine financial literacy in the context of the need to meet the living needs of individuals and their cultural, habitual and social environments. These aspects cannot be overlooked and it is important to work with them in increasing financial literacy, but also in solving other social problems.

At the same time, it is appropriate to draw attention to the need to apply scientific knowledge at the political level and in business practice, which may not use its potential for development, as stated by Petrů, Kramoliš, & Stuchlik (2020). The research from this article and also a number of researches listed in the literature provide valuable knowledge applicable in the management of private and corporate finances.

6. FUTURE RESEARCH DIRECTIONS

The results of the research are important findings in the field of financial literacy development not only for students but also for managers in companies. Effective training leads to the development of competencies and reduction of negative social manifestations, which include a higher percentage of foreclosures, inappropriate financial management of the company leading to stagnation of market value and bankruptcy, fraudulent sales practices, etc. Further research can focus on strong sources of motivation which will affect people's willingness to develop knowledge and skills related to financial literacy.

The findings of some authors (e.g. Kawamura *et al.*, 2021; Okamoto & Komamura, 2021; Park & Martin, 2021; Riepe *et al.*, 2022; Zhu, 2020) can be followed up and deepened for further research, as it is clear that having a community of financially literate individuals is not yet a target state in which social problems are solved. For this reason, it is also appropriate to focus research on identifying the factors of overconfidence that cause wrong decisions of financially literate individuals, as well as on the perception of risk itself, which will enrich the findings in the field of science.

The findings of the article also offer other research directions where it is possible to study the influence of financially literate individuals on their immediate surroundings and on the system as a whole within the countries of Central Europe, which have been involved in the development of financial literacy for several years.

In the context of current developments, it is necessary to perceive that the economies of many countries are declining and, hand in hand with this, the social situation of their inhabitants is deteriorating. Expectations are associated with this situation in the form of an increase in the number of foreclosures, an increase in the debt of individuals and families, a lower ability to save, etc. This leads to the possibility of follow-up research into the application of knowledge and skills in financial literacy in a time of crisis, where it is possible to examine which aspects mitigate the decline in private and corporate finance and which previously positively assessed aspects of financial literacy are failing.

The area of financial literacy is therefore far from exhausted, and the current state of events in society opens up new areas of research. As Ajaz Khan, Çera, & Pinto Alves (2022) stated, there is a need to fully understand the significant factors affecting financial capabilities.

7. CONCLUSION

In this paper, we focused on financial literacy research for students of economics (full-time and parttime - with a wider range of ages) who have access to education that supports the growth of financial literacy. We used the personal finance index (P-FIN index) as an instrument to measure financial literacy and we composed our questionnaire to cover all eight functional areas of financial literacy defined in the P-FIN index. We focused on the relationship between practical experience with financial market products and the level of financial literacy and we divided respondents into experienced and inexperienced groups. Individuals with a practical need to use the services of the financial market privately and/or professionally can be expected to have a deeper internal motivation for education and a practical grasp of our researched issues. We tested the hypothesis: Students with more practical experience with financial market tools and services achieve higher levels of financial literacy.

The results of our statistical analysis of the data clearly confirmed a higher percentage of students with a greater range of practical experience with financial sector products and services. This difference was reflected in the overall level of financial literacy as measured by the P-FIN index and also in the individual functional areas of financial literacy. We found a number of interesting facts. For example, young respondents do not tend to finance consumption through debt which can be perceived as positive. It is also surprising that not a single respondent mentioned the use of student loans.

However, the positive findings are connected with a certain weakness of our research. In the group of respondents having experience with debt instruments is our result not very conclusive due to the small size of this group. It was a group represented by younger respondents who do not yet have the need to incur debt. On the other hand, in all cases, our results confirm the validity of our research hypothesis. Another limitation of the research is associated with turbulent changes in society. The results of our research cannot fit into the period when the effects of rising energy prices, the looming energy crisis and the uncertainty of further economic development are economically apparent. This period will be a test of the practical application of knowledge and understanding of financial literacy in the management of private finances, but also in the adaptation of corporate finance and business economics as a whole to new conditions. E.g. for many family companies, you can find a habitual management style, which, however, does not correspond to the modern needs of financial management and the development of a competitive company.

We discussed the results of our research with the results of a number of current scientific articles, where we found certain similarities in the findings in areas such as risk perception, overconfidence factor, social environment factor, and the responsible approach of financially literate individuals to personal financial management and corporate finance management. These findings and the current economic situation open up further possibilities for follow-up research.

ACKNOWLEDGMENT

The paper was conducted within the project KEGA No 007ŽU-4/2021 Innovative courses to improve financial and economic literacy of the technical fields' students.

REFERENCES

- Ajaz Khan, K., Çera, G., & Pinto Alves, S. R. (2022). Financial Capability as a Function of Financial Literacy, Financial Advice, and Financial Satisfaction. *E+M Ekonomie a Management*, 25(1), 143–160. DOI: 10.15240/tul/001/2022-1-009
- Anderson, A., & Robinson, D. T. (2021). Financial Literacy in the Age of Green Investment. *Review of Finance*. DOI: 10.1093/rof/rfab031
- Bottazzi, L., & Lusardi, A. (2021). Stereotypes in financial literacy: Evidence from PISA. *Journal of Corporate Finance*, *71*, 101831. DOI: 10.1016/j.jcorpfin.2020.101831
- Fong, J. H., Koh, B. S. K., Mitchell, O. S., & Rohwedder, S. (2021). Financial literacy and financial decision-making at older ages. *Pacific-Basin Finance Journal*, 65, 101481. DOI: 10.1016/j.pacfin.2020.101481
- Fujiki, H. (2022). Household financial services, financial literacy, and COVID-19 pandemic. *Applied Economics Letters*, 29(7), 615–618. DOI: 10.1080/13504851.2021.1878092
- Gerth, F., Lopez, K., Reddy, K., Ramiah, V., Wallace, D., Muschert, G., ... Jooste, L. (2021). The Behavioural Aspects of Financial Literacy. *Journal of Risk and Financial Management*, 14(9), 395. DOI: 10.3390/jrfm14090395
- Hasan, M., Le, T., & Hoque, A. (2021). How does financial literacy impact on inclusive finance? *Financial Innovation*, 7(1), 40. DOI: 10.1186/s40854-021-00259-9
- Jin, J., Kanagaretnam, K., Liu, Y. I., & Cheng, M. (2021). Does Citizens' Financial Literacy Relate to Bank Financial Reporting Transparency? *European Accounting Review*, 30(5), 887–912. DOI: 10.1080/09638180.2021.1965897
- Kawamura, T., Mori, T., Motonishi, T., & Ogawa, K. (2021). Is Financial Literacy Dangerous? Financial Literacy, Behavioral Factors, and Financial Choices of Households. *Journal of the Japanese and International Economies*, 60, 101131. DOI: 10.1016/j.jjie.2021.101131
- Kim, H. H., Maurer, R., & Mitchell, O. S. (2021). How financial literacy shapes the demand for financial advice at older ages. *The Journal of the Economics of Ageing*, 20, 100329. DOI: 10.1016/j.jeoa.2021.100329
- Kozubik, A., Kozubíková, Z., & Polák, J. (2019). Financial Literacy of Full-Time and Part-Time University Students. *IJAEDU- International E-Journal of Advances in Education*, 35–43. DOI: 10.18768/ijaedu.478251
- Lusardi, A., Yakoboski, P. J., & Oggero, N. (2017). The TIAA Institute-GFLEC Personal Finance Index: A New Measure of Financial Literacy. New York.
- Okamoto, S., & Komamura, K. (2021). Age, gender, and financial literacy in Japan. *PLOS ONE*, *16*(11), e0259393. DOI: 10.1371/journal.pone.0259393
- Park, H., & Martin, W. (2021). Effects of risk tolerance, financial literacy, and financial status on retirement planning. *Journal of Financial Services Marketing*. DOI: 10.1057/ s41264-021-00123-y
- Petrů, N., Kramoliš, J., & Stuchlik, P. (2020). Marketing tools in the era of digitization and their Use in Practice by Family and other Businesses. *E+M Ekonomie a Management*, 23(1), 199–214. DOI: 10.15240/tul/001/2020-1-014
- Polák, J., Kozubíková, Z., & Kozubík, A. (2018). Financial Literacy of University Students and Effects of Practical Experience. In Socioint 2018: 5th International Conference on Education, Social Sciences and Humanities (pp. 17–26). Dubai: International Organization Center of Academic Research.
- Razen, M., Huber, J., Hueber, L., Kirchler, M., & Stefan, M. (2021). Financial literacy, economic preferences, and adolescents' field behavior. *Finance Research Letters*, 40, 101728. DOI: 10.1016/j.frl.2020.101728

- Riepe, J., Rudeloff, M., & Veer, T. (2022). Financial literacy and entrepreneurial risk aversion. *Journal of Small Business Management*, 60(2), 289–308. DOI: 10.1080/00472778.2019.1709380
- Sun, R., Zhang, H., Turvey, C. G., & Xiong, X. (2021). Impact of Financial Literacy on Retirement Financial Portfolio: Evidence from China*. *Asian Economic Journal*, 35(4), 390–412. DOI: 10.1111/asej.12256
- van Rooij, M., Lusardi, A., & Alessie, R. (2011). Financial literacy and stock market participation. *Journal of Financial Economics*, 101(2), 449–472. DOI: 10.1016/j.jfineco.2011.03.006
- Zhu, A. Y. F. (2020). Financial Literacy Types and Financial Behaviors Among Adolescents: Role of Financial Education. *Journal of Financial Counseling and Planning*, JFCP-19-00051. DOI: 10.1891/JFCP-19-00051


Preparation for the Career outside the Army: The View of Soldiers of the Czech Armed Forces

Renata Skýpalová¹ 💿 Kristýna Binková² 💿

Received: July 2, 2022 Accepted: August 24, 2022 Published: December 30, 2022

Keywords:

Military organization; Sociology; Soldiers; Second career; Labor market; Transition; Czech Armed Forces

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-Non-Commercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission. **Abstract:** The paper focuses on the means of support during soldiers' labor market integration after the termination of military contracts. To find out how the military personnel of the Czech Armed Forces cope with their training for a second career and whether they are interested in the support tools used abroad, the research included 176 students of military studies at the University of Defence and 284 soldiers in active military service. Broadening their knowledge and skills for a better opportunity in the labor market is the wish of 90 % of respondents. More than four-fifths of the respondents have not started training for their second career yet. According to the results, the respondents are interested in helping with preparation for a second career. However, compared to other NATO countries, a systematic approach is not being applied in the Czech Republic at this time.

1. INTRODUCTION

E very year a certain percentage of soldiers, get detached from the military due to retirement or various terms of service. Most of the soldiers who leave the army do not reach the statutory retirement age and belong to a relatively young population aged from 30 to 40 years. Hence, they remain with substantial productive age with a potential of working for 15 - 30 more years. Resettlement of soldiers becomes a challenge as they wish to maintain a certain living standard, financial security and expected quality of life for their families. Hence, they make an effort to build a new career in new civilian employment (Chopade & Gupta, 2020). Although the soldiers usually know the date of their terms of service, most of them do not prepare for this period or focus on rather unhelpful training activities. Unpreparedness is partly attributable to organizational barriers, limited initiative and motivation. This apparent lack of agency may be associated with the institutional nature of the military (Keeling et al., 2018).

However, career transition from armed forces to civilian life plays a prime role in the resettlement of military personnel (Chopade & Gupta, 2020; Alonso et al., 2021). Obtaining civilian employment after military separation not only provides financial stability but also a sense of purpose and meaning, which helps foster a new identity as a civilian (Kintzle & Castro, 2016). Conversely, a lack of employment following military separation is associated with poor mental health, increased stress, and financial insecurity (Zoli et al., 2015). Leaving military service creates an intense and major life transition, which requires a series of adjustments (Zoli et al., 2015; Chopade & Gupta, 2020; Geraci et al., 2020). Therefore, practical experience shows that it

² University of Defense, Faculty of Military Leadership, Department of Resources Management, Kounicova 65, Brno, Czech Republic



¹ AMBIS university, Department of Economics and Management, Lindnerova 575/1, Prague, Czech Republic

can be difficult for ex-servicemen to integrate into the civilian labor market (Kintzle et al., 2016; Keeling, 2018; Cooper et al., 2018; Aronson et al., 2019; Gonzalez & Simpson, 2021).

Although evidence suggests the majority of ex-servicemen eventually find civilian employment, some of them have persistent employment problems years after discharge (Keeling et al., 2019). Moreover, some ex-servicemen who had initial success in securing employment have jobs they are over-qualified for or uninterested in (Alonso et al., 2021) and later struggle to maintain their job (Keeling et al., 2018).

Almost half of post-9/11 veterans report that securing employment is their most significant challenge when transitioning from the military (Stern, 2017). For example, veterans were found to have a 58% higher odds ratio of being unemployed compared to matched civilian peers (Kleykamp, 2013). Research suggests that between 54% (Sayer et al., 2015) and 75% (Castro & Kintzle, 2017) of ex-servicemen report reintegration challenges, with approximately 80% leaving service without a job lined up and approximately two-thirds of veterans reporting not knowing where to go for assistance. This lack of assistance is problematic because ex-servicemen who feel supported during their transition report more successful transitions than those who lack support (Kintzle et al., 2016; Castro & Kintzle., 2017; Keeling et al., 2019).

Finding satisfying employment is a key aspect of successful transition (Ahern et al., 2015; Keeling et al., 2018). To assist with the transition out of the military and into the civilian workforce, transitioning service members must often rely on their networks of personal and professional contacts (Hall, 2017). James (2017) found that 10 out of 10 veterans reported networking as pivotal to securing civilian employment. Besides family (Bennett et al., 2015; Wilson, 2015), former colleagues (James, 2017) and civilians (Delbourg-Delphis, 2014; James, 2017), one of the most important circles of people for transitioning service members are formal resource persons, including professionals trained in helping them transition into civilian professions (Bennett et al., 2015; Alonso et al., 2021). Hence, it becomes the responsibility of the nation to utilize this capable and potential workforce for the betterment of society (Chopade & Gupta, 2020).

Studies have shown that planning and careful preparation for the second career period before the process of integration into the labor market is the only factor that provably facilitates the period of transition from the military to another career. The importance of employment preparedness to the well-being of individuals transitioning from one institution to another (military to civilian) has been noted in past research (Skomorovsky et al., 2020). According to Keeling et al. (2018) or Roy et al. (2020), preparation should not begin post-discharge but much earlier. Preparing for the transition well in advance (approximately 1 year before separation) can significantly reduce the difficulty of the transition (Ahern et al., 2015). Those who had adapted well were those who had carefully planned the transition. Keeling et al. (2019) found that the longer service members had to plan their transition, the easier the transition into the civilian workforce. Veterans from all service branches both employed and unemployed demonstrated that obtaining information about civilian careers before leaving the service eases the transition and lessens the time unemployed after separation (Keeling et al., 2019).

Individual responsibility and activity of the soldiers are essential, but the support of programs and services that help prepare transitioning service members with finding education and employment is necessary (Lee et. al, 2020). Facilitating the reintegration and ensuring a safe environment within which to find satisfying employment for military service members is a societal priority (Elnitsky et al., 2017).

The article aims to present:

- the literature review related to the second career of military professionals in order to appropriately introduce the research issue;
- the results of the content analysis of department-specific documents from foreign countries in order to summarize the approaches to the preparation and training of professional soldiers for a second career in selected NATO member states,
- the main findings from the authors' sociological survey that was carried out in order to find out how soldiers of the Czech Armed Forces approach their own preparation and training for a second career and whether they would be interested in the support tools used in foreign countries.

Following the aim of the survey, three research questions (RQ) were stated:

- **RQ 1:** Have the respondents already started preparing for their second career? Do the age of soldiers in active service, their length of a service contract and the remaining period of their service contract until the contracted date of termination plays a significant role in their current state of preparation for a second career?
- **RQ 2:** What are the job preferences of respondents in terms of a second career?
- **RQ 3:** What are the wishes and expectations of respondents pertaining to preparation for a second career? Are they interested in the support tools used in foreign countries?

2. THEORETICAL BACKGROUND

Studies suggest that service members' reintegration into civilian life has the potential to become a challenge for military personnel due to their deeply learned military self-identities (Grimell, 2016a). Years of being immersed in a rigid hierarchy and following orders may decrease one's ability to take initiative and rely on personal autonomy (Zarecky, 2014; Keeling et al., 2018). Leaving behind the structure of military existence, military culture, and comradeship to navigate civilian life in and of itself poses numerous challenges. Despite the fact that the majority of service leavers do not leave with physical injuries or mental health problems, many will have functioning issues that make reintegration difficult (Roy et al., 2020). While military standards, discipline and team spirit were valued by ex-servicemen and were an asset with regards to some employment opportunities, their absence in a civilian working context conversely resulted in some describing reentry into civilian employment straight out of services as a shock, often because of the divergence between the two cultures (Roy et al., 2020). Armed forces are "separate world" where people live a specific lifestyle, dress in a specific way and speak specific language. Military communication is marked by frequent use of acronyms, direct command-orientated exchanges, and nuances in vocabulary and colloquialisms. Mismatch in communication and being misunderstood can lead to feelings of invalidation, disconnection and uselessness (Ahern et al., 2015; Smith & True, 2014).

Barriers to employment include insufficient education or specialized military training that does not necessarily translate to the civilian world (Pease et al., 2016; Gordon et al., 2020). Military professionals master skills that have very little application in the civilian sector, and after approximately 15 years of military service, it is not realistic to expect they could get back to the vocation they had originally trained for (Gaither, 2014).

However, the qualities and habits that they establish during the service are highly positive and often even exceed the requirements of civilian employers. Their unique skill set, a vast amount of training, experiences, impeccable work ethics, and personal characteristics are beneficial to an organization (Minnis, 2017). Hiring an ex-serviceman conditionally presents the opportunity to secure an individual who is loyal, disciplined to mission, comprehends team success, appreciates hierarchy, respects orders and superiors and can make decisions under pressure, which can yield financial benefit to the organization (Stern, 2017; Pollak et al., 2019). However, although ex-servicemen were perceived as having greater leadership and teamwork skills, they were also viewed as having poorer social skills (Yanchus et al., 2018; Stone et al., 2018; Shepherd et al., 2019)

There are additional issues that can affect the employment of ex-servicemen. Research suggests, due to the possible unfamiliarity with military service, civilian employers may struggle to understand the different roles and training of military service, thus making it hard to assess how military experience may be transferrable and military skills translated to a civilian job (Stern, 2017; Roy et al., 2020). Moreover, due to the lack of awareness of military life among civilians and the increasing gap between civilians and the military (Hines et al., 2015; Liebert & Golby, 2017) civilian employers tend to treat ex-servicemen differently due to stereotypes they hold about the military (Shepherd, Kay, & Gray, 2019). They often believe that former soldiers are too regimented and that their strict adherence to autocracy and organizational policies will be ineffective or a poor fit for the organization (Stone & Stone, 2015). Former soldiers who gain employment tend to face discrimination, negative stereotypes, stigma, underemployment, identity strain, exclusion, and a lack of adjustment (Shepherd et al., 2019). For these and other reasons, many veterans struggle with integrating into a workplace as part of their transition into civilian life, which could include failing to find or maintain employment (Ford, 2017).

This speaks to the importance of advanced career planning (Wilson-Smith & Corr, 2019). The military literature has proposed and evaluated interventions and strategies aimed at helping ex-servicemen with regard to specific issues, such as developing desirable job market skills. Some government and private initiatives are helping military personnel in search of jobs and reemployment (Chopade & Gupta, 2020). Some countries have established various systems facilitating ex-servicemen integration into the labor market. These include usually governmental agencies collaborating with the defense ministries and other support organizations. Programs currently available offer employment services such as online job boards, job fairs, resume writing resources, career counseling, coaching, mentoring, training and certification programs, and formal networking opportunities (Aronson et al., 2019).

Ex-servicemen can utilize career counseling and mentoring services with experienced professionals. Career counselors help ex-servicemen translate their military experience into civilian terms so they can more effectively illustrate their relevant expertise on resumes and in interviews with hiring managers (Delbourg-Delphis, 2014). They help them set realistic expectations regarding salaries, learn about their career goals, identify their strengths and weaknesses and provide them with actionable steps to identify and achieve appropriate vocational outcomes (Buzzetta et al., 2017). An important factor for a successful transition from military to civilian life often involves the obtainment of educational goals to support the future obtainment of meaningful employment (Ainspan et al., 2018). Some organizations, therefore, attempt to meet the needs of ex-servicemen by providing the necessary training and certification for civilian careers (Owings et al., 2015). There is some evidence that participation in employment-related programs yields positive results and has its intended effects which can be seen for example in a low rate of unemployment reported by participants (Curry Hall et al., 2014; Kerrick et al., 2016; Kerrick et al., 2014; Perkins et al, 2019). Scholars have suggested that veteran-serving organizations should develop and implement a strategy to determine what types of programs need to be offered. (Batka & Hall, 2016; Perkins et al., 2019). Ex-servicemen should be taken into account, current understanding of transition and transition support services evaluated, and improved (Roy et al., 2020). Ex-servicemen may feel more confident and supported, and less abandoned, if there is visible, strong leadership at the highest levels of government.

The Ministry of Defense of the Czech Republic offers soldiers whose service contract lasted five or more years various retraining courses aimed at preparation for the civilian environment and allowing them to obtain such knowledge and skills that they could apply to the labor market. However, the maximum allowed period the soldiers can spend in the retraining courses, which is 3 months, is not always sufficient for attaining new professional skills. It is very difficult to fully reskill a specially trained professional in a such short period. Moreover, most of the retraining courses focus just on basic knowledge and skills required for the jobs for which the soldiers retrain, and this is insufficient for the actual labor market. Some of the departmental companies offer soldiers exiting active service opportunities for further employment. These most commonly include positions of technicians, aircraft mechanics, or cooks. However, this is a benefit primarily aimed at current war veterans (Binková, 2018).

In the operational program Human Resources and Employment, which had enabled the use of funds from the European Social Fund in human resources during the 2007 - 2013 program period, seven projects focused on employing war veterans. Their drawback, however, was their short-term functioning and the strict focus on war veterans, whereby soldiers without previous participation in military missions could not take part in the activities offered under the projects (Binková & Bednář, 2017).

According to previous research, on average 18 % of former soldiers of the Czech Armed Forces of productive age remained unemployed after they terminated military service for 8 months. This is a high percentage, which is more than three times higher than the average general unemployment rate in the Czech Republic in this period, which was 5.7% (Binková, 2019).

2.1. Examples of Good Practice from Selected NATO Member States

To assist professional soldiers with integrating into labor market, in some countries, various systems are facilitating the soldiers' transition to their second career. Most of them are provided by governmental agencies collaborating with the ministries of defense, as well as by other support organizations providing counselling, training, and mediating contact with the civilian labor market.

A role model in this aspect is France. Since 2009, the French Defense Mobility Agency (Défence Mobilité) has been a part of the workforce policy of the Ministry of Defense supporting external mobility of the military personnel and enhancing the attractiveness of professional soldiers on the labor market, both to public and private organizations. The agency draws from a network of over 700 local military and civilian experts in retraining, with excellent knowledge of the specific needs of both the candidates and employers and cooperates with more than 7,000 public

and private corporations. The agency monitors professional soldiers, provides professional assistance with developing their professional projects, expert training meeting their needs, guides them, and connects with potential employers. Every year, more than 16,000 candidates use the services of the agency. The headquarters of the agency is based in Paris; there are six regional centers, 57 local offices (52 in France and 5 overseas), one retraining institute for officers, one military training center, and one center for unemployment support. The agency provides its services based on the principle of equality, regardless of the military ranking or position in the army (Bilan reconversion, 2014).

| Table 1. Comparison of support tools in selected 14110 memoer countries | | | | | | | |
|---|-------------------|--------|--------|------------------|---------|--|--|
| Support tools | Czech Republic | Poland | France | Great Britain | Germany | | |
| Advisory | | Х | Х | Х | Х | | |
| Seminars | | Х | Х | Х | Х | | |
| Vocational education: training, reskilling, study | Х | Х | Х | х | х | | |
| Assistance with job seeking | | Х | Х | Х | X | | |
| Supported job opportunities | | Х | Х | Х | Х | | |
| Professional experience | | Х | Х | Х | х | | |
| Job fairs | | Х | Х | Х | X | | |
| Help with starting a business | | Х | Х | Х | X | | |
| Special job portal | | | Х | Х | X | | |
| Financial compensation, coverage of activities related to preparation for second career | Х | Х | Х | Х | х | | |

Table 1. Comparison of support tools in selected NATO member countries

Source: Own research

Great Britain has a similar approach. The Career Transition Partnership (CTP) project runs in collaboration between the Ministry of Defense and Right Management Ltd. Corporation, which provides full service to exiting soldiers for four years – two years before the end of their service contract and two years afterward. The aim is to provide full assistance to help the soldiers integrate into the labor market and to prepare them for future career growth. Throughout its existence, the agency has helped more than 200,000 soldiers. The agency runs one training center and nine contact centers that cooperate with employers interested in employing soldiers exiting the army. Support is provided to all soldiers, regardless of their ranking and length of their contract period, but the level of support differs upon the meeting of a set of predefined criterions. During the project, the soldiers can choose from a range of retraining courses, career advisories, or services of a special job portal (The Career Transition Partnership, 2018).

Germany is yet another example of good practice. Law governs the preparation of soldiers for a second career and it pertains to soldiers in active service as well as to soldiers whose service contract had already terminated. Berufsförderungsdienst der Bundeswehr, i.e. the agency responsible for preparing the soldiers for the second career, employs approximately one thousand civilians in 46 branches across all the military districts. The services of this institution ensured a planned and systematic preparation for the change of the soldiers' working environment, which begins with basic training and runs throughout the entire duration of one's service contract as well as after its termination. An important role in the expert preparation of professional soldiers for the needs of the labor market is played also by the Bundeswehrfachschule, which is a so-called "second chance" vocational institute established specifically for professional soldiers and training them for their second career (Jonsson, 2014). In Poland, there are one central and seven regional offices of The Professional Activation Center (Ośrodek Aktywizacji Zawodowej), which systematically prepares soldiers for a second career. The center provides information and training services, career advisory, help with finding employment and support during starting up a business; it offers internships and reskilling courses, job fairs, and other opportunities (Rekonwersja kadry i byłych żołnierzy, 2017).

A comparative analysis of the provided tools to support the integration of soldiers into the labor market in individual countries was performed – see Table 1.

3. METHODOLOGY

To find out how the soldiers of the Czech Armed Forces approach their preparation and training for a second career, the authors of this paper performed a questionnaire survey utilizing the Google Forms application. After data acquisition, the answers were coded and entered into a data matrix in Microsoft Excel. IBM SPSS Statistics 25 software was used to verify the trueness of the statistical hypotheses.

The respondents in the total number of 460 (which is approximately 2% of the total number of members of the army) comprised two groups:

- students of military studies at the University of Defence UD students (176 respondents, of which 141 men and 35 women). The age of respondents from the group of UD students varied from 19 up to 26 years and by military ranking, they included all the ranks from private first class up to staff sergeant.
- professional soldiers of the Czech Armed Forces in active service (284 respondents from 43 divisions of the armed forces, of which 260 men and 24 women). The age of respondents from the group of professional soldiers varied from 20 up to 51 years and by military ranking, they included all the ranks from private first class up to colonel. By achieving degree of education among professional soldiers, there were soldiers with a high school education (63 %) and a university degree (37 %). Besides the basic sociodemographic variables, the survey covered also the length of service employment and the period until the soldiers' expected exit from service employment. The length of service was as follows: up to five years of service (2.4 %) 6 10 years (7.3 %), 11 15 years (11.8 %), 16 20 years (17.4 %), and 21 years and more (27.9 %). The average remaining period of service was 3 years, with the shortest period of nine months and the longest of 10 years.

This research sample comprised both military professionals and military students, especially with regard to the priority measures of The Czech Armed Forces Development Concept 2030. It aims at increasing the attractiveness of military service, the motivation and the fulfillment of social needs of the personnel in all categories, declaring the University of Defence will be a high priority (Ministry of Defense & Armed Forces of the Czech Republic, 2021). The studied sample was not homogeneous in terms of sex, but this composition corresponds with the proportion of men and women in the studied reference population. Due to the very low percentage of women in the studied group, the factor of sex was not considered a discriminatory sign.

4. **RESULTS AND DISCUSSION**

RQ 1: Have the respondents already started preparing for their second career? Do the age of soldiers in active service, their length of the service contract and the remaining period of their service contract until the contracted date of termination plays a significant role in their current state of preparation for a second career?

Only 19 % of the respondents (23 % of professional soldiers in active service and 12 % of UD students) have already started their own preparation for a second career. More than four-fifths of the total number of respondents are not preparing for their second career yet. This finding is in line with the previous study by Keeling et al., (2018).

Table 2. Relation between the current state of preparation for a second career and age,duration of service employment and the remaining period of service contract until thecontracted date of termination – significance threshold 0.05

Statistical hypothesis 1: *Among soldiers in active service, there is no statistically significant correlation between the current state of preparation for second career and their age.*

| Preparation for second career | Ν | Average age | Levene's test for homogeneity of variance t-test | | | | |
|-------------------------------|-----|-------------|---|---------|--------|------------------------|-----------------------------------|
| Yes | 65 | 33.26 | F | p-value | t | p-value (bilateral) | Difference between averages |
| No | 219 | 33.96 | 0.252 | 0.616 | -0.635 | 0.526 | -0.702 |

Statistical hypothesis was not rejected.

Statistical hypothesis 2: Among soldiers in active service, there is no statistically significant correlation between the current state of their preparation for the second career and the length of their service contract.

| Preparation for second career | Ν | Average length of service contract in years | | ne's test eity of variance | | t-test | |
|-------------------------------|-------------|---|-------|-------------------------------|--------|------------------------|-----------------------------------|
| Yes | 65 | 9.65 | F | p-value | t | p-value (bilateral) | Difference between averages |
| No | 219 | 11.3639 | 2.468 | 0.117 | -1.418 | 0.157 | -1.714 |
| Statistical hypoth | onio tuto o | not rejected | | | | | |

Statistical hypothesis was not rejected.

Statistical hypothesis 3: Among soldiers in active service, there is no statistically significant correlation between the current state of preparation for second career and the remaining period of service employment until the contracted date of expiry.

| Preparation for second career | N | Average remaining period of service in years | Levene's test for homogeneity of variance | | | t-test | |
|--|-----|---|--|---------|--------|------------------------|-----------------------------------|
| Yes | 65 | 2,74 | F | p-value | t | p-value (bilateral) | Difference between averages |
| No | 219 | 3.1482 | 3.276 | 0.071 | -1.963 | 0.051 | -0.40817 |
| <u><u><u></u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u> | • | 1 | | | | | |

Statistical hypothesis was not rejected.

Source: Own research

Many authors examine the role of various factors affecting the course of preparation for a second career and the success of a soldier's integration into the civilian sector. E.g. age, sex, health status, length of a service contract, number of deployments, the remaining period of a service contract until the contracted date of termination or support of the family (Brunson, 1996; Vigoda-Gadot et al., 2010; Morin, 2011; Robertson, 2013; Hachey et al., 2016). Therefore, the group of soldiers in active service was further studied in terms of the relation between the current state of preparation for a second career and age, duration of service employment and the remaining period of their service contract until the contracted date of termination. To verify the individual statistical hypotheses, a two-sided (non-paired) Student t-test was used. With the presumption of homoscedasticity of variance, a test of the mean values agreement hypotheses was performed. Table 2 presents the results of the tests.

Statistical hypotheses were not rejected. Among soldiers in active service, there is no statistically significant correlation between their current state of preparation for a second career and their age, the duration of their service employment and the remaining period of their service contract until the contracted date of termination.

RQ 2: What are the job preferences of respondents in terms of a second career?

Struggling with civilian work culture points to the need for job roles in organizations that share a similar ethos of structure, discipline, and teamwork (Roy et al., 2020). The largest group of respondents -39 % (45 % of soldiers in active service and 29 % of UD students) wish to find a job opportunity in the public sector after terminating their service contract. Forty-nine percent of all respondents (58 % of UD students, and 43 % of professional soldiers in active service) want to be employed in the security forces (Police and Fire Rescue Service of the Czech Republic). This result is consistent with studies that show veterans pursue careers in the public sector and protective services jobs (Schulker, 2017). In the second position the preference of the target groups, differ. Whereas the students wish to start their own businesses, soldiers in active service prefer to work in the private sector.

RQ 3: What are the wishes and expectations of respondents pertaining to preparation for a second career? Are they interested in the support tools used in foreign countries?

Twenty-four percent of the respondents believe that preparation for a second career needs to start about 4 to 6 months before the end of their service employment. Twenty-three percent of the respondents state one year to 1.5 years before the end of their service employment, and 22 % of the respondents state 7 months to 1 year to the end of their service employment. Fourteen percent of the respondents state the need for continuous preparation throughout the entire military career; other responses showed a low percentage. This finding confirms previous studies emphasizing the importance of early preparation for a second career, starting well in advance before the separation from the army (Ahern et al., 2015; Keeling et al., 2018; Keeling et al., 2019; Wilson-Smith & Corr, 2019; Skomorovsky et al., 2020; Roy et al., 2020).

If an ex-serviceman is able to manage the idea of an occupational transition from military to college and is willing to engage in specific occupational engagement tasks, he or she will also be ready to accomplish the tasks needed to achieve career-planning activities (Ghosh & Fouad, 2015). The vast majority of respondents (90 %) wish to extend their knowledge and skills during their service employment in order to find better opportunities in the labor market. 47 % of all respondents prefer retraining courses, and 29 % prefer university studies. Among the most frequently stated fields in which the respondents wish to improve, are foreign languages, IT and computer skills, management and leadership, electrical engineering, and training for driving licenses.

Seventy-two percent of respondents would welcome, during their service employment, an individual approach focused on training for further career opportunities. They could also rate three 8th International Scientific Conference ERAZ 2022 Selected Papers

specific forms of such individual assistance on a scale of 1 to 5 (1 = the least useful, 5 = the most useful). According to the arithmetic average of achieved point rating respondents highlighted coaching as the most useful activity (4), while the help of a psychologist achieved the lowest score (2.4) and career counselling was rated 3.7. A job coach can help ex-servicemen learn and develop additional skills necessary to succeed in their civilian roles. Specific approaches to coaching can be used, for example, Zarecky's model (2014), focusing on soldiers' strengths.

Fifty-six percent of the respondents are interested in organizing group workshops focusing on preparation for a further career during their service employment. The topic of the highest interest is the principles of preparation and behavior at a job interview and self-presentation techniques. Research shows the usefulness of workshops of this kind. For example, a pre-and posttest comparison of results of 1315 young veterans who participated in a 5-day workshop aimed at facilitating their transition to civilian life showed that participation in the workshop reduced the participants' career decision-making difficulties and increased their career decision self-efficacy (Gati, Ryzhik, Vertsberger, 2013).

After leaving the service, those in transition may join one or more possible career pathways employment, further education, or business start-up (Gray et al., 2017). Therefore, if the soldiers, before the expiry of their service employment, had an opportunity of consultations aimed at support during the start-up of their business as one of the possible directions of their second career, 62 % of the respondents would like to use it.

Both target groups expressed great interest in job fairs - 83 % of the respondents. As previous studies suggest, job fairs are networking-specific events that are available to assist ex-servicemen in exploring career options and obtaining employment. Beyond giving participants direct personal interaction with potential employers, job fairs may help participants determine if a specific employment opportunity is a good fit for them (James, 2017; Stonebraker et al., 2019).

Special websites, job portals, or social media sites, such as LinkedIn and Facebook, are often used by ex-servicemen to identify job leads (Weinburger et al., 2015; Hall, 2017).

Services of a job portal for professional soldiers, which would allow them to contact potential employers, would be used by 82 % of the respondents. Regular publication of job offers from employers was found as the most beneficial feature of this portal.

Respondents were also asked whether they would be interested in going on an unpaid expert internship of 3-6 months with a civilian employer during their service employment; during this time, they would continue receiving their service salary. Seventy-four percent of the respondents are interested in increasing their chances of employment through an expert internship in a civilian organization. The respondents also stated whether they would wish other organizations from the civilian sector (not just those under the Ministry of Defense) to offer job opportunities for ex-professional soldiers. Seventy percent of the respondents expressed their wish that organizations from the public sector offer job opportunities for ex-soldiers. The respondents were then asked to state which institutions should be involved in such activity; the results suggest respondents would be predominantly interested in security units of the state, units of the integrated rescue system, public authorities and state authorities, municipal, local, and regional authorities, healthcare institutions and educational institutions.

5. FUTURE RESEARCH DIRECTIONS

The current results may form a basis for further research on the larger sample, as this issue will be constantly addressed by all the countries across NATO.

6. CONCLUSION

More than four-fifths of the respondents from the target group of University of Defence (UD) students and soldiers in active service have not started any preparation for their second career yet. Among soldiers in active service, the current approach to preparation for the second career does not depend on age, length of their service contract, or the remaining period of their service contract until the contracted date of termination. Almost one-quarter of the respondents think that the preparation for a second career should start 4 to 6 months before the termination of the service employment. Approximately the same percentage thought that year to 1.5 years, or 7 months to 1 year before the termination of the contract period were sufficient. The option of 1 to 3 months, which meets the current condition of attending retraining courses in the maximum length of 3 months, was stated by only 8 % of the respondents. Broadening their knowledge and skills for a better opportunity in the labor market is the wish of 90 % of respondents. Almost half of the respondents from both the target groups prefer attending retraining courses and obtaining certificates or diplomas, while nearly one-third prefer a study at a university and obtaining a degree in a civilian field.

Nearly four-quarters of respondents would be interested in an individual approach focused on preparation for further career development, most often in the form of coaching. Almost two-thirds of the respondents expressed their interest in attending group seminars focused on the preparation for their further careers. All of them express the greatest interest in learning about the principles of behavior at a job interview and self-presentation techniques. Two-thirds of respondents are interested also in consultations focusing on assistance with starting up their own business as a potential direction of their second career. More than four-fifths of the respondents wish to attend job fairs organized specifically for professional soldiers, as well as to have a specialized job portal, the most useful function of which would be the direct publishing of job offers from employers. Three-quarters of respondents would be interested in a 3 - 6 months unpaid internship in a civilian company, and almost three-quarters of the respondents wish that organizations from the public sector communicated job offers targeted specifically at ex-servicemen. Examples of such organizations included most commonly the security units of the state, integrated rescue system units, public authorities and municipalities, healthcare and educational institutions, and security agencies.

The focus on the Czech labor market, the situation of the professional soldiers' second career, and the relatively low number of respondents limit the general conclusions. However, the current results are of high interest to the Ministry of Defense of the Czech Republic because based on the performed analysis retraining is insufficient and a systematic approach is not being applied in the Czech Republic at this time. One of the objectives of The Czech Armed Forces Development Concept 2030 is to cooperate with the civilian sector and improve the offer of retraining in order for a better reintegration of the outgoing personnel into civilian life. It is therefore necessary to examine the attitudes of all categories of soldiers to ensure that this goal will be met in the future.

The results should be of interest also to HR specialists who are responsible for managing the workforce in various civilian organizations and may use the workforce of military professionals exiting their active service in the army.

REFERENCES

- Ahern, J., Worthen, M., Masters, J., Lippman, S. A., Ozer, E. J., & Moos, R. (2015). The challenges of Afghanistan and Iraq veterans' transition from military to civilian life and approaches to reconnection. *PloS One*, 10(7), 1–13. DOI: 10.1371/journal.pone.0128599
- Ainspan, N. D., Penk, W., & Kearney, L. K. (2018). Psychosocial approaches to improving the military-to-civilian transition process. *Psychological services*, 15(2), 129. DOI: 10.1037/ ser0000259
- Alonso, N. A., Porter, C. M., & Cullen-Lester, K. (2021). Building effective networks for the transition from the military to the civilian workforce: Who, what, when, and how. *Military Psychology*, 33(3), 152-168. DOI: 10.1080/08995605.2021.1897489
- Aronson, K. R., Perkins, D. F., Morgan, N., Bleser, J., Davenport, K., Vogt, D., & Gilman, C. L. (2019). Going it alone: Post-9/11 veteran nonuse of healthcare and social service programs during their early transition to civilian life. *Journal of Social Service Research*. DOI: 10.1080/01488376.2018.1493410
- Batka, C., & Hall, K. C. (2016). What's Good for Business and for Veterans. Perspective.
- D., Wellman, G., Mahmood, M., Freye, R., Remund, D., & Samples, P. (2015). Survey of retired military pharmacist's transition to a civilian pharmacy career path. *Military Medicine*, 180(12), 1219–1224. DOI:10.7205/MILMED-D-14-00531
- Bilan reconversion 2014. (2015). Paris: Défense Mobilité.
- Binková, K. (2019). Applicability of Former Soldiers at the Labour Market. *Czech Military Review*, 28(2), 82-98. DOI: 10.3849/2336-2995.28.2019.02.082-098
- Binková K. (2018). Retraining of Soldiers. Czech Military Review. 27 (4), 105-119. DOI: 10.3849/2336-2995.27.2018.04.105-119
- Binková K. & Bednář Z. (2017). Outplacement in the Defence Sector Environment. Czech Military Review, 26(1), 22-39. DOI: 10.3849/2336-2995.26.2017.01.022-039
- Brunson, B. H. (1996). Life satisfaction and retirement: Military mid-life career change. (Doctoral dissertation, Texas Tech University, 1996). Dissertation Abstracts International, 57(7-A).
- Buzzetta, M., Hayden, S. C. W., & Ledwith, K. (2017). Creating hope: Assisting veterans with job search strategies using cognitive information processing theory. *Journal of Employment Counseling*, 54(2), 63–74. DOI: 10.1002/joec.12054
- Career Transition Partnership. (2018). GOV.UK United Kingdom: Ministry of Defence. Retrieved February 10, 2021, from https://www.gov.uk/guidance/career-transition-partnership
- Castro C. A. & Kintzle S. (2017). The state of the American veteran: The San Francisco Veterans Study. USC Suzanne Dworak-Peck School of Social Work, Center for Innovation and Research on Veterans & Military Families. Retrieved April 14, 2021 from http:// joiningforces.issuelab.org/resources/30042/30042.pdf.
- Cooper, L., Caddick, N., Godier, L., Cooper, A., & Fossey, M. (2018). Transition from military into civilian life: An exploration of cultural competence. *Armed Forces & Society*, 44(1), 156-177. DOI: 10.1177/0095327X16675965
- Curry Hall, K., Harrell, M. C., Bicksler, B., Stewart, R., & Fisher, M. P. (2014). *Connecting veterans and employers*. Santa Monica, CA: RAND Corporation. DOI: 10.7249/RB9829
- Delbourg-Delphis, M. (2014). A relational approach to hiring veterans. *Employment Relations Today*, 41(1), 11–17. DOI: 10.1002/ert.21438
- Elnitsky, C. A., Blevins, C. L., Fisher, M. P., and Magruder, K. (2017). Military service member and veteran reintegration: a critical review and adapted ecological model. Am. J. Orthopsychiatry, 87, 114–128. DOI: 10.1037/ort0000244

- Elrond, A., Stoltenberg, C., Nissen, L., Nielsen, A., & Pedersen, J. (2019). Labor market affiliation after deployment: Danish soldiers fare well, but face increased risk of long-term sickness absence when returning from Afghanistan. *Scandinavian journal of work, environment & health*, 45(3), 289-297. DOI: 10.5271/sjweh.3773
- Ford, D. G. (2017). Talent management and its relationship to successful veteran transition into the civilian workplace: Practical integration strategies for the HRD professional. *Advances in Developing Human Resources*, 19, 36–53. DOI: 10.1177/1523422316682736
- Gaither, D. (2014). Military transition management. *Career Planning and Adult Development Journal*, 30(3), 215.
- Gati, I., Ryzhik, T., & Vertsberger, D. (2013). Preparing young veterans for civilian life: The effects of a workshop on career decision-making difficulties and self-efficacy. *Journal of Vocational Behavior*, 83(3), 373-385. DOI: 10.1016/j.jvb.2013.06.001
- Geraci, J., Murray, C., Kapil-Pair, K.N., Herrera, S., Sokol, Y., Cary, J., Landa, Y., Goodman, M., 2020. The modern-day Odysseus: how mental health providers can better reintegrate modern warriors and mitigate suicide risk. *Journal of Clinical Psychology*. 76 (5), 878– 895. DOI: 10.1002/jclp.22923.
- Ghosh, A., & Fouad, N. A. (2016). Career transitions of student veterans. *Journal of Career Assessment*, 24(1), 99-111. DOI: 10.1177/1069072714568752
- Gonzalez, J. A., & Simpson, J. (2021). The workplace integration of veterans: Applying diversity and fit perspectives. *Human Resource Management Review*, 31(2), DOI: 10.1016/j. hrmr.2020.100775
- Gordon, K., Burnell, K., & Wilson, C. (2020). Outside the military "bubble": Life after service for UK ex-armed forces personnel. Frontiers in public health, 8. DOI: 10.3389/ fpubh.2020.00050
- Gray, A., Wilson, R., Jenkings, K. N., Harrison, D., & Martin, M. (2017). Information-sharing in services for military personnel in transition to civilian life. *Public Money & Management*, 37(1), 15-22. DOI: 10.1080/09540962.2016.1249226
- Grimell, J. (2016a). The story of the self in the aftermath of crisis: A case study. *Journal of Constructivist Psychology*, 29(1), 66–79. DOI: 10.1080/10720537.2015.1079509
- Hachey, K. K., Sudom, K., Sweet, J., MacLean, M. B., & VanTil, L. D. (2016). Transitioning from military to civilian life: The role of mastery and social support. *Journal of Military*, *Veteran and Family Health*, 2(1), 9-18. DOI: 10.3138/jmvfh.3379
- Hall, A. F., Jr. (2017). Veterans stories: A narrative inquiry research study examination of veterans navigating second careers (Unpublished doctoral dissertation). ProQuest Dissertations & Theses Global. (Order No. 10258880).
- Hamilton, A. B., Williams, L., &Washington, D. L. (2015). Military and mental health correlates of unemployment in a national sample of women veterans. *Medical Care*, 53, S32– S38. DOI: 10.1097/mlr.0000000000297
- Hines, L. A., Gribble, R., Wessley, S., Dandeker, C., & Fear, N. T. (2015). Are the armed forces understood and supported by the public? A view from the United Kingdom. *Armed Forc*es & Society, 41(4), 688-713. DOI: 10.1177/0095327X14559975
- Chopade, M. P., & Gupta, K. S. (2020). A Soldier's Preparation for Second Innings–A Review. International Journal of Environmental Science and Technology 29(12s):2480-2485
- James, A. (2017). A phenomenological study: Female veteran commissioned officers' successful transition to the civilian workforce (Unpublished doctoral dissertation). ProQuest Dissertations & Theses Global. (Order No. 10267210)
- Jonsson, M. (2014). What do you know? Easing Soldiers' transition to the civilian labour market by making knowledge understandable. Sweden, University of Gothenburg.

- Keeling, M. (2018). Stories of transition: US veterans' narratives of transition to civilian life and the important role of identity. Journal of Military, *Veteran and Family Health*, 4(2), 28–36. DOI: 10.3138/jmvfh.2017- 0009
- Keeling, M., Ozuna, S. M., & Millsap, C. (2018). Employment after the military. In E. L. Weiss & C. A. Castro (Eds.), *American military life in the 21st century: Social, cultural, and economic issues and trends* (pp. 497–509). Santa Barbara, CA: ABC-CLIO, LLC.
- Keeling, M., Ozuna, S. M., Kintzle, S., & Castro, C. A. (2019). Veterans' civilian employment experiences: Lessons learnt from focus groups. *Journal of Career Development*, 46(6), 692–705. DOI: 10.1177/0894845318776785
- Keeling, M., Kintzle, S., & Castro, C. A. (2018). Exploring US veterans' post-service employment experiences. *Military Psychology*, 30(1), 63-69. DOI: 10.1080/08995605.2017.1420976
- Kintzle, S., Rasheed, J. M., & Castro, C. A. (2016). *The state of the American veteran: The Chicagoland veterans study.* Los Angeles, USA: University of Southern California
- Kleykamp, M. (2013). Unemployment, earnings and enrollment among post 9/11 veterans. *Social Science Research*, 42(3), 836–851. DOI: 10.1016/j.ssresearch.2012.12.017
- Lee, J. E., Dursun, S., Skomorovsky, A., & Thompson, J. M. (2020). Correlates of perceived military to civilian transition challenges among Canadian Armed Forces Veterans. *Jour*nal of Military, Veteran and Family Health, 6(2), 26-39. DOI: 10.3138/jmvfh-2019-0025
- H. & Golby, J. (2017). Midlife crisis? The all-volunteer force at 40. Armed Forces & Society, 43(1), 115-138. DOI: 10.1177/0095327X16641430
- Ministry of Defence & Armed Forces of the Czech Republic (2021). The Czech Armed Forces Development Concept 2030. Retrieved August 12, 2021 from https://www.army.cz/images/id_8001_9000/8503/CAFDC.PDF
- Minnis, S. E. (2017). Preface: Veterans in career transition and employment. *Advances in Developing Human Resources*, 19(1), 3–5. DOI: 10.1177/1523422316682951
- Morin, R. (2011). *The difficult transition from military to civilian life*. Washington, DC: Pew Research Center.
- Owings, W. A., Kaplan, L. S., Khrabrova, I., & Chappell, S. (2015). Troops to teachers update: Changing, but still pleasing principals with high teaching quality. *NASSP Bulletin*, 99(1), 70–98. DOI: 10.1177/0192636515571933
- Pease, J. L., Billera, M., & Gerard, G. (2016). Military culture and the transition to civilian life: Suicide risk and other considerations. *Social Work*, 61(1), 83–86. DOI: 10.1093/sw/swv050
- Perkins, D. F., Aronson, K. R., Morgan, N. R., Bleser, J. A., Vogt, D., Copeland, L. A., Gilman, C., et al. (2019). Veterans' use of programs and services as they transition to civilian life: Baseline assessment for the Veteran Metrics Initiative. *Journal of Social Service Research* (pp. 1–15). DOI: 10.1080/01488376.2018.1546259
- Pollak, M., Arshanapalli, B., & Hobson, C. (2019). The business case for hiring military veterans/reservists: Stock price performance of military friendly firms. Journal of Veterans Studies, 4(2), 52. DOI: 10.21061/jvs.v4i2.99
- Rekonwersja kadry i byłych żołnierzy. (2017). Wojskowa Komenda Uzupełnień w Radomiu. Retrieved February 13, 2021 from http://radom.wku.wp.mil.pl/pl/6786.html
- Robertson, H. C. (2013). Income and support during transition from a military to civilian career. *Journal of Employment Counseling*, 50, 26-33.
- Roy, D., Ross, J., & Armour, C. (2020). Making the transition: How finding a good job is a risky business for military Veterans in Northern Ireland. *Military Psychology*, 32(5), 428-441. DOI: 10.1080/08995605.2020.1785805
- Sayer, N. A., Orazem, R. J., Noorbaloochi, S., Gravely, A., Frazier, P., Carlson, K. F. & Oleson, H. (2015). Iraq and Afghanistan war veterans with reintegration problems: differences by

veterans affairs healthcare user status. *Administration and Policy in Mental Health and Mental Health Services Research*, 42(4), 493-503. DOI: 10.1007/s10488-014-0564-2.

- Shepherd, S., Kay, A. C., & Gray, K. (2019). Military veterans are morally typecast as agentic but unfeeling: Implications for veteran employment. *Organizational Behavior and Human Decision Processes*, 153, 75–88. DOI: 10.1016/j.obhdp.2019.06.003
- Schulker, D. (2017). The recent occupation and industry employment patterns of American veterans. *Armed Forces & Society*, 43(4), 695-710. DOI: 10.1177/0095327X16659875
- Skomorovsky, A., McCuaig Edge, H. J., Lee, J. E., Wan, C., & Dursun, S. (2020). Military to civilian transition challenges, caregiving activities, and well-being among spouses of newly released Canadian Armed Forces Veterans. *Journal of Military, Veteran and Family Health*, 6(1), 15-27. DOI: 10.3138/jmvfh-2019-0026
- Smith, R. T., & True, G. (2014). Warring identities: Identity conflict and the mental distress of American veterans of the wars in Iraq and Afghanistan. Society and Mental Health, 4, 147–161. DOI: 10.1177/2156869313512212
- Stern, L. (2017). Post 9/11 veterans with service-connected disabilities and their transition to the civilian workforce: A review of the literature. *Advances in Developing Human Resources*, 19, 66–77. DOI: 1177/1523422316682928
- Stone, C., & Stone, D. L. (2015). Factors affecting hiring decisions about veterans. Human Resource Management Review, 25(1), 68-79. DOI: 10.1016/j.hrmr.2014.06.003
- Stone, C. B., Lengnick-Hall, M., & Muldoon, J. (2018). Do stereotypes of veterans affect chances of employment? The Psychologist-Manager Journal, 21, 1–33. DOI: 10.1037/mgr0000068
- Stonebraker, I., Maybee, C., & Chapman, J. (2019). Undergraduate students' experiences of using information at the career fair: A phenomenographic study conducted by the libraries and career center. *The Journal of Academic Librarianship*, 45(4), 358–367. DOI: 10.1016/j. acalib.2019.05.002
- Vigoda-Gadot, E., Baruch, Y., & Grimland, S. (2010). Career transitions: An empirical examination of second career of military retirees. *Public Personnel Management*, 39(4), 379-404.
- Weinburger, C. M., Strider, S. H., & Vengrouskie, E. F. (2015). Social media job searches and the strength of veteran ties. *Human Resource Management Research*, 5(6), 133–153. DOI: 10.5923/j.hrmr.20150506.01
- Wilson, P. A. (2015). *The experience of enlisted female veterans transitioning from the military to the civilian workforce* (Unpublished doctoral dissertation). ProQuest Dissertations & Theses Global. (Order No. 3687617).
- Wilson-Smith, K. M., & Corr, P. J. (2019). *Help with finding a civilian job. In Military Identity and the Transition into Civilian Life* (pp. 79–98). Cham, Switzerland: Palgrave Pivot; Cham: SpringerNature.
- Yanchus, N. J., Osatuke, K., Carameli, K. A., Barnes, T., & Ramsel, D. (2018). Assessing workplace perceptions of military veteran compared to nonveteran employees. *Journal of Veterans Studies*, 3(1), 37-50. DOI: 10.21061/jvs.3
- Zarecky, A. (2014). How strengths-focused coaching can help military personnel in their transition to 'civvy street'. *International Journal of Evidence Based Coaching and Mentoring*, Spec Issue 8, 54–66.
- Zoli, C., Maury, R., & Fay, D. (2015). Missing perspectives: Servicemembers' transition from service to civilian life — data-driven research to enact the promise of the post-9/11 GI bill. Syracuse, NY: Institute for Veterans & Military Families, Syracuse University.



The Impact of Personality Traits and Internal Locus of Control on Entrepreneurial Intention

Katerina Fotova Čiković¹ 💿

Received: June 2, 2022 Accepted: July 4, 2022 Published: December 30, 2022

Keywords:

Personality traits; Internal locus of control; Entrepreneurship; Entrepreneurial intention; Literature review



Abstract: A country's economic growth and development are closely linked to entrepreneurship and innovation, which is why entrepreneurs are recognized as one of the key drivers of the economy. Moreover, job creation, employment and innovation of products and services not only affect the national economy but also the global economy. Having that in mind, it is comprehensible why the ability to understand and predict entrepreneurial intention (EI) turned into a central category that is widely researched in the literature.

The main goal of this PRISMA-guided systematic literature review is to identify, present and qualitatively analyse relevant papers regarding entrepreneurial intention, personality traits and especially the internal locus of control. Moreover, other goals are to highlight the importance and influence of personality traits and the internal locus of control on entrepreneurial intentions and to present the findings regarding the interrelatedness of entrepreneurial intention, internal locus of control and personality traits.

This article uses a systematic review according to the PRISMA statement guidelines with stringent selection criteria to include relevant articles on the impact of personality traits and internal locus of control on entrepreneurship (i.e. entrepreneurial intentions). The two largest scientific databases, i.e. Scopus and Clarivate Web of Science databases have been explored with the key phrases "PERSONALITY TRAITS", "INTERNAL LOCUS OF CON-TROL" and "ENTREPRENEURIAL INTENTION". This PRISMA method led to a total of 9 relevant papers regarding the impact and influence of personality traits and internal locus of control on entrepreneurship (i.e. entrepreneurial intentions).

The scientific contribution of this paper is mirrored in the presentation of these relevant articles, their findings and in raising awareness among academic members, researchers and governments across the globe regarding the traits that impact entrepreneurial intentions, with a special focus on the findings regarding the internal locus of control. This article provides practical implications for national governments regarding their entrepreneurial policies, support system and climate.

1. INTRODUCTION

Entrepreneurship and venture creation have been recognized as key drivers and significant forces of economic and social growth, regional and national competitiveness, social development and job creation through self-employment, innovation and technological advancement in society in both developed and underdeveloped countries (Denanyoh et al., 2015; Voda & Florea, 2019; Munir et al., 2019). It is now generally accepted that entrepreneurship is positively associated with higher economic development (Rajh et al., 2018). However, entrepreneurship is no longer associated with economic growth only but also with social and environmental performance (Fatoki, 2020). More importantly, entrepreneurs are considered to "bring an enormous contribution to a country's economic growth" (Voda & Florea, 2019). Moreover, "the

University North, Trg dr. Žarka Dolinara 1, 48000 Koprivnica, Republic of Croatia



creation of a country's wealth and dynamism depends upon the competitiveness of its firms and this, in turn, relies fundamentally on the capabilities of its entrepreneurs and managers" (Cuervo et al., 2007). Both national governments and major international organizations worldwide, such as The World Bank and United Nations Industrial Development Organization (UNIDO) are starting to focus on improving countries' business and investment environments (e.g. establishing units to promote private sector development in developing countries and providing technical assistance in the formulation of SME and entrepreneurship policy) for increasing self-employment through entrepreneurship, as a part of their poverty reduction, growth and economic development programs (Acs & Virgill, 2009).

Entrepreneurship has a multidimensional nature and it is a subject of interest to many disciplines: economics, sociology and psychology, which all try to give a summary of the debatable question: what stimulates the self-employability of people and why do different people have different occupational choices (Jovanov Aspasieva et al., 2020). According to Eroğlu et al. (2011), "entrepreneurship is mostly about risks, innovation, creative thinking and the entrepreneur is the one who creates and innovates something recognized around perceived opportunities by accepting risks and failures". "Because entrepreneurship is increasingly seen as a way of dealing with global challenges, it becomes relevant to understand how young people might develop into entrepreneurs" (Ozaralli et al., 2016). Entrepreneurship is often defined as a career path characterised by considerable risk and uncertainty, which renders entrepreneurship attractive to individuals with a high propensity for risk-taking (Karimi et al., 2015). The entrepreneurial function, which discovers and exploits opportunities, leads to the creation of additional economic activities and the generation of new values and benefits for society (Reynolds, 2005). Entrepreneurship as a concept is a subject of great interest in the practice and academic literature. This is since entrepreneurship as an activity creates value for academics, business people and the government in each country around the world (Cuervo, et al. 2007). Therefore, considering the positive impact that entrepreneurship exercises on economic and overall societal development by integrating people, introducing innovations, and creating competition, it is clear why interest in this topic has been increasing rapidly.

Entrepreneurship is an essential element for social progress as it manifests its fundamental importance in different ways: a) by identifying, assessing and exploiting business opportunities; b) by creating new firms and/or renewing existing ones by making them more dynamic; and c) by driving the economy forward – through innovation, competence, job creation- and by generally improving the wellbeing of society (Cuervo et al., 2007). The fascination with entrepreneurs is not brand new and literature dating to the 18th century explores what drives entrepreneurs and whether their traits matter for the outcomes of their ventures (Kerr et al., 2017). Therefore, it should come as no surprise that entrepreneurship research appears to be "one of the most widely cited subjects in the management discipline", mostly due to its practical implications (Paul et al., 2017).

The main objective of this paper is to explore the two most relevant global scientific databases (the Scopus and Clarivate Web of Science) and to identify relevant papers regarding entrepreneurship, entrepreneurial intention, personality traits and internal locus of control. For this reason, a PRISMA-guided systematic literature review has been conducted. For more on the methodology see Liberati et al. (2009). Other objectives of this paper are to highlight the importance and impact of personality traits and the internal locus of control on entrepreneurial intentions and to present and elaborate on the findings regarding the interrelatedness of entrepreneurial intention, internal locus of control and personality traits. This paper is organized as follows. The next section provides a brief theoretical background for understanding entrepreneurial intention (EI), Personality traits and Internal locus of control. The third section then presents the research approach. Section 4 presents the research results, i.e. the 9 surveyed papers and their findings. The study ends with a discussion and conclusions regarding the study's implications, limitations and directions for future research.

2. THEORETICAL BACKGROUND

2.1. A Theoretical Approach towards Understanding Entrepreneurial Intention

Entrepreneurial intention (EI) is referred to as "the first step in understanding the entrepreneurial process" (Ajzen, 1991; Munir et al., 2017). Entrepreneurial intention refers to the "positive or negative intention of an individual toward creating a new business". Thus, "entrepreneurial intention is the base of entrepreneurial actions" (Çolakoğlu & Gözükara, 2016). Entrepreneurial intention shows the intention of a person to choose to be an entrepreneur for his/her career. People who have entrepreneurial intentions plan to take calculated risks, gather required resources and establish their own ventures. Entrepreneurial intention initiates entrepreneurial actions (Karabulut, 2016). In an entrepreneurial context, the entrepreneurial intention is "a significant construct that determines the process of creating ventures" (Çolakoğlu & Gözükara, 2016).

A large number of academic members and researchers have covered EI (entrepreneurial intention) (Lüthje & Franke, 2003; Fini et al., 2012; Nabi & Linan, 2013; Karabulut, 2016; Kerr et al., 2017; Rajh et al, 2018; Šubić et al., 2019). It can be defined as one's willingness to undertake an entrepreneurial activity or the intention to be self-employed and start one's own business. Strong entrepreneurial intention is likely to result in an attempt to start a new business (Ozaralli et al., 2016). According to Voda & Florea, (2019), "intentionality represents a state of mind that can turn actions into actual behaviour". They also claim that EI is influenced by three global concepts as follows: "attitude (how attractive the target behaviour is for the individual), perceived social norms (social pressure in favour or against target behaviour) and perceived control over behaviour (individual perception and the ability to adopt the respective behaviour)".

Theory of Planned Behaviour (TPB) has also shown to be very useful in the examination of the entrepreneurial intent of people. When it comes to TPB, the intention of an individual to perform certain behaviour is considered as its central factor, i.e. the stronger the intention is, the more likely would the performance be (Ajzen, 1988). The intention of the (potential) entrepreneur as a factor, grasps all the motivational factors that stimulate people to behave in a certain way, and their willingness to try to put an effort into performing the behaviour. Papers researching TPB and entrepreneurial intent show a direct link between the personal attitude and perceived behavioural control with behavioural (entrepreneurial) intention, which can be used in predicting behavioural (entrepreneurial) achievement (Rajh, et al. 2018; Jovanov Apasieva, et al., 2020).

2.2. A Theoretical Approach towards Understanding Personality traits

The personality traits approach is one of the classical and early approaches to entrepreneurship (Pupavac, 2015). At the same time, it is one of the more controversial areas of research (Rauch & Frese, 2007). In the 1980s, the literature led some researchers to criticize the trait approach and to draw conclusions that there was no correlation between personality and entrepreneurship. Namely, according to Gartner (1989), the trait approach has been unfruitful and the attempt to answer the

8th International Scientific Conference ERAZ 2022 Selected Papers

question "Who is an entrepreneur?" (which focuses on the traits and personality characteristics of entrepreneurs) will neither lead to a definition of the entrepreneur nor help us understand the phenomenon of entrepreneurship. There were other scholars in the 2000s (Utsch & Rauch, 2000) who believe research on entrepreneurial personality and entrepreneurial success has three major weaknesses: first, most studies investigate global traits that ignore the specific requirements of the entrepreneurial task. Second, studies of the direct relationship between personality and entrepreneurial success ignore the role of entrepreneurial behaviour. Whenever one investigates the personality of an entrepreneur one must also study the actions of the entrepreneur. Third, studies of entrepreneurship often use inadequate statistical methods or fail to exhaust the advantages of modern statistical methods. Furthermore, Utsch & Rauch (2000) argue that the problem with the trait approach in entrepreneurship research is that most investigated traits are not linked to the entrepreneurial situation or task. Personality characteristics are useful in explaining the generation of behaviour when the situation is considered. This means that the power of personality characteristics to predict a certain behaviour depends on the fit between these personality characteristics and the environment in which the behaviour is shown. Therefore, entrepreneurship research should use personality characteristics, which have face validity for the specific entrepreneurial task and work situation. Notwithstanding, a significant number of scholars and researchers have stood up for the traits approach and argued it cannot be dismissed and undervalued and that it still provides a number of promising issues for exploration and insight. Over time, the tide changed and there is now a revival of personality research in performance and leadership research and many other areas of industrial and organizational psychology too. There is now the consensus that there is ample evidence for the validity of certain personality variables in entrepreneurship research (Rauch & Frese, 2007). Consequently, the better recognition of heterogeneity among entrepreneurs and powerful new data sources suggest it might be fruitful to re-examine some of these areas again, some 20-30 years later. After all, the focus for many is now on describing how personality may influence the creation of new organizations, addressing at least some of Gartner's concerns (Kerr et al., 2017).

In the theoretical background, personal history, social context, attitudes toward entrepreneurship, planned behaviour and personality traits are pointed out as factors that have an impact on the propensity to engage in entrepreneurship, and these factors are categorized as internal (personality) and external (contextual or environment) (Yıldırım et al., 2016).

Personality traits are the constructs describing behavioural patterns in individuals' lives (Karabulut, 2016). They have also been extensively researched by previous researchers (Karabulut, 2016; Rajh et al., 2018; Jovanov Apasieva et al., 2020; Kerr et al., 2017; Utsch & Rauch, 2000; Karimi et al., 2015; Lüthje & Franke, 2003).

Rauch & Frese (2007) have distinguished two sets of personality traits in their model: broad personality traits (extraversion, emotional stability, openness to experience, agreeableness, conscientiousness) and specific personality traits (need for achievement, risk-taking, innovative-ness, autonomy, locus of control and self-efficacy).

While personality theory remains rife with its own set of contentions, researchers have primarily gravitated over the last few decades to the Big-5 factor personality model. Several additional traits have been fused into the Big-5 for entrepreneurial work, including self-efficacy, innovativeness, locus of control, and risk attitudes (Kerr et al., 2017). Locus of Control and Risk-Taking Propensity represent personality variables and might allow exploring how personality differences affect entrepreneurial intention (Rajh et al., 2017).

2.3. A Theoretical Approach towards Understanding Internal locus of control

An important trait in entrepreneurship literature is locus of control (LOC). LOC is a personality characteristic that shows the level of control feeling (Karabulut, 2016) and is considered to be a culturally dependent trait (Kerr et al., 2017). Locus of control is conceptualized in two forms, internal and external, former is concerned with the self-efficacy of an individual to influence the outcomes and later talks about the influence of external determinants of the outcomes (Farrukh et al., 2018). A person with an internal LOC "conceptualizes that their own decisions control their lives". In contrast, those with an external LOC believe "the true controlling factors are chance, fate, or environmental features that they cannot influence and thus externally controlled people may be more passive" (Karabulut, 2016). If one believes that one is not able to control business outcomes, one has no reason to actively change one's environment (Rauch & Frese, 2007). Persons with internal LOC believe that they can influence outcomes through their own ability, effort, or skills, rather than external forces controlling these outcomes.

Locus of Control (LoC) is represented as the extent to which a person believes he/she has control over his/her life, i.e. "the level of feeling of being in control" and is considered "a relevant trait in the literature of entrepreneurship" (Karabulut, 2016). The locus of control is a personality variable that reveals "how one's personality can influence entrepreneurial intention" (Rajh et al., 2017). There is internal locus of control and external locus of control. The internal locus of control shows the extent to which a person believes has control over their own life, whereas the external locus of control reveals the extent to which a person believes they are affected by external variables such as destiny, luck and other individuals (Karabulut, 2016). Moreover, LoC is considered one of the "core personality traits in entrepreneurial activities" (Çolakoğlu & Gözükara, 2016). According to Karabulut (2016), "internal locus of control, need for achievement, risk tolerance, and entrepreneurial alertness are dimensions of personality traits which lead a person to develop the entrepreneurial intention".

A person with an internal LOC conceptualizes that their own decisions control their lives. In contrast, those with an external LOC believe the true controlling factors are chance, fate, or environmental features that they cannot influence and thus externally controlled people may be more passive (Rauch & Frese, 2007). Persons with internal LOC believe that they can influence outcomes through their own ability, effort, or skills, rather than external forces controlling these outcomes.

Personality reflects the factors that affect the individual's emotions, thoughts and behaviours, and it reflects both the traits of an individual and the traits of the society. The effect that personality has on entrepreneurship intention has been a widely debated topic recently (Irengun & Arikboga, 2015). According to Lumpkin (1985), locus of control (internal control and external control, i.e. chance) alongside risk-taking propensity according to Luthje & Franke (2003), are considered to be key personality traits that greatly influence entrepreneurial intent of a person. Due to the limited research on internal locus of control and its impact on entrepreneurial intent, this review focuses solely on studies regarding internal locus of control.

3. RESEARCH APPROACH

This in-depth systematic literature review has surveyed both Scopus and Clarivate Web of Science (WoS) scientific databases without imposing any exclusion criteria regarding the period of publication. Therefore, all of the published work has been included in the survey. The keywords "personality traits", "internal locus of control" and "entrepreneurial intention" have been applied to identify the relevant studies that focus on personality traits that impact the entrepreneurial intention, with a special focus on the internal locus of control. The survey was focused on peer-reviewed papers in journals in the English language with no exclusion criteria regarding the year of publication. This approach resulted in 6 hits in Scopus and 10 hits on WoS, which eventually (after the exclusion of 3 duplicate papers, and after the abstract screening), led to a total of nine studies remaining for the qualitative analysis (in the inclusion phase).

The complete survey process with the use of the PRISMA-guided systematic literature review is shown in Figure 1, and a more detailed analysis of these studies and their findings is presented in Section 4.



Figure 1. The selection process using the PRISMA method Source: Author, based on Liberati et al., (2009)

The systematic literature review as a methodology provides a summary of past findings in a research area, and it "minimizes bias and maximizes a review's contribution to science". The PRISMA guidelines were developed in 2009 and consist of a "four-phase flow diagram that describes the identification, screening, eligibility and inclusion criteria of the reports that fall under the scope of a review" (Selçuk, 2019). According to Liberati et al. (2009), a PRISMA-guided systematic literature review represents an "essential tool for summarizing evidence accurately and reliably".

4. **RESEARCH RESULTS**

By researching the world literature with the implementation of the PRISMA-guided systematic literature review, it is obvious that too few scientists are dealing with the research field of entrepreneurial intention and personality with a focus on internal locus of control. Considering that the framework of entrepreneurial behaviour and economic development provides a tool for analysis of the long-term economic performance of national economies, and particularly of the rise and decline of nations (Thurik et al., 2002), this study reveals there is a literature gap waiting to be addressed by scholars.

This study identifies all the relevant previous work regarding entrepreneurial intention, personality traits and internal locus of control and it provides a presentation of the studies and their findings. A short overview of the surveyed papers is presented in Table 1, and a thorough presentation and analysis of the findings are presented thereafter.

| Author/s and year of publication | Title of the paper | Data sample | Country |
|----------------------------------|---|--|-----------------------|
| Saeed et al. (2013) | Who is the most potential entrepreneur? A case of Pakistan | 100 business graduates from 5 public universities | Pakistan |
| Karabulut (2016) | Personality Traits on Entrepreneurial Intention | 480 students | Turkey |
| Çolakoğlu & Gözükara (2016) | A comparison study on personality traits based on the attitudes of university students toward entrepreneurship | 226 vocational school students | Turkey |
| Molino et al. (2018) | Personality and social support as determinants of entrepreneurial intention. Gender differences in Italy | 658 Italian participants | Italy |
| Munir et al. (2019) | Personality traits and theory of planned behavior comparison of entrepreneurial intentions between an emerging economy and a developing country | 1,016 students | China and Pakistan |
| Voda & Florea (2019) | Impact of Personality Traits and Entrepreneurship Education on Entrepreneurial Intentions of Business and Engineering Students | 270 students | Romania |
| Mahmood et al. (2020) | Attitude towards entrepreneurship: a study among Asnaf Millennials in Malaysia | 310 Asnaf Millennials | Malaysia |
| Fatoki (2020) | Determinants of Sustainability-Oriented Entrepreneurial Intentions of University Students | 408 participants | South Africa |
| Presenza et al. (2020) | Start-up entrepreneurs' personality traits. An exploratory analysis of the Italian tourism industry | 112 members of the Italian Tourism Start-up Association | Italy |

Table 1. Surveyed papers

Source: Author's processing

8th International Scientific Conference ERAZ 2022 Selected Papers

Saeed et al. (2013) found that "among all big-five personality traits, only extraversion and openness to experience are significantly related to entrepreneurial intentions among business graduates in Pakistan". Moreover, risk taking is found to be "significantly related to entrepreneurial intentions". However, both internal and external locus of control do not impact entrepreneurial intentions.

Karabulut (2016) explored the impact of personality traits on entrepreneurial intention (EI), by using a sample of 480 graduate students in Turkey. The used methodologies are factor analyses and multiple regression analysis. The results confirm that "personality traits have a positive impact on the entrepreneurial intention". His study is one of the most popular studies in the research field of personality traits.

Çolakoğlu & Gözükara (2016) conducted research at a foundation university in Istanbul, Turkey with the goal "to compare personality traits based on the attitudes of university students toward entrepreneurship". Their findings reveal that "students with entrepreneurial intention are more innovative, have a higher need for achievement and greater internal locus of control than those who do not have such intention". Their results confirm that "entrepreneurial intention is significantly related to certain personality traits". They furthermore suggest promoting entrepreneurship and including entrepreneurship education in order to "create an economically strong country".

Molino et al. (2018) have focused on exploring the impact of personality traits and social support on entrepreneurial intention, with a special emphasis on gender differences in Italy. In their study, they gathered information from a sample of 658 Italian participants, of which 319 male and 339 female participants and developed a multi-group structural equation model. Their findings reveal that "self-efficacy totally mediated the relationship between internal LoC, self-regulation and EI". Moreover, they found that contextual characteristics were the key mediational function of self-efficacy and found gender differences in the entrepreneur role.

Munir et al. (2019) have explored the differences between the entrepreneurial intentions between an emerging economy and a developing country, i.e. China and Pakistan. Their sample consists of 1.016 university students in their final year of studies. The used methodologies are structural equation modelling and the partial least square (PLS) method and present PLS path modelling, mediation analysis and multigroup analysis. Their findings show that the impact of TPB (theory of planned behaviour) "was positive and significant in both countries". However, interestingly enough, the "TPB demonstrated more explaining power in China's student sample", meaning they found a stronger influence of personality traits among Chinese students. Their study is the first of a kind due to the comparison between two economies and it brings new insights for academic members as well as policymakers. Moreover, the integrated model they present is novel and original.

Voda & Florea (2019) have analysed the entrepreneurial intentions, locus of control, need for achievement and entrepreneurial education among 270 students from two universities in Romania in the period from March 2017 to October 2017. In their study, they have implemented multivariate logistic regression estimates, frequency analysis, internal consistency reliability of the constructs and moderating effects, and their findings indicate that "locus of control, need for achievement and entrepreneurial education proved to be important determinants for venture creation among young students, both independently and under the action of control variables".

Moreover, they found that the respondents' gender impacts "one's intention of opening a business in the future, with males being more inclined to become entrepreneurs than females". Interestingly, not only gender but also age and faculty type impact the distribution of the answers for entrepreneurial intention. This study provides new and valuable insights regarding entrepreneurial intentions, locus of control, need for achievement and entrepreneurial education among students from a post-transition country.

Mahmood et al. (2020) explored the effects of selected entrepreneurial traits on the attitude of 310 randomly selected Asnaf Millennials from Kelantan, Malaysia towards entrepreneurship. They used structured interviews and found a "positive and statistically significant effect of innovativeness, internal locus of control, need for achievement and proactive personality on the attitude of Asnaf Millennials on entrepreneurship". Their work is novel since emphasizes the "non-cognitive skills for individual characteristics regarding entrepreneurship for the benefit of development practitioners and policymakers".

Fatoki (2020) has developed a unique multi-dimensional model that incorporates personality traits as antecedents of SOEI (sustainability-oriented entrepreneurial intentions), which is based on Ajzen's TPB (Theory of Planned Behaviour). This model has been later on examined by Partial Least Square Structural Equation modelling (The PLS SEM) and tested on a sample of 408 participants (220 females and 189 males) between the age of 20 to 30 years old. The results imply that "attitude towards sustainable entrepreneurship has a significant positive relationship with SOEI". Moreover, a significant positive relationship between perceived behavioural control and SOEI has been found and results show the "internal locus of control has a significant positive relationship with SOEI".

Presenza et al. (2020) examine the personality traits of "start-up entrepreneurs within the Italian tourism industry". Their goal is to "investigate how the specific personality traits of narcissism, locus of control, and the Big Five (i.e. extraversion, agreeableness, conscientiousness, neuroticism, openness to experience) influence the tourism start-up entrepreneur's behaviour". Their study has a significant scientific and practical contributes due to the two main results that arose from the empirical analysis, as follows: "tourism entrepreneurs seem to be particularly narcissistic, extroverted and friendly (extraversion), disciplined (agreeableness), self-assured, efficient and organized (conscientiousness), unconventional (openness to experience), and proactive and vigorous (internal locus of control); some personality traits are found to influence the efficiency and innovative capacity of entrepreneurs which ultimately lead to the success of start-ups".

5. DISCUSSION AND CONCLUSION

The main goal of this paper was to survey, identify and present the relevant studies regarding entrepreneurial intention, personality traits and especially the internal locus of control. Moreover, other goals were to highlight the importance and influence of personality traits and the internal locus of control on entrepreneurial intentions and to present the findings regarding the interrelatedness of entrepreneurial intention, internal locus of control and personality traits. For this reason, the PRISMA-compliant systematic literature review of publications regarding entrepreneurship, personality traits and internal locus of control has been the main methodology for this research.

The research approach (shown in Figure 1) has been initiated by the exploration of the two most reputable and renowned scientific databases nowadays (i.e. the Scopus and the Web of Science

databases), with the use of three key phrases as follows: "entrepreneurial intention", "personality traits" and "internal locus of control". This resulted in a total of 9 published manuscripts (after the exclusion of the non-relevant papers). The research approach and the PRISMA method have led the authors to exclusion of papers that were not in English (1 paper), exclusion of papers after the screening and abstract revision (6 papers), which ultimately led to a total of 9 relevant papers for further qualitative analysis.

Eight out of nine surveyed studies use samples from one national economy. Only the study of Munir et al. (2019) is a cross-country study, employing the same research to samples from both China and Pakistan. Two of these studies analyse the entrepreneurial intention of Turkish students (Karabulut, 2016; and Çolakoğlu & Gözükara, 2016), two are regarding Italian students (Molino et al., 2018; and Presenza et al., 2020), while two analyse the entrepreneurial intention in Pakistan (Saeed et al., 2013 and Munir et al., 2019). It is evident that these studies are mostly conducted in developing and emerging countries in Asia, Africa and Europe, and the countries from Northern Europe and the Americas are less presented. This could be due to the fact that entrepreneurship education has already been integrated into the system thus providing a greater percentage of successful entrepreneurs that contribute to the economy.

The largest sample of 1016 students is used in the study of Munir et al. (2019), whereas the smallest sample has been applied to the study of Saeed et al. (2013), i.e. 100 students.

The surveyed papers give new insights into the entrepreneurial intent, i.e. SOEI (sustainability-oriented entrepreneurial intention) in South Africa, EI of start-up entrepreneurs in Italy, Asnaf Millennials in Malaysia, etc. Most of these studies confirm that personality traits and internal locus of control have a positive impact on entrepreneurial intention. These findings offer profound implications for governments, policymakers and entrepreneurship educators.

A major limitation of this study (as for any systematic literature review study) is the exploration and survey of only two scientific databases (in this case, the Scopus and the Clarivate Web of Science). Therefore, not all published relevant papers and studies may have been presented in this study due to their indexation in other scholarly databases. However, this study, through the presented findings, largely contributes to scholarly literature and the policymakers, as well as the interested public.

In future work, an empirical study is to be conducted to explore the impact of personality traits, internal locus of control and risk propensity on entrepreneurial intent in students in public universities in Croatia.

REFERENCES

- Acs, Z., Virgill, N. (2009). Entrepreneurship in developing countries, *Jena Economic Research Papers*, No. 2009, 023, Friedrich Schiller University Jena and Max Planck Institute of Economics, Jena.
- Ajzen, I. (1988). Attitudes, personality, and behavior. Chicago: DorseyPress.
- Ajzen, I. (1991). The theory of planned behavior, *Organizational Behavior and Human Decision Processes*, *50*(2), pp.179-211, DOI: 10.1016/0749-5978(91)90020-T.
- Çolakoğlu, N., Gözükara, I. (2016). A comparison study on personality traits based on the attitudes of university students toward entrepreneurship, *Procedia - Social and Behavioral Sciences 229*, pp. 133 – 140. DOI:10.1016/j.sbspro.2016.07.122
- Cuervo, Á., Ribeiro, D., Roig, S. (2007). Entrepreneurship: Concepts, Theory and Perspective. Introduction. In: Cuervo, Á., Ribeiro, D., Roig, S. (eds) Entrepreneurship. Springer, Berlin, Heidelberg. DOI: 10.1007/978-3-540-48543-8_1.
- Denanyoh, R., Adjei, K., Nyemekye, G.E., (2015). Factors That Impact on Entrepreneurial Intention of Tertiary Students in Ghana, *International Journal of Business and Social Research*, 5(3), pp.19-29.
- Eroğlu, O. & Picak, M. (2011). Entrepreneurship, national culture and Turkey, *International Journal of Business and Science 2*, pp. 146-151.
- Farrukh, M., Alzubi, Y., Shahzad, I.A., Waheed, A. and Kanwal, N. (2018). Entrepreneurial intentions: The role of personality traits in perspective of theory of planned behaviour, *Asia Pacific Journal of Innovation and Entrepreneurship*, 12(3), pp. 399-414. DOI: 10.1108/ APJIE-01-2018-0004.
- Fatoki, O. (2020). Determinants of Sustainability-Oriented Entrepreneurial Intentions of University Students. *Southern African Business Review*, 24, DOI: 10.25159/1998-8125/7795.32.
- Fini R., Grimaldi R., Marzocchi G.L., Sobrero M., (2012). The determinants of corporate entrepreneurial intention within small and newly established firms, *Entrepreneurship Theory* and Practice, 36, pp. 387-414. DOI: 10.1111/j.1540-6520.2010.00411.x
- Gartner, W. (1989). Who Is an Entrepreneur? Is the Wrong Question. *Entrepreneurship Theory and Practice. 13.* DOI: 10.1177/104225878801200401.
- Irengun, O. & Arikboga, S. (2015). The effect of personality traits on social entrepreneurship intentions: a field research. *Social and Behavioral Sciences 195*, pp. 1186-1195. DOI: 10.1016/j.sbspro.2015.06.172
- Jovanov Apasieva, T., Cabuleva, K., Mitreva, M. (2020). Theory of planned behavior: personal attitude and perceived behavioral control as key determinants in creation of entrepreneurial societies and social inclusion of young people. *Balkan Social Science Review, 15,* pp. 275-297. DOI: https://doi.org/10.46763/BSSR20150276ja
- Karabulut, A. T. (2016). Personality Traits on Entrepreneurial Intention, *Procedia Social and Behavioral Sciences 22*, pp. 12 21. DOI: 10.1016/j.sbspro.2016.07.109
- Karimi, S., Biemans, H., Mahdei, K.N., Lans, T., Chizari, M. & Mulder, M. (2015). Testing the Relationship between Personality Characteristics, Contextual Factors and Entrepreneurial Intentions in a Developing Country. *International Journal of Psychology*, 52. DOI: 10.1002/ijop.12209.
- Kerr, S.P., Kerr, W.R., Xu, T. (2017). Personality Traits of Entrepreneurs: A Review of Recent Literature. Harvard Business School. *Working Paper 18-047*.
- Liberati, A., Altman, D., Tetzlaff, J., Mulrow, C., Gøtzsche, P, Ioannidis, J., Clarke, M., Devereaux, P.J. Kleijnen, J. & Moher, D. (2009). The PRISMA Statement for Reporting Systematic Reviews and Meta-Analyses of Studies That Evaluate Health Care Interventions:

Explanation and Elaboration. *Journal of clinical epidemiology.* 62. e1-34. DOI: 10.1016/j. jclinepi.2009.06.006.

- Lumpkin, J. R. (1985). Validity of a brief locus of control scale for survey research. *Psychological Reports*, 57(2), pp. 655–659.
- Lüthje, C., & Franke, N. (2003). The 'making' of an entrepreneur: Testing a model of entrepreneurial intent among engineering students at MIT. *R&D Management*, *33*(2), pp. 135–147. DOI: 10.1111/1467-9310.00288
- Mahmood, T.M.A.T., Al Mamun, A. & Ibrahim, M.D. (2020). Attitude towards entrepreneurship: a study among Asnaf Millennials in Malaysia. *Asia Pacific Journal of Innovation* and Entrepreneurship 14(1), pp. 2-14. DOI 10.1108/APJIE-06-2019-0044.
- Molino, M., Dolce, V., Cortese, CG., Ghislieri, C., (2018). Personality and social support as determinants of entrepreneurial intention. Gender differences in Italy. *PLoS ONE 13*(6): e0199924. DOI: 10.1371/journal.pone.0199924.
- Munir, H., Jianfeng, C. & Ramzan, S. (2019). Personality traits and theory of planned behavior comparison of entrepreneurial intentions between an emerging economy and a developing country. *International Journal of Entrepreneurial Behavior & Research*, 25 (3), pp. 554-580. DOI: 10.1108/IJEBR-05-2018-0336.
- Nabi, G., & Linan, F. (2013). Considering business start-up in recession time: The role of risk perception and economic context in shaping the entrepreneurial intent. *International Journal of Entrepreneurial Behaviour & Research, 19*(6), pp. 633-655. DOI: 10.1108/ IJEBR-10-2012-0107
- Ozaralli, N., Rivenburgh, N.K. (2016). Entrepreneurial intention: antecedents to entrepreneurial behavior in the U.S.A. and Turkey. *Journal of Global Entrepreneurship Research*, 6(3), pp. 1-32, DOI: 10.1186/s40497-016-0047-x.
- Paul, J., Hermel, P., Srivatava, A. (2017). Entrepreneurial intentions theory and evidence from Asia, America, and Europe, *Journal of International Entrepreneurship*, 15(3), pp. 324-351. DOI: 10.1007/s10843-017-0208-1
- Presenza, A., Abbate, T., Meleddu, M. & Sheehan, L. (2020). Start-up entrepreneurs' personality traits. An exploratory analysis of the Italian tourism industry, *Current Issues in Tourism*, 23(17), pp. 2146-2164, DOI: 10.1080/13683500.2019.1677572.
- Pupavac, D. (2015). Extroverted Personality As An Advantage In Entrepreneurship. *Obrazovanje za poduzetništvo - E4E*, 5 (2), pp. 133-143.
- Rajh, E., Budak, J., Ateljević, J., Davčev, Lj., Jovanov, T. & Ognjenović, K. (2017). Entrepreneurial intentions in selected South-East European countries. In: *Economic development* and entrepreneurship in transition economies: is free trade working for transitional and developing economies?: Conference proceedings. Faculty of Economics, Banja Luka, pp. 592-610.
- Rauch, A., & Frese, M. (2007). Born to Be an Entrepreneur? Revisiting the Personality Approach to Entrepreneurship. In J. R. Baum, M. Frese, & R. A. Baron (Eds.), The organizational frontiers. The psychology of entrepreneurship (pp. 41–65). Lawrence Erlbaum Associates Publishers.
- Reynolds, P. D. (2005). Understanding business creation: Serendipity and scope in two decades of business creation studies. *Small Business Economics*, 24, pp. 359-364.
- Saeed, R., Nayyab, H.H., Rashied, H., Lodhi, R.N., Musawar, S. & Iqbal, A. (2013). Who is the most potential entrepreneur? A case of Pakistan. *Middle East Journal of Scientific Research. 17.* pp. 1307-1315. DOI: 10.5829/idosi.mejsr.2013.17.09.12296.
- Selçuk A. A. (2019). A Guide for Systematic Reviews: PRISMA. Turkish archives of otorhinolaryngology, 57(1), pp. 57–58. DOI: 10.5152/tao.2019.4058.

- Šubić, R., Načinović Braje, I. & Žagi, K. (2019). Family background and financial literacy as a prerequisite for entrepreneurial intention of university students. In: *Proceedings of FEB Zagreb 10th International Odyssey Conference on Economics and Business* DOI: 10.22598/odyssey.
- Thurik, A. R., L. M. Uhlaner and S. Wennekers, (2002). Entrepreneurship and Its Conditions: A Macro Perspective, *International Journal of Entrepreneurship Education 1*(1), pp. 25–64.
- Utsch, A. & Rauch, A. (2000). Innovativeness and initiative as mediators between achievement orientation and venture performance. *European Journal of Work and Organizational Psychology*, *9*(1), pp. 45–62. DOI: 10.1080/135943200398058.
- Voda, A.I., Florea, N. (2019). Impact of Personality Traits and Entrepreneurship Education on Entrepreneurial Intentions of Business and Engineering Students. *Sustainability*, 11, 1192; DOI: 10.3390/su11041192.
- Yıldırım, N., Çakır, Ö., Bige Aşkun, O. (2016). Ready to Dare? A Case Study on the Entrepreneurial Intentions of Business and Engineering Students in Turkey, *Procedia - Social and Behavioral Sciences 229*, pp. 277 – 288. DOI: 10.1016/j.sbspro.2016.07.138



Barriers to Effective Interpersonal Communication in Retail

Petra Kováčová¹ D Ondřej Drahotský² D

Communication partner;

Keywords:

Employees;

Organization

Received: May 30, 2022 Accepted: July 8, 2022 Published: December 30, 2022

Abstract: Communication is an inseparable part of our daily lives. Interpersonal communication appears between two and more people and consists not only of verbal communication by using words and consciously talking to others but also consciously or unconsciously by using non-verbal communication such as gestures, body movement and posture, facial expressions, haptics (touch), eye contact, proxemics (physical distance), paralanguage, etc. In verbal and non-verbal communication enter some issues that can cause misunderstandings in the communication process. People communicate everywhere and anytime, therefore it is natural that some difficulties and problems occur in interpersonal communication and are encountered in the workplace. There are different types of communication in the workplace, but this paper focuses mainly on interpersonal communication between managers and their team members. The way how they communicate influences the output of the entire communication process and has direct or side effects on performance, work environment, company goals and similar. The main goal of this paper is the identification of barriers to effective interpersonal communication in a selected branch of an international retail organization in Germany. Based on the research, it is possible to identify the barriers to the effective communication process between managers and their team members. Of a total of 187 employees working at this branch, 48 employees participated in this survey. This paper provides results from conducted research where some positive aspects and barriers were explored as well. Among positive results are that the employees do not afraid to address the company's management directly, employees are not interrupted while speaking, do not get the information they do not understand, the managers do not use the terms their employees do not understand, employees are not often overwhelmed with the information, not facing a misunderstanding from manager's side. On the other hand, the organization may support more communication between departments and provide more information to employees to feel well informed. This paper correlates with other research conducted in this field by different researchers in different countries and adds the missing barriers or confirmed already often occurring barriers to effective interpersonal communication.

1. INTRODUCTION

further permission.

Creative Commons Non EV NC Commercial CC BY-NC: This

article is distributed under the terms of

the Creative Commons Attribution-Non-

Commercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which

permits non-commercial use, reproduc-

tion and distribution of the work without

Nowadays, we often come across various publications in conjunction with communication. Communication is an essential element of society that we meet every day. We constantly communicate, whether verbal or nonverbal. It is not possible not to communicate. Whether to talk about common problems, joys, work, or talk to friends, colleagues or family, people are constantly communicating. It means transmitting and receiving signals. Individuals are also affected by the environment in which they communicate. Not only in the workplace, but also in every situation, they should try to create suitable conditions so that communicators in the communication

² Alexander Dubcek University of Trencin, Faculty of Social-Economics Relationship, Študentská 3, 911 50 Trenčín, Slovakia



¹ Alexander Dubcek University of Trencin, Faculty of Social-Economics Relationship, Študentská 3, 911 50 Trenčín, Slovakia

8th International Scientific Conference ERAZ 2022 Selected Papers

process feel relaxed and comfortable because this can affect the course of the entire communication. The basis for achieving the company's goals is a functioning communication and communication network. In general, it can be considered a critical area of management skills. The manager is responsible for effective communication in the organization. Communication is a complex process, so it needs to be understood in context. It is a unique process that reflects the life of the individual, the group and the entire organization. To complete the communication process, communication partners need to understand the importance of feedback and keep on delivering the completed message (Zahid, Qinghe and Sohail, 2021). Managers are people whose job success depends on their communication abilities. They need to learn how to maximize the effectiveness of their communication skills and how to achieve a meaningful work effect. In communication, people often encounter certain problems, and barriers. In order to be able to prevent them, they should know them. These can be barriers on the part of the sender, recipient, or environmental factors. Deficiencies and ineffective communication lead to problems in interpersonal relationships, lower work efficiency and demotivation. For the failure of a communication process, there can be a lot of reasons, therefore, communication partners need to communicate clearly, take the feedback from another person, and make sure that the message is communicated in proper way. Every activity of a communication partner involves direct or indirect interaction (Zahid, Qinghe and Sohail, 2021). Managers communicate daily with their team, whether about routine matters (work, private), tasks, goals, problems, problem-solving. Through communication, managers lead their team, evaluate employees, motivate them, guide them, communicate between departments, with the company board, personnel department and customers, clients. In every company can occur different barriers in communication which can cause immediate negative effects on the communication process and communication partners (Yusof and Rahmat, 2020). Lately, there were conducted only a few studies focusing on barriers to effective communication therefore it is needed to bring up this topic and go deeper to research what the main barriers are and how to avoid them.

The structure of this paper is the following: (1) Literature review – introduces the main topic of this paper and the latest results of research in this area. (2) Methodology – describes the data used in research and the whole methodology. (3) Discussion – presents the results of the survey. (4) Conclusion – shows concluding remarks about employee development and the limits of the research.

2. LITERATURE REVIEW

Communication is an active process that is occurring all around us. About 70 % of our time, we spend communicating, by sharing and receiving information through various meanings (Ansari, 2021). The best way to transfer information, attitude and emotions is through face-to-face communication. This form of communication provides immediate feedback between communication partners and non-verbal communication supports the message. It is also a good way to develop positive relations between people. Both verbal and non-verbal communication is important for meaningful interaction (Venter, 2019). If there is a conflict between verbal and non-verbal communication, verbal expression is often overlooked in favor of the non-verbal understanding of the message. Non-verbal communication builds comfort by avoiding ambiguity and creating a common foundation for understanding (Lamichhane, 2016). There can occur a lack of meaningful conversation between communication partners due to limited non-verbal cues, where they seek to understand the more emotional content of a conversation (Venter, 2019). Showing empathy and understanding the perspective of team members does not cause changes in goals or profession-al expectations. If managers have high emotional intelligence, they can still keep team members

responsible for results and positively address inefficiencies. The perception of the manager influences the work environment, team togetherness and cohesiveness. Observing individuals and listening to what others say can reveal more about their personalities, how they are and how they want to be treated (Nguyen et al., 2019). An inaccurate understanding of the meaning of the message can cause a communication barrier between communication partners (Demchenko, Khoroshevskaya and Krukov, 2021). Among communication barriers can be found the loss of the original meaning of the message, distorted information, increase in the level of topics to be communicated, attitudes and behaviors in the communication process, giving and receiving feedback, managers expressing their superiority, missing effective listening skills, lack of interest, information overload, low level of trust between manager and employee, lack of understanding and concern and similar. These barriers were also confirmed in the research conducted by Vasilev and Stefanova (2021). A better look can help to overcome these and similar barriers and improve effective communication (Vasilev and Stefanova, 2021). The positive value is seen when the managers invest in improving their interpersonal skills and as a result improve the relationship between manager and team members which also impacts organizational economic growth (Saadi, 2018). There is evidence of the positive influence of effective communication in daily work. Interpersonal communication is seen as effective when there is a reaction or response from the communication partner. The meaning of the message has to be transferred in the communication process; the process is successful if the other side understands the basic idea of a message (Naumovski et al., 2016). Face-to-face conversations use communication strategies using verbal and non-verbal communication to express feelings and attitudes. Quality interpersonal communication requires listening and empathy, the ability to express emotions, lead the meaningful conversation for both communication partners as well as give and receive feedback (Venter, 2019). According to Rana (2013) one of the top reasons the organization fails is poor communication in the workplace which causes employees to tend to be cynical, lose their trust and respect for the leaders and the whole organization. Rani (2016) confirmed that when we communicate, we have a tendency to raise barriers that hinder our ability to communicate. Among the potential communication barriers belong the selection of words (e.g., too technical), grammar, sentence structure, punctuation and spelling can also hinder the understanding of the message, physical form of the message (appearance can affect its readability), personal appearance of the communicator (e.g., the credibility of the oral message, distraction), environmental factors (e.g., noise), the ability of communication partner (e.g., any disabilities), ineffective listening skills or other communication barriers (e.g., lack of interest, lack of knowledge, different cultural perceptions, linguistic differences, biases and similar). The communication partners should try to avoid any possible barriers of ineffective communication, related to communication barriers (Naumovski et al., 2016). Interpersonal communication skills make a person more charismatic which can help to sustain stability in the workplace, moreover, managers can maintain the work environment as positive and productive helping to build trust with their teams and retain them (Ansari, 2021). Also, the function of interpersonal communication is to reduce uncertainty, avoid and overcome conflicts, share knowledge and experience (Lusiawati, 2019).

3. METHODOLOGY

The subject of the survey is the identification of barriers to effective interpersonal communication in the organization. Two research questions were set: 1. What are the main barriers to effective interpersonal communication? 2. How is the communication flow in the organization? This second research question is answered by these questions from the questionnaire: "The organization supports communication between departments." and "The company's management provides all necessary information in relation to employees." Based on these questions and the characteristics of the way of communication in the organization and the identification of barriers occurring in the communication process, may be determined the deficiencies as well as the positives of communication in the workplace. This paper shows how employees and managers communicate and their mutual satisfaction. The survey was conducted before Covid-19 crisis has started in a selected international company in Germany.

The research is based on the quantitative-qualitative method, where the questionnaire consisted of 10 closed questions and 5 open questions focusing on describing what communication problems the employees can encounter in the organization, what they suggest to prevent the problem or how to fix it, how the way of communication in the organization can be improved, whether the organization provides training to increase communication skills, and the last question provides an answer to the overall satisfaction of employees with the way of communication in the organization. The questionnaire was co-created with HR professionals that consisted of questions where respondents answered whether they agree with the statement, disagree, or partially agree/disagree in the first part of the questionnaire, followed by five open-ended questions where respondents had the opportunity to present their views. Similar ways of working with this topic were chosen by other researchers and studies are presented in this paper.

48 employees (32 women and 16 men) participated in the questionnaire survey of a total of 187 employees, which means the return rate of the questionnaire is 25.67%.

4. **RESULTS**

The results of the research are summed up in the following Table 1, where are shown the respondents' responses (in percentage) on barriers to effective communication:

| Barriers to effective communication in the organization | Agree with the statement | Disagree with the statement | Partially agree / partially disagree with the statement |
|--|--------------------------|-----------------------------|---|
| I am not afraid to address the company's management directly. | 78,7 % | 12,6 % | 12,5 % |
| 2. The organization supports communication between departments. | 37,6 % | 31,3 % | 31,3 % |
| 3. The company's management provides all necessary information in relation to employees. | 20,9 % | 37,6 % | 41,7 % |
| 4. I often encounter a misunderstanding of my manager in the communication process. | 10,5 % | 54,3 % | 35,5 % |
| 5. I am often overwhelmed with a lot of information. | 16,7 % | 75,1 % | 8,4 % |
| 6. I don't understand the terms my manager uses. | 4,2 % | 89,7 % | 6,3 % |
| 7. I often encounter my manager's lack of interest in my problem. | 6,3 % | 73 % | 20,9 % |
| 8. I'm getting information I don't understand. | 6,3 % | 85,5 % | 8,4 % |
| 9. When I talk, my manager often interrupts me. | 8,4 % | 75,1 % | 16,7 % |
| 10. When I talk to my co-workers, they often interrupt me. | 12,6 % | 68,8 % | 18,8 % |

Table 1. Barriers to effective communication in the organization

Source: Own research

The results of the survey differ on whether the organization supports communication between departments. Employees tended to believe that the company's management did not provide all the necessary information to them, according to 36.7% of respondents and 41.7% partially agreed and partially disagreed that the company's management provide enough information to employees. On other issues, employees did not see significant problems. For example, when it comes to using terms that employees do not understand, as many as 89.7% of respondents said they understood the technical terms used by their manager. It was similar in terms of receiving information from managers that employees do not understand, where only 6.3% of employees do not understand them and 85.5% of employees do. According to Yusof and Rahmat (2020), if the communication partners are not able to understand each other, the communication barrier can occur and they cannot achieve an agreement. Moreover, some employees can have communication issues because of sending unclear messages for them, which can cause a lack of clarity (Adu-Oppong and Agyin-Birikorang, 2014). Different data have to be summarized or simplified to make a message clearer (Demchenko, Khoroshevskaya and Krukov, 2021). As we can see in Table 1, 73% of respondents do not agree that they would not be interested in their problem with their manager. It is expected of managers to create an environment with easy and effective communication flow (Oladiran and Burghate, 2016). According to one research, most organizations and managers have difficulties with communication strategies that influence employees' productivity and the success of the entire organization (Nelly and Mosley, 2018). The quality of communication exchange depends also on the status differences of employees (employees do not always inform their managers and managers also separate some information for their employees consciously or by using different communication channels) (Demchenko, Khoroshevskaya and Krukov, 2021).

The next part of the questionnaire consisted of three open questions. The first question is: Describe what communication problems you encounter in the organization? Responses to this question are divided into several areas: a) Information flow, hierarchy and feedback (little or no feedback, little direct communication between levels, especially manager vs. co-workers in the organization, the organization is led through managers, it is difficult to provide open criticism, the problem of finding the right person to communicate with, poorer flow of information, lengthy information flow through the organization, departments, team leaders, assistants and employees divide information or a gap in understanding, a large amount of information to pass to all employees are different, e.g. if they are also not at work every day, so it is difficult to find the person you are looking for), b) Time and workload (failure to provide information in advance, what is planned, open deadlines, no precise determination e.g. annual talk, increased time pressure, too many tasks divided among few people, often interrupt the phone during conversations - this one may be also include in area "openness and empathy), c) Openness and empathy (disregard for ideas, worries, fears, problems, dealing with only one's own interests, low ability to empathize, lack of interest), d) Ways of communication (barriers due to poor language skills, sometimes problems communicating with people for whom German is a foreign language, the responsibility of other areas, there are many ways of communication and it is often not clear which is the easiest for employees, insufficient information about the current problem, use of technical terms). Quantitative research conducted by Yusof and Rahmat (2020) for working adults coming from different types of companies found similar results to the presented survey above, people need some time to understand foreign colleagues when communicating and have some problems understanding the foreign accent. Among other results, there can be seen a correlation between the survey presented in this paper and research by Yusof and Rahmat in responses connected to time - not enough time to do work or other personal barriers such as

8th International Scientific Conference ERAZ 2022 Selected Papers

"not allowed to voice out the opinion unless someone asks, cannot question the seniors at the workplace, awareness of the emotional states of colleagues, ability to put into another's shoes". Studies present the importance of effective interpersonal communication in the workplace and the role of managerial communication in managing employees, conflict resolution, job responsibilities and organization's productivity (Oladiran and Burghate, 2019). The information distortions may increase in organizations with multiple hierarchical levels (Demchenko, Khoroshevskaya and Krukov, 2021). Another study supports the face-to-face conversation between managers and their employees as an effective way of communication due to paying attention to the body language and more accurately interpreting the message (Jonsdottir and Fridriksdottir, 2019). Moreover, the pressure of work often blocks effective communication, time pressure and heavy workload can cause passing out insufficient or incomplete information to employees (Oladiran and Burghate, 2019). These examples clearly describe the problems that employees face in the organization.

In the second question, employees were asked what they suggest to prevent the problem or how to fix it. The most common suggestions include sharing information on what is planned for tomorrow / next week, providing feedback (even negative), talking about the problem with all stakeholders (either not to promise), meeting deadlines (talks, interviews), and collecting other opinions from the department. Other responses of respondents include increasing soft as well as professional skills such as increasing empathy, seeing each colleague as part of a team working with the same goal, raising team spirit, increasing motivation, adhering to, creating a central position or delegating an employee who is always located in a certain place, providing information from managers to employees, entrusting a company employee to whom employees can turn when their line manager is not available, follow the agreed goals more together, managers should be less committed and have more time spontaneously for employees, the structure of meetings at the level of managers should be understandable for employees, learning technical terms, more employees (staff planning), division of priorities, improve management structure, create training for each manager on various topics, e-mail access for all employees, regular mutual interviews, language course, etc. To communicate effectively, managers need to know how to handle formal as well as informal communication within the organization (Oladiran and Burghate, 2019). Employees suggested several options to solve various problems. Their implementation could prevent misunderstandings that currently arise in the workplace. Openness, active listening, body language, conflict resolution, assertiveness, negotiation skills, positive attitude, decision making and problem-solving, assertiveness, empathy and teamwork are the competencies needed for effective interpersonal communication (Begiri, 2018).

The third question follows from the previous one, how the way of communication in the organization can be improved. The answers can be used to formulate useful recommendations such as talking at eye level and conducting open conversations more often (multiple meetings), improving time management (managers should devote more time to employees), leading conversations/ meetings also between departments, but also to cooperate and explain the context for better understanding, to ask, request information and connect questions so that everyone can as quickly as possible come to more relevant, more transparent communication between departments, to be open and friendly to each other. The responses to this question are very similar to the previous one, they provide solutions to problems without suggestions for a specific implementation. According to a study conducted by Jonsdottir and Fridriksdottir (2019), in the communication process is important the understand both communication partners, use open questions to obtain certain information and lead the conversation, seek further clarification by asking. Another
survey added - what employees expect from their manager: openness, active listening, problem-solving and teamwork (Ansari, 2021). Having strong communication skills is a base for managers to communicate effectively and be able to make a decision and resolve problems; furthermore, it supports a smooth workflow and establishes a positive influence on the entire organization (Oladiran and Burghate, 2016).

Two questions were asked in the last part of the questionnaire survey. To the first question of whether the organization provides training to increase communication skills, 27.1% of employees say yes, 33.3% of employees say no and 39.6% of employees do not know. Organizing training to increase communication skills is important, at least at the level of managers, so that they can have effective talks, conversations, and interviews with their employees. Disilva and Arun (2017), who conducted a study in the hospitality industry, revealed that the programs created for communication skills development have a positive effect on employee development. What is more, 65,2% of respondents lack the training and development programs and see it among the biggest communication barriers. Therefore, they (89,1%) request an appropriate training program to overcome the communication barriers. In another survey conducted by Ansari (2021), 88,9 % of respondents stated that their organization lacks practical training such as body language. The second question provides an answer to the overall satisfaction of employees with the way of communication in the organization. Overall, exactly 50% of employees are satisfied.

5. CONCLUSION

The paper analyses the barriers to effective communication in a selected organization with the goal to identify them. The results of the research show that half of the employees are satisfied with the communication process in the organization. Open questions discovered many issues, even little ones in effective communication but the employees were able to seek solutions and suggested some. As positive results are seen that the employees do not afraid to address the company's management directly, employees are not interrupted while speaking, do not get the information they do not understand, the managers do not use the terms their employees do not understand, employees are not often overwhelmed with the information, not facing a misunderstanding from manager's side. On the other hand, the organization may support more communication between departments and provide more information to employees to feel well informed.

The current research has several implications for theory and practice. At the theoretical level, it extends the research from the fields of interpersonal and managerial communication about specific barriers occurring in the organization and summarized the similarities in other research conducted in this field. It also offers practical information for organizations, managers and human resources professionals to learn more about barriers and try to avoid them in daily work life. For scholars, it provides the base about barriers to effective communication and gives the opportunity to continue exploring this area.

This research required addressing several limitations. Firstly, the sample size is too small, only 25,67 % of employees participated. Secondly, the survey was conducted before the COVID-19 pandemic crisis has started, therefore there can be some differences between that time and now-adays. Thirdly, the questionnaire could go deeper and seek more details on barriers to effective communication. If the research will be conducted repeatedly with a higher number of respondents, after COVID-19 pandemic crisis and with detailed questions in the questionnaire, it can offer an even better overview of barriers occurring in the organizations.

REFERENCES

- Adu-Oppong, A. A., & Agyin-Birikorang, E. (2014). Communication in the workplace: Guidelines for improving effectiveness. Global Journal of Commerce & Management Perspective, 3(5), 208-213. Retrieved from https://www.researchgate.net/publication/304782482_ COMMUNICATION_IN_THE_WORKPLACE_GUIDELINES_FOR_IMPROVING_ EFFECTIVENESS
- Ansari, A. (2021). The Role of Interpersonal Communication Skills in Human Resource and Management. International Journal of Science and Research (IJSR), 10(11), 2319-7064. DOI: 10.21275/SR211118020911
- Beqiri, G. (2018). Key interpersonal skills you need to improve. https://virtualspeech.com/blog/ interpersonal-communication-skills
- Demchenko, V., Khoroshevskaya, J., & Krukov, K. (2021). Communication Barriers of a Construction Company's Network Management. International Science and Technology Conference (FarEastCon 2020), 1079 (2021) 032089. DOI: 10.1088/1757-899X/1079/3/032089
- Disilva, X., & Arun, A. (2017). Communication Barrier Encountered By Employees Working In Hotel Industry. Advances in Social Sciences Research Journal, 4(14) 180-184. DOI: 10.14738/assrj.414.3509
- Jonsdottir, I. J., Fridriksdottir, K. (2019). Active Listening: Is it the forgotten dimension in managerial communication?. International Journal of Listening, 1932-586X. DOI: 10.1080/10904018.2019.1613156
- Lamichhane, Y. R. (2016). Non-verbal skills: Unavoidable in communication. Repositioning 1(1), 91–98. https://doi.org/10.3126/repos.v1i0.16046
- Lusiawati, I. (2019). Public Service in Interpersonal Communication. International Journal for Educational and Vocational Studies, 1 (5), 484-488. DOI: https://doi.org/10.29103/ijevs. v1i5.1719
- Naumovski, V., Dana L.-P., Pesakovic, G., Fidanoski, F. (2016). The importance of interpersonal communication in public administration. ResearchGate. https://www.researchgate.net/ publication/313642138_The_importance_of_interpersonal_communication_in_public_ administration/stats
- Nguyen, T., White, S., Hall, K., Bell., R., L., Ballentine, W. (2019). Emotional Intelligence and Managerial Communication. American Journal of Management, 19(2). DOI: 10.33423/ ajm.v19i2.2068
- Oladiran, T., Burghate, S. (2016). Strategies to Improve Managerial Communication. International Journal of Commerce and Management Studies (IJCAMS), 6(1), 2456-3684. https:// ijcams.com/wp-content/uploads/2021/03/STRATEGIES-TO-IMPROVE-MANAGERI-AL-COMMUNICATION.pdf
- Rana, R. (2013). Effective Communication in a Diverse Workplace. International Journal of Enhanced Research in Management and Computer Applications, 2(2). Retrieved from https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2225761.
- Rani, K., U. (2016). Communication barriers. Journal of English language and literature (JO-ELL), 3(2), pp. 74-76. https://joell.in/vol-3-issue-2-2016
- Saadi, I., A. (2018). The Impact of Interpersonal Communication Skills on Sustainable Development. FKP Postgraduate Colloquium 2018, Cultivating Excellence Through Research. https://www.researchgate.net/profile/Ibraheem-Saadi-3/publication/337760134_ The_Impact_of_Interpersonal_Communication_Skills_on_Sustainable_Development/links/5de8f571299bf10bc340d206/The-Impact-of-Interpersonal-Communication-Skills-on-Sustainable-Development.pdf

- Vasilev, V. & Stefanova, D. (2021). Complex communication barriers in the organization in a crisis context. KNOWLEDGE – International Journal, Vol. 49.1. Retrieved from https:// www.researchgate.net/publication/357553890
- Venter, E. (2019). Challenges for meaningful interpersonal communication in a digital era. HTS Teologiese Studies/ Theological Studies 75(1), a5339. DOI: 10.4102/hts.v75i1.5339
- Yusof, A., N., A. M., & Rahmat, N., H. (2020). Communication barriers at the workplace: A case study. European Journal of Education Studies, 7(10), 2501-1111. DOI: 10.46827/ejes. v7i10.3299
- Zahid, A., Qinghe, Z., & Sohail, A. (2021). Communication Barriers and Process of Feedback in Social Interactions. *Mediterranean Journal of Social & Behavioral Research*, *5(1)*, *3-12*. DOI: 10.30935/mjosbr/10804



Virtual vs. Live Conferences: Measuring the Participants Memorable Conference Experience

Dora Rašan¹ 💿 Marina Laškarin Ažić² 💿

Received: June 18, 2022 Revised: November 24, 2022 Accepted: December 28, 2022 Published: December 30, 2022

Keywords:

Virtual conference; Memorable conference experience; Behavioural intentions; Independent t-test

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-Non-Commercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission. Abstract: Due to Croatian government measures to prevent the spread of COVID - 19, Croatian citizens have been in several lockdowns, which causally affected the normal function of society as a whole. One of the most common consequences of such restrictions was the reduction in social contacts and reliance on virtual contacts in the online environment. The scientific community and any other sphere of business had to change its processes, particularly in organizing conferences. Virtual conferences from "rescue solution" became "pleasant with the useful solution" for organizing committees but also for participants. The purpose of this study is to detect the differences in the memorable experiences of participants concerning different auality elements in virtual and online environments. Furthermore, the aim is to detect predictors of behavioural intentions for both virtual and live conferences. For the purpose of this research, the authors have created a questionnaire, which was distributed online and onsite. The T-test and regression analysis were conducted on a sample of 256 respondents. Results confirmed significant differences between virtual and live conferences in ten variables. Furthermore, rearession results revealed that only "Memorable experience" is the common predictor for both virtual and live conferences.

1. INTRODUCTION

usiness tourism is one of the quickest growing segments of tourism (Rogers, 1998) for **D**which the acronym M.I.C.E is used in the scientific literature because it includes the following forms: (a) meetings, (b) incentives, (c) congress/conference and (d) exhibitions (Gračan & Rudančić Lugarić, 2011). This type of tourism, due to high tourist consumption (Štetić, 2007) generates multi-economic advantages for the host's place and at the same time represents one of the most profitable types of tourism (Oppermann & Chon, 1997). Therefore, its participants are of crucial necessity for tourist destinations (Zhang et al., 2007). For that reason, the tourist offer of a destination has adapted to this segment of tourism, with emphasis on specialising activities according to the type of business tourism which participants most frequently visit. The most important element of M.I.C.E tourism is definitely conferences; representing one of the most important sectors of the tourism industry (Šušić & Mojić, 2014). In support of this view is the fact that in the last twenty years, the highest rate of investment was present in the construction/building and reconstruction of congress and conference facilities (Šušić & Mojić, 2014). Due to their importance in the tourism sector, conferences have been chosen as the topic of this research. Conference implies counsel, agreement, meeting to reach an agreement, discussing an issue or a public lecture which usually includes a discussion (Anić et al., 2002). Also, it can be said that it represents an event which sometimes lasts for a few days, and where a certain topic is discussed (Cambridge Dictionary, online).

² University of Rijeka, Faculty of Tourism and Hospitality Management, Primorska 42, Opatija, Croatia



¹ University of Rijeka, Faculty of Tourism and Hospitality Management, Primorska 42, Opatija, Croatia

8th International Scientific Conference ERAZ 2022 Selected Papers

According to Marušić et al. (2019), business is the secondary motif of tourists coming to a destination (4%), and only 18% of business tourists decide to go on a journey to participate in a conference. But, with the occurrence of COVID - 19 and after introducing several lockdowns, there has been a decrease in the number of journeys with the purpose of attending conferences. Due to this, conference organizers have been forced to secure alternative means of this type of event, allowing participants passive and active participation in online surroundings (Pacchioni, 2020). Despite this new practice, the objective of each conference organising board is to continue the contentment of the wishes and needs of participants through the insured high-quality memorable experience. This experience can, in the long term, contribute to a feeling of excitement and satisfaction (Csikszentmihalyi, 1990; Lian Chan & Baum, 2007) which is remembered forever (Larsen, 2007). The memory will in this context represent a single and the most important source for the individual during the return journey and while spreading positive word of mouth (Oh et al., 2007). So, a memorable conference experience is possible to define as the experience of positive intention which forever stays in the participant's memory, and which has mostly been influenced by the affordable conference cost, dealing with broader and more specific topics during the conference, easy access to information about the conference and the opportunities for professional growth and development (Rašan & Laškarin Ažić, 2021).

By researching databases and scientific literature, it has been established that there is only one existing research which dealt with measuring the quality of memorable conference experience (Rašan & Laškarin Ažić, 2021), but so far there are no studies aimed at examining participants on their accomplished conference experience in virtual environments. Therefore, this empirical research is of crucial importance and can later serve as the basis for future research. However, the literature contains scientific articles focused on the research of participants' satisfaction with the quality of a held conference (Chatzigeorgiou et al., 2017; Halim & Mokhtar, 2016; Price, 1993; Riper et al., 2013; Severt et al., 2007), which helped the authors to get a better understanding of the researched field. Rašan & Laškarin Ažić (2021) created a four-dimensional MCE model which measures memorable conference experience, determining in their study the following factors: comprehensive characteristics, outdoor activities, conference organisers and networking. They also came to the result that there is a connection between the main construct and behavioural intentions, or to be more exact, they confirm that affectively loyal participants show stronger behavioural intentions in relation to affectively disloyal participants. On the other hand, Severt et al. (2007) state that major motivators to take part in a conference are activities and possibilities, networking, the benefit of the conference, educational benefits, products and offers, while according to Price (1993) those are education, networking, career improvement and travelling to desirable places. Chatzigeorgiou et al. (2017), by analysing gathered primary data, came to know that the conference quality performance has a positive effect on the experience quality of the participants and that the perceived experience quality, together with individual aspects, has a significant influence on the overall examinee's experience. Also, this research has demonstrated the predictive relationship between the overall service quality and satisfaction with future behavioural intentions of the participants.

As mentioned so far, it is essential for conference organisers to pay more attention to the way participants perceive their experience (Halim & Mokhtar, 2016), as this is the only approach that will allow them to influence the future intentions of participants' attendance. Therefore, the main goal of this research is to examine the overall quality of memorable conference experience and behavioural intentions of participants in virtual and live environments. Starting from the purpose of the research, hereafter are the formulated research aims: (1) to examine and analyse the socio-demographic characteristics of participants in virtual and live conferences; (2) to examine and analyse the key stakeholders of conference environments, memorable conference experience and behavioural intentions of participants; (3) to examine and analyse the differences in the perception of the memorable conference experience for participants who have taken part in live and of those who have taken part in a virtual conference; (4) to examine whether there are significant predictors of behavioural intentions with regard to MCE attributes and type of conferences. Based on the participants' attitudes, the conference organisers will through this research get a more detailed insight into the difference in the quality of virtual and live conferences. In accordance with this, the conference organisers will be able to determine the differences between the wishes and desires of virtual and live participants and will, according to the form of the conference (virtual or live), be able to adapt and enhance their current quality when organising future conferences. This scientific paper is structured in four parts: introduction, methodology, results and conclusion.

2. METHODOLOGY

Measuring the memorable conference experience is based on the MCE model (Rašan & Laškarin Ažić, 2021) which has been modified for the purpose of conducting this research. The following is the presentation of the modified MCE model, or more accurately, its study framework (figure 1).



Figure 1. Study framework: modified MCE model Source: Authors' research

Figure 1 shows that the study's framework contains three constructs, the conference environment, memorable conference experience and behavioural intentions of the participants. In this research, the instrument for gathering primary data was an autonomous survey questionnaire which was distributed in online and onsite form. Accepting the above-mentioned constructs, the designed questionnaire included the examinees' sociodemographic characteristics. The first part of the measuring instrument was the main construct- conference environments, which included three dimensions: (1) comprehensive characteristics, (2) conference organizers and (3) networking. In this part of the questionnaire, all particles were in the form of structured questions where a Likert scale of importance was used for measuring and one (1) presents "not important", and five (5) were "very important". Furthermore, in the second and third parts of the questionnaire, the examinees were asked to, using the Likert scale of agreement from one (1 = I completely agree) to five (5 = I completely disagree) determine the achieved memorable conference experience and their behavioural intentions. The final part of the measuring instrument included statements referring to the examinees' sociodemographic profile, including their age, sex, marital status, level of education and the frequency of participating in conferences.

Moreover, the target group were participants of virtual and live conferences, and participation in research was voluntary without any compensation. It was not possible to apply a random sample in this research, due to the fact that there is no list of passive conference participants, while for active ones, we only have access to the Conference Agenda. Because of this, the research will apply a convenience sample which is based exclusively on the availability of units. Since the appropriate sample size is possible to determine only for random samples, the decision on the size of the sample will depend on the researchers' subjective judgment. In distributing and collecting samples, the researchers used their private contacts, academic experts and students from various scientific fields who frequently attend conferences. An online survey questionnaire has been distributed via email and various social media with the goal of collecting a representative sample. Both ways of collecting, online and onsite, were conducted in the period between the first of December, 2021 and the thirtieth of January, 2022. A descriptive statistical analysis was used to examine the sociodemographic characteristics of the examinees and to determine the average value of all the particles within questioned constructs, while bivariate statistical analysis (independent t-test) was used to estimate the differences in the relevance of the particles' average values. After bivariate statistical analysis, regression analysis was to examine the influence of the independent variables - conference environment and memorable conference experience on the dependent variable - the behavioural intention of the conference participants.

3. **RESULTS**

Descriptive statistics were used to analyse the demographic characteristics of the respondents. The obtained results are presented in Table 1 below.

| | | | FERENCE 131) | VIRTUAL CONFERENCE (N=125) | | |
|---|---------------------|-----------|-----------------|-------------------------------|---------|--|
| Characteristics | | Frequency | Percent | Frequency | Percent | |
| C 1 | Male | 42 | 32.06 | 57 | 45.6 | |
| Gender | Female | 89 | 67.94 | 68 | 54.4 | |
| | Less than 20 | 2 | 1.53 | 6 | 4.58 | |
| | 21-30 | 39 | 29.77 | 34 | 25.95 | |
| Age | 31-40 | 45 | 34.35 | 30 | 22.90 | |
| | 41-50 | 22 | 16.79 | 26 | 19.85 | |
| | More then 51 | 23 | 17.56 | 29 | 22.14 | |
| A | Married | 68 | 51.91 | 48 | 36.64 | |
| Marital status | Single | 25 | 19.08 | 36 | 27.48 | |
| | In a relationship | 38 | 29.01 | 41 | 31.30 | |
| | High school diploma | 34 | 25.96 | 28 | 21.37 | |
| E desertion | Bachelor's degree | 13 | 9.92 | 19 | 14.50 | |
| Education | Magister degree | 66 | 50.38 | 49 | 37.40 | |
| | PhD | 18 | 13.74 | 29 | 22.14 | |
| Working status | Employed | 113 | 86.26 | 91 | 69.47 | |
| | Unemployed | 3 | 2.29 | 5 | 3.82 | |
| | Student | 15 | 11.45 | 26 | 19.85 | |
| | Retired | 1 | 0.76 | 3 | 2.29 | |
| En anti-in di | 1 per year | 103 | 78.63 | 77 | 61.60 | |
| Frequency of participation in conferences | 2 -3 per year | 26 | 19.85 | 41 | 32.80 | |
| III conferences | 3 and more | 2 | 1.53 | 7 | 5.60 | |

Table 1. Demographic characteristics of respondents

Source: Authors' research

As shown in Table 1, in both samples the representation of women was higher compared to men. Specifically, 67.94% of women participated in live conferences and 54.4% in virtual conferences. In terms of age, in the observed samples the largest number of respondents was between the ages of 21 and 30 (29.77% at live conferences and 25.95% at virtual conferences). In contrast, the lowest number of respondents at live (1.53%) and virtual (4.58%) conferences were under 20 years of age. More than half of the participants in the live conference stated that they were married (51.91%) with a master's degree (50.38%). Also, the largest number of participants in virtual conferences are married (36.64%) with a master's degree (37.40%) and a doctoral degree (22.14%). In both cases, the majority of respondents are employed (79.69%) and attend conferences most often once a year (61.60%).

Descriptive statistics were applied to compute means (M) and standard deviations of conference attributes for virtual and live conferences. To compare the mean scores of conference attributes (conference environment, overall quality of memorable experience, behavioural intentions) between two independent groups (virtual vs. live) independent samples t-test was applied. Although the t-test assumes a normal distribution of data, the normality test has shown asymmetry among analysed data. However, statisticians have proven that the t-test is a robust method, and it can be used even when data does not show normal distribution (Raspor, 2012). Moreover, since in social sciences, the normal distribution is hard to realize, some authors (Opić, 2011) justify the use of parametric statistics with the "Central limit theorem" even if the normal distribution isn't achieved. According to Opić (2011), this is allowed since the standard error decreases as the sample size rises. Thus, the authors decided to proceed with a t-test to determine whether there is a statistically significant difference among mean scores for observed variables.

The results of descriptive and bivariate statistical analysis are presented below. Table 2 presents the significance of the difference in mean scores between live participants (N = 131) and online conferences (N = 125).

Both participants of live and virtual conferences graded "*Comprehensive characteristics*" with the highest average scores. Results also indicate that the dimension "Networking" had the lowest average score (3.65) among participants of virtual conferences. Moreover, it is noticeable that participants of virtual conferences better graded the majority of statements (15 out of 19), while participants of virtual conferences graded only the following four attributes higher: high expertise of keynote speakers (4.50), the relevance of topics for the transfer of new knowledge (4.63), easy availability of conference information (4.68), affordable conference price (4.64). Among participants of live conference to my colleagues" (1.056) and "I plan to participate in this conference again" (1.021), and in three statements among participants of virtual conferences: "conference networking opportunities" (1.087), "opportunities to make new contacts" (1.139), "opportunities to talk to other conference participants" (1.114), indicating that the data are here more spread out in relation to the mean.

According to T-test results, there are some significant discrepancies between live and virtual conferences in their attributes. In the first dimension "*Comprehensive characteristics*" within the construct conference environment, all variables except one showed insignificant differences. Respondents' opinions only significantly differ in the variable "relevance of topics for the transfer of new knowledge, insights and current events to the participants" (p=0.000). Results show that participants of virtual conferences gave significantly higher scores for this attribute of the conference. For the second dimension "*Conference organization*", independent samples t-test results have shown significant differences in two variables "easy availability of conference information" (p=0.020) and "interactive workshops" (p=0.000). Namely, participants of a virtual conference gave higher scores for easy availability of conference information than those who participated in a live conference. In contrast, interactive workshop as an attribute was significantly better graded among participants of live conferences. In addition, the t-test indicated differences for all networking variables (the third dimension of conference environment construct), where participants of live conferences significantly better-graded networking opportunities during the conference. Interesting results of the t-test are noticeable for the construct "*Memorable conference experience*". For this construct, participants of the live conference had significantly better evaluated the quality of memorable conference experience compared to those who participated online. For the construct "*Behavioural intentions*", participants of the live conference better graded all three variables, however, the t-test indicated a significant difference in only one variable – positive WOM (word of mouth) (p=0.000).

| | Participa | ints of live | Partici | pants of onference | t - tes | Sig. (2 tailed) |
|---|-----------|--------------|---------|-----------------------|---------|--------------------|
| Attributes | | SD | M | SD | | |
| CONFERENCE ENVIRONMENT | 4.32 | - | 4.10 | - | | |
| Comprehensive characteristics | 4.42 | - | 4.43 | - | | |
| Timely topics. | 4.53 | 0.599 | 4.42 | 0.732 | 1.316 | 0.189 |
| High expertise of keynote speakers. | 4.41 | 0.743 | 4.50 | 0.667 | -1.038 | 0.300 |
| Elaboration of specific and broader topics. | 4.42 | 0.754 | 4.23 | 0.863 | 1.857 | 0.064 |
| Combination of lecturers: scientists and practitioners. | 4.37 | 0.748 | 4.36 | 0.797 | 0.145 | 0.884 |
| Relevance of topics for the transfer of new knowledge, insights and current events to the participants. | 4.36 | 0.657 | 4.63 | 0.532 | -3.664 | 0.000* |
| Conference organization | 4.29 | - | 4.22 | - | | 1 |
| Easy availability of conference information. | 4.50 | 0.661 | 4.68 | 0.590 | -2.350 | 0.020* |
| Affordable conference price. | 4.48 | 0.798 | 4.64 | 0.559 | -1.854 | 0.065 |
| Defined conference topic. | 4.31 | 0.812 | 4.22 | 0.819 | 0.876 | 0.382 |
| High quality introductory and closing sessions. | 4.22 | 0.797 | 4.11 | 0.961 | 0.989 | 0.324 |
| Interactive workshops. | 3.97 | 0.992 | 3.45 | 0.954 | 4.283 | 0.000* |
| Networking | 4.25 | - | 3.65 | - | | |
| Conference networking opportunities. | 4.15 | 0.707 | 3.45 | 1.087 | 4.382 | 0.000* |
| Opportunities to make new contacts. | 4.41 | 0.606 | 3.65 | 1.139 | 6.727 | 0.000* |
| Opportunities to talk to other conference participants. | 4.19 | 0.860 | 3.66 | 1.114 | 4.221 | 0.000* |
| MEMORABLE CONFERENCE EXPERIENCE | 4.24 | - | 3.91 | - | | |
| I have wonderful memories of this conference. | 4.23 | 0.891 | 3.97 | 0.813 | 0.092 | 0.015* |
| I will not forget my experience from the conference. | 4.22 | 0.862 | 3.89 | 0.854 | 0.867 | 0.002* |
| I will remember many positive things from the conference. | 4.27 | 0.823 | 3.89 | 0.854 | 0.487 | 0.000* |
| BEHAVIOURAL INTENTIONS | 4.17 | - | 3.98 | - | | |
| I plan to participate in this conference again. | 3.98 | 1.056 | 3.94 | 0.910 | 0.268 | 0.789 |
| I plan to recommend this conference to my colleagues. | 4.16 | 1.021 | 4.06 | 0.821 | 0.833 | 0.405 |
| I will say positive things about this conference. | 4.37 | 0.896 | 3.92 | 0.829 | 4.132 | 0.000* |

| Table 2. The results comparison of MCE model attributes between virtual |
|--|
| and live conferences participants |

* p<0.05

Source: Authors' research

To test internal consistency and the reliability of each construct/dimension, the authors conducted a reliability analysis (Cronbach's alpha). The α coefficient ranged from 0.714 to 0.925. The highest value of α coefficient was for the "Networking" 0.925, following "Memorable experience" 0.922, "Behavioural intentions" 0.822, "Comprehensive characteristics" 0.763, and "Conference organization" 0.714. All values were greater than the cut-off value of 0.70 (Baggio & Klobas, 2011), hence, all factors and their variables were kept for further analysis. Regression analysis is conducted on a sample of 131 respondents of live conferences, and 125 respondents of virtual conferences, which satisfies recommended criteria of 100 respondents in a sample, and the smallest proposed ratio of the number of elements in a sample and the number of independent variables (5:1) (Raspor, 2012). In particular for *Model 1*, the ratio is 9:1 (134 respondents and 15 independent variables), and for *Model 2* ratio is 8:1 (125 respondents and 15 independent variables).

In the first model (*Live conference*), the OLS results showed that only one variable is a significant predictor of the participants' behaviour intentions to live conferences. Results imply that respondents who remember many positive things from the conference are most likely to come again and recommend their memorable conference experience to friends and colleagues (p=0.000). In other words, it is a variable that is part of the construct of a *"Memorable conference experience"* (abbreviation MCE) for the participants. The obtained results show that in this virtual model, the VIF coefficients range from 1.2706 to 3.701. Therefore, the VIF coefficients for all regressors are less than 10 and very close to 1, which means that there is no serious problem of multicollinearity among regressors.

When it comes to regression results of the second model (Virtual conferences), results showed that 5 out of 15 variables turn out to be significant predictors of the behavioural intentions of virtual conferences. The results confirm that participants of virtual conferences who are most likely to have behavioural intentions are those who believe that interactive workshops and the relevance of topics for the transfer of new knowledge, insights and current events (p=0.024) are important in creating memorable conference experience. Hence, of the five possible variables within the "Conference characteristic" (CC) dimension, only the aforementioned variable showed a predictor relationship with the behavioural intentions of conference participants. Also, within the "Conference organizers" (CO) dimension, only one variable, more precisely the "affordable conference price" (p = 0.043), showed a significant correlation with the behavioural intentions of the participants. Furthermore, within the conference environment construct, none of the "Networking" (N) variables proved to be a statistically significant predictor of the virtual participants' behavioural intentions. Additionally, the results also imply that participants who had wonderful memories (p=0.000) and will not forget (p=0.002) the recalled conference are the ones who will come again and recommend their memorable conference experience to friends and colleagues. Thus, two of the three variables within the construct "Memorable conference experience" showed a significant predictor of the participant's behavioural intentions. In this virtual model, all values of the VIF coefficient are also within the given frames, which means that the regressors do not have multicollinearity issues.

Although, this is the first study that tests the relationships between MCE and behavioural intentions, there are some previous results on memorable experiences and behavioural intentions comparable with current ones. Namely, our results agree with results reported by Coudounaris & Sthapit (2017) who noted that hedonism (enjoying the trip, being thrilled with experience, exciting experience) had a positive effect on tourist behavioural intention toward the destination. Similarly, Yu et al. (2019) reported that memorable experience such as hedonism (thrilled about having new experiences, excitement, enjoyment in tourism experience etc.) positively influence revisit intentions. Moreover, current study results align with Sthapit et al. (2019) findings who reported that memorability is a significant predictor of tourists' behavioural intentions.

Table 3. Behavioural intentions predictors: Virtual vs. Live

| Model 1: Live (n=131) |
|-----------------------|
|-----------------------|

| Independent variable | В | S.E. | sig. | Tolerance | VIF |
|---|--|---|---|---|---|
| Constant | 0.468 | 0.638 | 0.464 | | |
| CC1: Timely topics. | 0.102 | 0.095 | 0.285 | 0.737 | 1.357 |
| CC2: High expertise of keynote speakers. | 0.016 | 0.084 | 0.852 | 0.620 | 1.614 |
| CC3: Elaboration of specific and broader topics. | -0.108 | 0.088 | 0.224 | 0.543 | 1.843 |
| CC4: Combination of lecturers: scientists and practitioners. | 0.055 | 0.074 | 0.456 | 0.784 | 1.276 |
| CC5: Relevance of topics for the transfer of new knowledge, | 0.065 | 0.087 | 0.459 | 0.733 | 1.365 |
| insights and current events to the participants. | 0.005 | 0.087 | 0.439 | 0.755 | 1.505 |
| CO1: Easy availability of conference information. | -0.046 | 0.101 | 0.647 | 0.540 | 1.851 |
| CO2: Affordable conference price. | 0.102 | 0.091 | 0.266 | 0.450 | 2.220 |
| CO3: Defined conference topic. | 0.008 | 0.085 | 0.927 | 0.499 | 2.003 |
| CO4: High quality introductory and closing sessions. | -0.019 | 0.082 | 0.819 | 0.557 | 1.794 |
| CO5: Interactive workshops. | 0.005 | 0.064 | 0.933 | 0.594 | 1.684 |
| N1: Conference networking opportunities. | 0.020 | 0.104 | 0.850 | 0.440 | 2.270 |
| N2: Opportunities to make new contacts. | -0.046 | 0.129 | 0.723 | 0.389 | 2.568 |
| N3: Opportunities to talk to other conference participants. | -0.082 | 0.088 | 0.350 | 0.421 | 2.375 |
| MCE1: I have wonderful memories of this conference. | 0.064 | 0.091 | 0.482 | 0.366 | 2.730 |
| MCE2: I will not forget my experience from the conference. | 0.023 | 0.105 | 0.825 | 0.291 | 3.441 |
| MCE3: I will remember many positive things from the | 0.699 | 0.114 | 0.000 | 0.270 | 3.701 |
| | | 0.114 | 0.000 | 0.270 | 5.701 |
| R ² =0.604; (F=10.849), p<0.05 | | | | | |
| R ² =0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) | B | S.E. | sig. | Tolerance | VIF |
| Model 2: Virtual (n=125) | | S.E. 0.473 | sig. 0.294 | Tolerance | VIF |
| R ² =0.604; (F=10.849), p<0.05 <i>Model 2: Virtual (n=125)</i> Independent variable Constant | B | | - | Tolerance | |
| R ² =0.604; (F=10.849), p<0.05 <i>Model 2: Virtual (n=125)</i> Independent variable Constant CC1: Timely topics. | B 0.498 | 0.473 | 0.294 | | 3.278 |
| R ² =0.604; (F=10.849), p<0.05 <i>Model 2: Virtual (n=125)</i> Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. | B 0.498 -0.112 | 0.473 0.101 | 0.294 0.272 | 0.305 | 3.278 2.679 |
| R ² =0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. | B 0.498 -0.112 -0.074 | 0.473 0.101 0.101 | 0.294 0.272 0.465 | 0.305 0.373 | 3.278 2.679 3.714 |
| R ² =0.604; (F=10.849), p<0.05 <i>Model 2: Virtual (n=125)</i> Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, | B 0.498 -0.112 -0.074 -0.055 -0.123 | 0.473 0.101 0.101 0.092 0.091 | 0.294 0.272 0.465 0.547 0.179 | 0.305 0.373 0.269 0.322 | 3.278 2.679 3.714 3.108 |
| R ² =0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 | 0.473 0.101 0.101 0.092 0.091 0.103 | 0.294 0.272 0.465 0.547 0.179 0.024 | 0.305 0.373 0.269 0.322 0.557 | 3.278 2.679 3.714 3.108 1.796 |
| R ² =0.604; (F=10.849), p<0.05 <u>Model 2: Virtual (n=125)</u> Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 | 0.305 0.373 0.269 0.322 0.557 0.461 | 3.278 2.679 3.714 3.108 1.796 2.169 |
| R ² =0.604; (F=10.849), p<0.05 <u>Model 2: Virtual (n=125)</u> Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 | 3.278 2.679 3.714 3.108 1.796 2.169 2.098 |
| R ² =0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. CO3: Defined conference topic. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 -0.109 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 0.084 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 0.196 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 0.359 | 3.278 2.679 3.714 3.108 1.796 2.169 2.098 2.788 |
| R ² =0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. CO3: Defined conference topic. CO4: High quality introductory and closing sessions. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 -0.109 0.024 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 0.084 0.076 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 0.196 0.753 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 0.359 0.315 | 3.278 2.679 3.714 3.108 1.796 2.169 2.098 2.788 3.178 |
| R ² =0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. CO3: Defined conference topic. CO4: High quality introductory and closing sessions. CO5: Interactive workshops. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 -0.109 0.024 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 0.084 0.076 0.091 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 0.196 0.753 0.002 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 0.359 0.315 0.223 | 3.278 2.679 3.714 3.108 1.796 2.169 2.098 2.788 3.178 4.493 |
| R ² =0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. CO3: Defined conference topic. CO4: High quality introductory and closing sessions. CO5: Interactive workshops. N1: Conference networking opportunities. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 -0.109 0.024 0.285 0.172 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 0.084 0.076 0.091 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 0.196 0.753 0.002 0.116 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 0.359 0.315 0.223 0.121 | 3.278 2.679 3.714 3.108 1.796 2.169 2.098 2.788 3.178 4.493 8.284 |
| R²=0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. CO3: Defined conference topic. CO4: High quality introductory and closing sessions. CO5: Interactive workshops. N1: Conference networking opportunities. N2: Opportunities to make new contacts. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 -0.109 0.024 0.285 0.172 -0.082 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 0.084 0.076 0.091 0.109 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 0.196 0.753 0.002 0.116 0.469 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 0.359 0.315 0.223 0.121 0.102 | 3.278 2.679 3.714 3.108 1.796 2.169 2.098 2.788 3.178 3.178 4.493 8.284 9.829 |
| R²=0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. CO3: Defined conference topic. CO4: High quality introductory and closing sessions. CO5: Interactive workshops. N1: Conference networking opportunities. N2: Opportunities to make new contacts. N3: Opportunities to talk to other conference participants. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 -0.109 0.024 0.285 0.172 -0.082 -0.056 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 0.084 0.076 0.091 0.103 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 0.196 0.753 0.002 0.116 0.469 0.630 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 0.359 0.315 0.223 0.121 0.102 0.102 | 3.278 2.679 3.714 3.108 1.796 2.169 2.098 2.788 3.178 4.493 8.284 9.829 9.809 |
| R²=0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. CO3: Defined conference topic. CO4: High quality introductory and closing sessions. CO5: Interactive workshops. N1: Conference networking opportunities. N2: Opportunities to make new contacts. N3: Opportunities to talk to other conference participants. MCE1: I have wonderful memories of this conference. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 -0.109 0.024 0.285 0.172 -0.082 -0.056 0.455 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 0.084 0.076 0.091 0.109 0.109 0.113 0.115 0.116 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 0.196 0.753 0.002 0.116 0.469 0.630 0.000 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 0.359 0.315 0.223 0.121 0.102 0.189 | 3.278 2.679 3.714 3.108 1.796 2.169 2.098 2.788 3.178 4.493 8.284 9.829 9.809 5.301 |
| R²=0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. CO3: Defined conference topic. CO4: High quality introductory and closing sessions. CO5: Interactive workshops. N1: Conference networking opportunities. N2: Opportunities to make new contacts. N3: Opportunities to talk to other conference participants. MCE1: I have wonderful memories of this conference. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 -0.109 0.024 0.285 0.172 -0.082 -0.056 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 0.084 0.076 0.091 0.103 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 0.196 0.753 0.002 0.116 0.469 0.630 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 0.359 0.315 0.223 0.121 0.102 0.102 | 3.278 2.679 3.714 3.108 1.796 2.169 2.098 2.788 3.178 4.493 8.284 9.829 9.809 5.301 |
| R ² =0.604; (F=10.849), p<0.05 Model 2: Virtual (n=125) Independent variable Constant CC1: Timely topics. CC2: High expertise of keynote speakers. CC3: Elaboration of specific and broader topics. CC4: Combination of lecturers: scientists and practitioners. CC5: Relevance of topics for the transfer of new knowledge, insights and current events to the participants. CO1: Easy availability of conference information. CO2: Affordable conference price. CO3: Defined conference topic. CO4: High quality introductory and closing sessions. CO5: Interactive workshops. N1: Conference networking opportunities. N2: Opportunities to make new contacts. N3: Opportunities to talk to other conference participants. MCE1: I have wonderful memories of this conference. | B 0.498 -0.112 -0.074 -0.055 -0.123 0.236 0.001 0.218 -0.109 0.024 0.285 0.172 -0.082 -0.056 0.455 | 0.473 0.101 0.101 0.092 0.091 0.103 0.102 0.106 0.084 0.076 0.091 0.109 0.109 0.113 0.115 0.116 | 0.294 0.272 0.465 0.547 0.179 0.024 0.996 0.043 0.196 0.753 0.002 0.116 0.469 0.630 0.000 | 0.305 0.373 0.269 0.322 0.557 0.461 0.477 0.359 0.315 0.223 0.121 0.102 0.189 | VIF 3.278 2.679 3.714 3.108 1.796 2.169 2.098 2.788 3.178 4.493 8.284 9.829 9.809 5.301 5.729 8.109 |

R²=0.690; (F=15.025), p<0.05

Source: Authors' research

4. CONCLUSION

The COVID-19 crisis has affected the changes in the functioning of society as a whole. This can be seen in the suspension of a large number of events, and this decision made by the National board has affected the need to create new ways of conducting events. For that reason, conference organisers had to ensure alternative ways of conducting conferences for its participants, introducing the possibility of online participation. Regardless of this fact, the conference's organising board still aims to ensure a high-quality memorable experience for its participants. Therefore, the contribution of this research is of exceptional importance because it represents the first research in literature dealing with the comparison of the virtual and live memorable conference experience. At the same time, it offers mite from the managerial aspect, to be precise for the conference organisers who can based on the analysed data, determine the differences in wants and needs of participants who take part in the virtual and live conference with the goal of determining the strengths and weaknesses in the existing elements of the offer. The obtained findings indicate that there are statistically significant differences between live and virtual conferences in 10 of the 19 possible variables. Comparing both models, virtual and live, there are also differences in the prediction of participants' behavioural intentions. In live conference participants, while in the case of virtual conferences, there were five variables. On the basis of the above findings, they will be able to adapt organisational elements regarding the way the conference is held (virtual or live).

Despite the fact that this paper provides insight into the current state of memorable conference experiences in virtual and live environments, it also has some drawbacks. The overall number of examinees in this research is 256, but if we look at the samples based on the groups of virtual (N=125) and live (N=131) participants, it is possible to see that we have small representative samples. Subsequently, future research should include a higher number of participants in order to get a more representative sample, where research should be conducted before, during and after the conference. Since most of the examinees are the authors' private contacts, future research should include other conference participants. In addition, this research was conducted in a small geographical area, in the Republic of Croatia to be precise, so the suggestion is to conduct the research in other countries around the world. With the goal of achieving as much detailed insight into the wishes and wants of participants, it would be desirable to apply the MCE model in the future to examine the memorable conference experience from the professional aspect.

REFERENCES

- Anić, Š., Klaić, N., & Domović, Ž. (2002). *Rječnik stranih riječi-Tuđice, posuđenice, izrazi, kratice i fraze*. Zagreb: Sani-plus.
- Baggio, R., & Klobas, J. (2011). *Quantitative methods in tourism: A handbook*. Bristol: Channel View Publications.
- Cambridge Dictionary (nd.) Conference. In Cambridge Dictionary (online). Available at https://dictionary.cambridge.org/dictionary/english/conference [02 November 2021].
- Csikszentmihalyi, M. (1990). Flow: The psychology of optimal experience-steps toward enhancing the quality of life. New York: Harper Perennial Modern Classics.
- Chatzigeorgiou, C., Christou, E., & Simeli, I. (2017). Delegate satisfaction from conference service quality and its impact on future behavioural intentions. In C. Sarmaniotis, G. Wright (Ed.), *Proceedings of ICCMI 2017*. Presented at 5th ICCMI 2017, Thessaloniki, Greece, 21-23 June 2017 (pp. 532-544). Thessaloniki: Alexander Technological Institute of Thessaloniki.
- Coudounaris, D., & Sthapit, E. (2017). Antecedents of memorable tourism experience related to behavioral intentions. *Psychology of Marketing*, 34, 1084-1093.
- Gunawardena, C. N., & Zittle, F. J. (1997). Social presence as a predictor of satisfaction within a computer-mediated conferencing environment. *American journal of distance education*, 11(3), 8-26.
- Gračan, D., & Rudančić Lugarić, A. (2011). Business tourism–Modern form for improvement of the competitiveness of Croatian tourism. *Ekonomska misao i praksa*, 5(2), 579-590.

- Halim, A. H. A., & Mokhtar, A. R. M. (2016). Creating memorable visitor experiences by assessing the satisfaction level and behavioural consequences of attendees. *Procedia Economics and Finance*, 37(1), 1-6. https://doi.org/10.1016/S2212-5671(16)30084-3
- Larsen, S. (2007). Aspects of a psychology of the tourist experience. *Journal of Hospitality and Tourism*, 7(1), 7-18. https://doi.org/10.1080/15022250701226014
- Lian Chan, J. K., Baum, T. (2007). Ecotourists' Perception of Ecotourism Experience in Lower Kinabatangan, Sabah, Malaysia. *Journal of Sustainable Tourism*, 15(5), 574–590. doi:10.2167/jost679.0
- Marušić, Z., Čorak, S., Ivandić, N., Beroš, I., & Ambrušec, I. (2019). *Stavovi i potrošnja turista u Hrvatskoj -TOMAS istraživanje*. Zagreb: Institut za turizam.
- Oh, H., Fiore, A. M., & Jeoung, M. (2007). Measuring experience economy concepts: Tourism applications. *Journal of travel research*, 46(2), 119-132. https://doi. org/10.1177/0047287507304039
- Opić, S. (2011). Testiranje normalnosti distribucije u istraživanjima odgoja i obrazovanja. *Školski vjesnik, 60*, 181-197.
- Oppermann, M., & Chon, K. S. (1997). Convention participation decision-making process. Annals of tourism Research, 24(1), 178-191. https://doi.org/10.1016/S0160-7383(96)00049-7
- Pacchioni, G. (2020). Virtual conferences get real. Nature Reviews Materials, 5(1), 167-168.
- Price, C. (1993). An empirical study of the value of professional association meetings from the perspective of attendees, In doctoral dissertation. Virginia: Virginia Polytechnic and State University.
- Raspor, S. (2012). Multivarijatna statistička analiza zadovoljstva klijenata u hotelskoj industriji. In Doktorska disertacija. Opatija: Fakultet za menadžment u turizmu i ugostiteljstvu.
- Rašan, D. & Laškarin Ažić, M. (2021). Measuring student's memorable conference experience and behavioural intentions. In R. Alkier, S. Marković, M. Perić (Ed.), *ToSEE – Smart, Experience, Excellence & ToFEEL – Feelings, Excitement, Education, Leisure*. Presented at 6th International Scientific Conference ToSSE 2021, Opatija, 30 June – 02 July, 2021 (pp. 657-672). Opatija: Faculty of Tourism and Hospitality Management, University of Rijeka. https://doi.org/10.20867/tosee.06.44
- Riper, C., Riper, C., Kyle, G. T., & Lee, M. E. (2013). Understanding how social networking influences perceived satisfaction with conference experiences. *Annals of Leisure Research*, 16(1),104–114.
- Rogers, T. (1998). *Conference: A twenty-first century industry*. Boston: Addison Wesley Longman Publishing.
- Severt, D., Wang, Y., Chen, P.-J., & Breiter, D. (2007). Examining the motivation, perceived performance, and behavioral intentions of convention attendees: Evidence from a regional conference. *Tourism Management*, 28(1), 399–408.
- Sthapit, E., Del Chiappa, G., Coudounaris, D., & Björk, P. (2019). Tourism experiences, memorability. *Tourism Review*, 75(3), 533-558. https://doi.org/10.1108/TR-03-2019-0102
- Štetić, S. (2007). Posebni oblici turizma. Beograd: Forma B.
- Šušić, V., & Mojić, J. (2014). Congress tourism as a market niche of business tourism. *Economic Themes*, *52*(4), 513-530.
- Zhang, H. Q., Leung, V., & Qu, H. (2007). A refined model of factors affecting convention participation decision-making. *Tourism Management*, 28(4), 1123-1127. https://doi.org/10.1016/j. tourman.2006.07.008
- Yu, C., Chang, W.-C., & Ramanpong, J. (2019). Assessing Visitors' Memorable Tourism Experiences (MTEs) in Forest Recreation Destination: A Case Study in Xitou Nature Education Area. *Forests*, 10, 1-15.



Typology of Social Paradigm: Literature Review of Tourist's Gastronomic Experience

Dora Rašan¹ 💿 Kristina Pilko²

Received: June 18, 2022 Revised: November 24, 2022 Accepted: December 28, 2022 Published: December 30, 2022

Keywords:

Experience economy; Gastronomic experience; Paradigm; Content analysis

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-Non-Commercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission.

Abstract: In an examination of the tourist's gastronomic experience, scientific experts approach it in different ways, depending on the research paradiam to which they belong. In scientific literature, there are paradiams such as positivism, constructivism, criticism, postmodernism, and transmodernism. The purpose of this study was to analyze 55 scientific articles on the tourist's gastronomic experience according to key terms (ontology, epistemology, and methodology) and thus determine their affiliation to the appropriate paradigm. The empirical research results indicate that there are no studies of postmodernism and transmodernism, but there are studies of positivist (70.91%), critical (18.18%), and constructivist (10.91%) research views. Therefore, in future research, scientists should direct their gaze precisely in the direction of these paradigms. Further research findings indicate that the largest number of analyzed studies published in the last three years is in Asia (40%), North America (34.55%), and Europe (18.18%) where the authors predominantly apply a quantitative research approach (69.09%). The contribution of this research is reflected in the improved theoretical knowledge of the researched issues as well as the application of the same in determining the papers with regard to the type of paradigm to which the authors of selected papers belong. Also, this study is the first in conducting a paradigmatic analysis of gastronomic tourism studies.

1. INTRODUCTION

The experience of consuming food is a key behavior that can complement the sensory, cultural, social, and epistemic motivation of tourists (Correia et al., 2020). Tourist pleasure is actually his/her experience based on a product or service that has provided him/her with an unexpected level of value or satisfaction (Crotts & Magnini, 2011), and this satisfaction will depend on his/her perspective on how to look at things and situations around him/her. In other words, two gastro tourists travelling together may have different perspectives depending on their position. In this context, the position represents the ideology of each person, more precisely, the paradigm to which they belong. A paradigm denotes a worldview or set of related assumptions common to a particular group of scientists exploring the world (Deshpande, 1983), asking ontological, epistemological, and methodological questions. Ontology implies a view of the nature of reality (Taylor & Medina, 2011), while epistemology deals with the question "What is the nature of the relationship between researchers and what is already known in the literature?" (Guba, 1990). Ontological and epistemological assumptions directly influence the methodological approach of researchers. The methodology is a part of a structure (Guba, 1990) that includes theoretical principles for providing guidance on how research is conducted in the context of a particular paradigm (Sarantakos, 2005).

² University of Rijeka, Faculty of Tourism and Hospitality Management; Primorska 46, Opatija, Croatia



¹ University of Rijeka, Faculty of Tourism and Hospitality Management; Primorska 46, Opatija, Croatia

Although scholars have argued sharply over the 20th century about the superiority of their paradigms, at the beginning of the 21st century these wars finally came to an end as scientists come to realize that no research paradigm is superior to another (Taylor & Medina, 2011). It is important to emphasize that each scientific expert applies a different approach to research depending on the paradigm to which they belong. The following scientific paradigms are most often distinguished in scientific literature: (1) positivism, (2) constructivism, (3) criticism, (4) postmodern, and (5) transmodern. Based on the above, the main goal of this research was based on the theoretical knowledge of scientific paradigms (described in Chapter 2) to analyze in detail the scientific papers on the tourist's gastronomic experience according to key terms (ontology, epistemology, and methodology) and thus determine the scientific paradigm. The specific objectives of this research are to conduct a paradigmatic: (a) frequency analysis by year, (b) geographical analysis, and (c) analysis by research type. The qualitative approach in this research is extremely important because paradigmatic analysis of current scientific literature on the tourist's gastronomic experience allows not only a better understanding of this topic but also to determine the representation of each of these paradigms. Such an approach will ensure the reduction of current gaps in the literature and contribute to the improvement of knowledge and understanding of the researched issues. This research can be considered unique because so far there is no research in the literature that has dealt with the paradigmatic analysis of papers in the field of gastronomic tourism. This study is structured in such a way that the introduction is followed by a theoretical framework, analysis, conclusion, and references.

2. THEORETICAL BACKGROUND

The positivist paradigm prevails in science starting from the assumption of a single comprehensible reality of independent facts for which quantitative measurements are made (Tsoukas, 1989). In other words, scientists study the world through a one-way mirror (Guba & Lincoln, 1994). Such an approach is inappropriate especially in the context of the social sciences because research involves people and their real-life experiences (Healy & Perry, 2000). This research paradigm seeks to investigate, validate, and predict patterns of behavior and is commonly used in research to test a theory or hypothesis (Taylor & Medina, 2011). It is strictly focused on the objectivity of the process (Creswell, 2012) which is the main reason why the researcher of this ideology is only the controller of the research process outside the research site (Taylor & Medina, 2011). If we look at the assumptions of the paradigm, we can say that ontology represents realism, which tells us that society is predictable. So, it is possible to test and measure human behavior (Taylor & Medina, 2011). Positivists skillfully avoid theorizing about the topic they are researching but are already focused on measuring instrumental variables (Lougen, 2009). On the other hand, epistemology represents an objective relationship between researchers and respondents where the researcher cannot and does not influence the obtained results (Ponterotto, 2005). The methodological approach used by positivists is quantitative (deductive) research that uses mathematical-statistical methods to test hypotheses (Guba & Lincoln, 1994). What distinguishes positivists from other researchers is the application of quantitative research methods, while other ideologies have mostly used a qualitative approach. Likewise, the fundamental discrepancy between positivism and other paradigms lies in the fact that positivists separate themselves from the world they study, while researchers from other paradigms recognize that to some extent, they must participate in real-world life to better understand and express its apparent features and characteristics (Tsoukas, 1989).

The opposite of the positivist paradigm is the constructivist (interpretive) paradigm, in which researchers change their approach to research. They turn to the ideologies and values behind the findings, so they believe that reality actually consists of the relativistic multiple complex

realities that people have in their minds (Healy & Perry, 2000). So, there is not just one real truth. Exploring this constructed reality depends on the interactions between interviewers and respondents. Specifically, the researcher must be a passionate participant during his/her field-work (Guba & Lincoln, 1994). Ontologically speaking, there is no access to one real world, but the goal is to step into people's skin, that is, to see the world through their eyes (Josselson, 2007). Thus, the epistemology of this paradigm is inter-subjectivism based on understanding and knowing about the other through an extended process of interaction such as interviewing and observing participants with the aim of presenting them as credibly and authentically as possible (Healy & Perry, 2000). An interpretive researcher asks himself questions such as (Taylor & Medina, 2011): (1) What is the impact of my own (past and present) values and beliefs in interpreting the thoughts and feelings of others? and (2) What hidden assumptions are limiting the way we understand others? Finally, from a methodological point of view, various inductive interpretive research methods are applied, such as narrative research, observation, interview, and autobiographical and autoethnographic methods (Ellis & Bochner, 2000; Richardson, 2000; Taylor & Settlemarier, 2003).

On the other hand, critical theory is at odds with traditional theory, which explores and confirms the status quo (Asghar, 2013), while critical theory challenges it and strives for a balanced and democratic society. Unlike the positivist paradigm, proponents of the critical paradigm assume that social science can never be truly objective or worthless and believe that scientific research should be conducted with the explicit goal of social change as the ultimate priority (Scotland, 2012). Critical paradigm researchers advocate system bias, and this bias is particularly visible in a negative context through the social position of women and marginalized ethnic groups (Scotland, 2012). Also, their research projects aim to encourage positive change in the research participants and the systems being studied, as well as in the collection of important data (De-Carlo, 2018). The critical paradigm not only studies power imbalances but also seeks to change them (DeCarlo, 2018). Critical researchers are not particularly tied to either method, so they use qualitative, quantitative, or a combination of both (Asghar, 2013). The ontological position of the critical paradigm is historical realism, where reality is considered to be shaped by social, political, cultural, economic, ethnic, and gender values. Realities are socially constructed entities that are under the constant internal influence (Scotland, 2012). Accordingly, epistemology is subjectivism based on real-world phenomena and associated with social ideologies. Knowledge is both socially constructed and influenced by power relations within society, more specifically what is considered knowledge is determined by the social and positional power of the proponents of that knowledge (Cohen et al., 2017).

Postmodernism is a paradigm that questions almost every mode of cognition, which many social scientists take for granted (Ghasarian et al., 1996). The postmodern has inherent problems with previous paradigms because the truth is always bound within a historical and cultural context, where there are no universally true explanations (DeCarlo, 2018). Absolute truths, identities, and major human values are criticized. Postmodernism can be defined as a comprehensive paradigm, which refers to a set of assumptions about ontology (realities are created), epistemology (knowledge is fluid and temporary), methodology (interpretive and critical methods are more appropriate to study plural society), and axiology (study of values: no set of values is by definition better than another) (Kroeze, 2012). We can define this scientific paradigm as a widespread and deeply incised cultural movement that changes people's perception of the existence and knowledge of the whole world (Watson, 2012). The combination of philosophy, art policy with changes in the economy, and technological changes are one of the most important features of postmodernism, and it was brought to the fore by feminist movements (Bačić, 2013). In the era of postmodernism, cultural changes are emphasized, where new forms of production are present, such as 'lean production', 'concept team', and 'just-in-time' production with the use of modern information technologies (Wood, 1997). The presence of skepticism towards certain-ty and great explanations in the social sciences is characteristic (DeCarlo, 2018).

The last paradigm, transmodernism, is structurally opposed to modernism and postmodernism, where it criticizes them on the one hand and takes the best of them on the other. Thus, it represents a synthesis of modernist (positivism, constructivism, and criticism) and postmodernist paradigms. It is completely future-oriented with the goal of moral liberation of all mankind with the establishment of moral liberal norms, and pragmatic and pluralistic principles (Cojocaru & Sandu, 2011). This paradigm has brought a shift in the value system, and new reflections in global consciousness, which from a historical point of view has not been experienced since the Renaissance. If we look at the assumptions of the transmodernist paradigm, it is possible to notice that the ontology is based on the equality of materials, and experience in creating new visions, where the center of truth is open to all with the existence of the so-called authentic "I". On the other hand, the epistemology of transmodernism says that current knowledge creates new knowledge, reconstructing reality in order to achieve idealism (Ivanović, 2015). This epoch addresses four core values (Pritchard et al., 2011): gender equality, global culture and ethnic equality, sustainability and survival of humanity, individuality, globalism, and interconnectedness. Transmodernism gives us the necessary political and epistemological position to overcome contradictions and treatments of diversity, providing theorizing that can provide a foundation for coexistence and existence without domination over each other (Ateljević, 2013).

3. METHODOLOGY

Relevant scientific literature was searched on the online databases Google Scholar and Emerald insight by keywords: "gastronomic experience", "restaurant experience" and "food experience". The following elimination criteria were used in the selection of publications: (a) papers written in English, (b) papers published only in the period from 2008 to 2021, and (c) papers examining the tourist's gastronomic experience. In addition to respecting the elimination criteria, 59 papers were downloaded from the aforementioned online databases into Mendeley software (available for free at: https://mendeley.en.softonic.com/) for reference management. In the further process of selecting papers in the Mendeley software, all abstracts of the downloaded papers were read, in order to determine the correspondence with the researched problems. Out of a total of 59 papers taken, 55 were selected for the final sample. After the sample was established, data on its publication year, geographical scope, and research type were collected for each paper, which was later analyzed by descriptive statistical methods in Excel. In the analysis of papers according to the appropriate paradigm, content analysis was used as a scientific research method (Tkalac Verčić et al., 2010). After pragmatic selection, the papers are presented in tabular absolute and relative frequencies.

4. **RESEARCH RESULTS**

For the purposes of this research, descriptive statistics methods were applied, based on which results were obtained for paradigmatic analysis (Table 1), frequency analysis by year (Figure 1), geographical analysis (Figure 2), and analysis by research type (Table 2).

| Type of paradigms | Source | f | % |
|-------------------|---|----|--------|
| (Post) positivism | (Bekar, 2017; Björk & Kauppinen-Räisänen, 2015; Campbell & DiPietro, 2014; Canny, 2014; Cao et al., 2019; Chen et al., 2021; Githiri, 2016; Gracia et al., 2011; Ha & Jang, 2012; Ha & (Shawn) Jang, 2010; Hanks & Line, 2018; Horng et al., 2013; Horng & Hsu, 2021; Hyun & Han, 2012; Kiatkawsin & Han, 2019; Kim & Moon, 2009; Lee et al., 2015; Lee et al., 2016; Lee, 2015; Liao & Fang, 2019; Liu & Tse, 2018; Liu & Jang, 2009; Mahalingam et al., 2016; Marković et al., 2021c; Namkung & Jang, 2010; Ouyang et al., 2018; Richardson et al., 2019; Ryu et al., 2012; Ryu & Shawn Jang, 2008; Sipe & Testa, 2018; Sthapit et al., 2019; Sulaiman & Haron, 2013; Sunghyup Sean Hyun, 2010; Tang & Bougoure, 2011; Wardono et al., 2012; Wu & Liang, 2009; Yaris, 2019; Yoon & Chung, 2018; Zhang et al., 2019) | 39 | 70,91% |
| Constructivism | (Marković et al., 2021a, 2021b; Park et al., 2021; Sthapit, 2017; Stone et al., 2018; Tsaur & Lo, 2020) | 6 | 10,91% |
| Criticism | (Ding & Lee, 2017; Horng & Hsu, 2020; Hussein, 2018; Kala, 2020; Moon et al., 2020; Paakki et al., 2019; Ryu et al., 2008; Tsaur et al., 2015; Wen et al., 2020; Wu & Mohi, 2015) | 10 | 18,18% |
| Postmodern | - | - | - |
| Transmodern | - | - | - |
| In total | <u> </u> | 55 | 100% |

| | D | 1 | . • 1 | 1 . | | | • | • |
|----------|---------|-----------|---------|----------|----|------------|----------------|----------|
| Table I | . Para | digmatic | article | analysis | on | tourist s | gastronomic ex | nerience |
| I HOIV I | • 1 414 | anginatio | artiere | anaryono | on | tour ibt b | Subtronomic en | |

Source: Authors' research

With a paradigmatic analysis of 55 papers, it was determined that almost three-quarters of the papers are (post) positivist (70.91%). The authors in these post (positivist) articles turn to realism, an objectivist view, applying a quantitative deductive research approach. On the other hand, induction was used in the articles of the constructivist view which is represented in the sample with 10.91%. Constructivist research is relativistic and subjectively oriented, nurturing a qualitative research approach. Furthermore, although the critical paradigm is focused on power relations in society, no research of such applicative value has been present in this study. However, in 18.18% of the research, a mix of inductive and deductive research approaches was used, which seeks to see a broader picture in examining the perception of gastro tourists. For this reason, these 10 studies are categorized into a critical realism worldview. Postmodernism and transmodernism were not represented in any of the studies in the sample. In order to understand the research topic in more detail, several more analyzes were presented, and the analysis of the article by year follows below (Figure 1). Since the paradigmatic analysis established that the study of postmodernism and transmodernism was not represented in the sample, further analyzes will be performed exclusively for the modern period.



Articles analysis by year, it was determined that the largest number of studies of (post) positivism was published in 2019 (12.73%) and 2018 (9.09%). Constructivism studies are represented in only four years (2017, 2018, 2020, 2021), of which the largest number of studies were published in 2021 (5.45%). Criticism stands out the most in 2020 when 7.27% of studies on the tourists' gastronomic experience were published. The following is a geographical article analysis according to the paradigm to which they belong.





The geographical analysis of articles on the tourist's gastronomic experience was conducted according to the types of research paradigms - positivism, constructivism, and criticism. The analysis found that the largest number of positivist authors are from North America, specified from the USA (29.09%). This is followed by positivist authors from Asia (25.45%), and from countries such as Taiwan, Korea, Indonesia, Vietnam, India, Japan, and Malaysia. In Europe, a total of 10.91% of positivist research were conducted in Turkey, Finland, Spain, and Croatia. In Kenya on the African continent, 1.82% of research was conducted, and the same amount of research was conducted in New Zealand. An unknown location of the study was present only in the Sulaiman & Haron study (2013). On the other hand, as far as constructivist authors are concerned, the largest number of studies was conducted in Europe (5.45%), more precisely in Croatia and Finland, while only one study was conducted in Taiwan and the USA. An in-depth interview at the international level in an online environment was conducted by the constructivists Stone et al (2018). The critical approach was most applied in Asia in the following countries: Taiwan, Indonesia; China, India, and Korea. Asia is followed by critical research in North America (3.45%), while only 1.82% of studies have been published in Europe, Australia, and Africa. The graph also shows that the summation of the obtained values by continents for all three paradigms is led by Asia with 40%, followed by North America with 34.55% and Europe with 18.18%. After the conducted geographical analysis, the following is an article analysis on the tourist's gastronomic experience according to the research type.

| approach | Source | f | % |
|------------------------|--|---|--------|
| | Positivism paradigm | | |
| Quantitative approach | | | |
| Survey research | (Bekar, 2017; Björk & Kauppinen-Räisänen, 2015; Campbell & DiPietro, 2014; Canny, 2014; Cao et al., 2019; Chen et al., 2021; Githiri, 2016; Gracia et al., 2011; Ha & Jang, 2012; Ha & (Shawn) Jang, 2010; Hanks & Line, 2018; Horng et al., 2013; Horng & Hsu, 2021; Hyun & Han, 2012; Kiatkawsin & Han, 2019; Kim & Moon, 2009; Lee et al., 2015; Lee et al., 2016; Lee, 2015; Liao & Fang, 2019; Liu & Tse, 2018; Liu & Jang, 2009; Mahalingam et al., 2016; Marković et al., 2021c; Namkung & Jang, 2010; Ouyang et al., 2018; Richardson et al., 2019; Ryu et al., 2012; Ryu & Shawn Jang, 2008; Sipe & Testa, 2018; Sthapit et al., 2019; Sulaiman & Haron, 2013; Sunghyup Sean Hyun, 2010; Tang & Bougoure, 2011; Wu & Liang, 2009; Yaris, 2019; Yoon & Chung, 2018; Zhang et al., 2019) | | 69,09% |
| Experimental research | (Wardono et al., 2012) | 1 | 1,82% |
| | Constructivism paradigm | | |
| Qualitative approach | | | |
| In-depth interview | (Sthapit, 2017; Stone et al., 2018; Tsaur & Lo, 2020) | 3 | 5,45% |
| Content analysis | (Marković et al., 2021a, 2021b; Park et al., 2021) | 3 | 5,45% |
| | Criticism paradigm | | |
| Mixed approach | | | |
| In-depth interview and | (Ding & Lee, 2017; Horng & Hsu, 2020; Moon et al., 2020; Paakki et al., 2019; Tsaur et al., 2015; Wen et al., 2020) | 6 | 10,91% |
| survey | | | |

Table 2. Article analysis by research type

Source: Authors' research

The paradigmatic analysis by research type is shown in the previous table. The obtained results of the analysis indicate that only one of the positivist researchers used the experimental quantitative approach (1.82%), while in the remaining studies, the survey was used (67.27%). Of the constructivist researchers, 5.45% conducted an in-depth interview, while the same number applied content analysis (5.45%). A mixed research approach was observed in 18.18% of studies on the tourist's gastronomic experience, led by a mix of tourist's in-depth interviews and surveys (10.91%), followed by a mix of focus groups and surveys (7.27%). Thus, from the methodological point of view, the quantitative approach, more precisely the examination of tourists using the questionnaire as a measuring instrument, definitely prevails in the literature on the gastronomic experience of tourists.

5. CONCLUSION

Type of research

Understanding different types of paradigms is extremely important for scientific experts, especially in the context of research terms - ontology, epistemology, and methodology. It is the knowledge of ontology and epistemology that directly influences the choice of methodological approach to research. Considering the importance of gastronomy in the development of tourist destinations, gastronomic tourism was chosen as the topic of this research. Therefore, for the purposes of this research, 55 scientific articles on the gastronomic experience of tourists belonging to different types of paradigms were analyzed. A detailed analysis found that each of the authors of selected scientific papers has its own research position that serves as a basis for categorizing the type of paradigm to which it belongs. The results of this research indicate that most of the papers are represented from the positivist, followed by a critical and constructivist scientific research point of view. No research belonging to the postmodern and transmodern was represented in the sample, so further analyzes were conducted only on the example of positivism, constructivism, and criticism. Although this study includes studies in the period from 2008 to 2021, the results of frequency analysis by year indicate that the largest number of studies on the observed topic was published in the last three years. The geographical analysis found that the largest number of studies was conducted in Asia and North America, where the authors mainly apply a quantitative research approach.

Despite the fact that this review provided insight into the current state of representation of paradigms in research on the tourist's gastronomic experience, it also has certain limitations. Given that only papers published in English in the period from 2008 to 2021 have been considered in this literature review, future research should include articles in other languages in the sample, covering a longer period of time. In order to achieve greater representativeness of the results, the proposal for future studies is to use other relevant databases in finding papers, in addition to Google Scholar and Emerald Insight. Furthermore, to gain a more detailed insight into previous research on the tourist's gastronomic experience, it would be desirable to apply a bibliographic analysis according to the journal in the future. Definitely, the biggest limitation in this research is the lack of studies belonging to the postmodern and transmodern paradigm, so it is suggested that researchers direct their future research precisely in the direction of these two research ideologies. The contribution of this research is extremely important for academic experts because based on the analyzed data, one can gain insight into the current paradigmatic structure of articles on the tourist's gastronomic experience. It is different research approaches that can deepen knowledge and insights into the needs of tourists, and the data obtained from different perspectives can serve the management of gastronomic products and services in the long-term improvement of current elements of the gastronomic offer.

REFERENCES

- Asghar, J. (2013). Critical paradigm: A preamble for novice researchers. *Life Science Journal*, *10*(4), 3121–3127.
- Ateljević, I. (2013). Visions of Transmodernity: A New Renaissance of our Human History? *Integral Review*, 9(2), 200-220.
- Bačić, A. (2013). Konstitucionalizam, historizam i postmodernizam. Zbornik radova Pravnog fakulteta u Splitu, 50 (1), 1-15.
- Bekar, A. (2017). The effects of aesthetic value in food and beverage businesses on the aesthetic experiences and revisit intentions of customers. *The Journal of Academic Social Science Studies*, 1(54), 373–388. https://doi.org/10.9761/JASSS6915
- Björk, P., & Kauppinen-Räisänen, H. (2015). Contemporary insights to the dynamic pre-trip information sourcing behaviour. *Tourism and Hospitality Research*, 15(1), 39–53. https://doi. org/10.1177/1467358414553871
- Campbell, J. M., & DiPietro, R. B. (2014). Sign of the times: Testing consumer response to local food signage within a casual dining restaurant. *Journal of Retailing and Consumer Services*, 21(5), 812–823. https://doi.org/10.1016/j.jretconser.2014.06.010
- Canny, I. (2014). Measuring the Mediating Role of Dining Experience Attributes on Customer Satisfaction and Its Impact on Behavioral Intentions of Casual Dining Restaurant in Jakarta. *International Journal of Innovation, Management and Technology*, 5(1), 25-43. https:// doi.org/10.7763/IJIMT.2014.V5.480
- Cao, Y., Li, X. (Robert), DiPietro, R., & So, K. K. F. (2019). The creation of memorable dining experiences: formative index construction. *International Journal of Hospitality Management*, 82, 308–317. https://doi.org/10.1016/j.ijhm.2018.10.010

- Chen, L.-H., Wang, M.-J. S., & Morrison, A. M. (2021). Extending the memorable tourism experience model: a study of coffee tourism in Vietnam. *British Food Journal*, 123(6), 2235–2257. https://doi.org/10.1108/BFJ-08-2020-0748
- Cohen, L., Manion, L., & Morrison, K. (2017). *Research Methods in Education*. Research Methods in Education. London: Routledge. https://doi.org/10.4324/9781315456539
- Cojocaru, D., & Sandu, A. (2011). (BIO)ethical and social reconstructions in transmodernity. *Journal for the Study of Religions and Ideologies*, *10*(30), 258–276.
- Correia, A., Kim, S. (Sam), & Kozak, M. (2020). Gastronomy experiential traits and their effects on intentions for recommendation: A fuzzy set approach. *International Journal of Tourism Research*, *22*(3), 351–363. https://doi.org/10.1002/jtr.2340
- Creswell, J. W. (2012). *Educational research: Planning, conducting, and evaluating quantitative and qualitative research.* Educational Research. Boston: Pearson Education, Inc.
- Crotts, J. C., & Magnini, V. P. (2011). The customer delight construct. Annals of Tourism Research, 38(2), 719–722. https://doi.org/10.1016/j.annals.2010.03.004
- DeCarlo, M. (2018). *Scientific Inquiry in Social Work*. Open Social Work Education. https://scientificinquiryinsocialwork.pressbooks.com/
- Deshpande, R. (1983). "Paradigms Lost": On Theory and Method in Research in Marketing. Journal of Marketing, 47(4), 101-110. https://doi.org/10.2307/1251403
- Ding, D., & Lee, H. M. (2017). A measurement scale for food festival visitor experience. *International Journal of Tourism Sciences*, 17(3), 180-197. https://doi.org/10.1080/15980634.20 17.1349069
- Ellis, C., & Bochner, A. (2000). Autoethnography, Personal Narrative, Reflexivity: Researcher as Subject. *Handbook of Qualitative Research*, 28, 32-45.
- Ghasarian, C., Best, S., & Kellner, D. (1996). Postmodern Theory. Critical Interrogations. *Revue Française de Sociologie*, *37*(1), 151. https://doi.org/10.2307/3321956
- Githiri, M. N. (2016). Influence of Physical Environment on Customer Satisfaction and Return Intention in Kenyan Rated Restaurants. *Asian Journal of Social Science Studies*, 2(1), 11-18. https://doi.org/10.20849/ajsss.v2i1.82
- Gracia, E., Bakker, A. B., & Grau, R. M. (2011). Positive Emotions: the connection between customer quality evaluations and loyalty. *Cornell Hospitality Quarterly*, 52(4), 458–465. https://doi.org/10.1177/1938965510395379
- Guba, E. G. (1990). The alternative paradigm dialog, in E. Guba (ed.), *The Paradigm Dialog*, (pp. 17–30). Newbury Park, CA: Sage
- Guba, E. G., & Lincoln, Y. S. (1994). Competing paradigms in qualitative research. In N. K. Denzin, & Y. S. Lincoln (Eds.), *Handbook of qualitative research* (pp. 105-117). Thousand Oaks: Sage Publications.
- Ha, J., & Jang, S. (Shawn). (2012). The effects of dining atmospherics on behavioral intentions through quality perception. *Journal of Services Marketing*, 26(3), 204–215. https:// doi.org/10.1108/08876041211224004
- Ha, J., & (Shawn) Jang, S. (2010). Perceived values, satisfaction, and behavioral intentions: The role of familiarity in Korean restaurants. *International Journal of Hospitality Management*, 29(1), 2–13. https://doi.org/10.1016/j.ijhm.2009.03.009
- Hanks, L., & Line, N. D. (2018). The restaurant social servicescape: Establishing a nomological framework. *International Journal of Hospitality Management*, 74, 13–21. https://doi. org/10.1016/j.ijhm.2018.01.022
- Healy, M., & Perry, C. (2000). Comprehensive criteria to judge validity and reliability of qualitative research within the realism paradigm. *Qualitative Market Research: An International Journal*, 3(3),118 - 126. https://doi.org/10.1108/13522750010333861

- Horng, J.-S., Chou, S.-F., Liu, C.-H., & Tsai, C.-Y. (2013). Creativity, aesthetics and eco-friendliness: A physical dining environment design synthetic assessment model of innovative restaurants. *Tourism Management*, 36, 15–25. https://doi.org/10.1016/j.tourman.2012.11.002
- Horng, J.-S., & Hsu, H. (2020). A holistic aesthetic experience model: Creating a harmonious dining environment to increase customers' perceived pleasure. *Journal of Hospitality and Tourism Management*, 45, 520–534. https://doi.org/10.1016/j.jhtm.2020.10.006
- Horng, J.-S., & Hsu, H. (2021). Esthetic Dining Experience: The relations among aesthetic stimulation, pleasantness, memorable experience, and behavioral intentions. *Journal of Hospitali*ty Marketing & Management, 30(4), 419–437. https://doi.org/10.1080/19368623.2021.1859425
- Hussein, A. S. (2018). Revisiting the importance of casual dining experience quality: an empirical study. *International Journal of Quality and Service Sciences*, 10(3), 233–252. https://doi.org/10.1108/IJQSS-04-2017-0041
- Hyun, S. S., & Han, H. (2012). A model of a patron's innovativeness formation toward a chain restaurant brand. *International Journal of Contemporary Hospitality Management*, 24(2), 175–199.
- Josselson, R. (2007). The ethical attitude in narrative research. In Clandinin, D.J. (2007) *Handbook of Narrative Inquiry* (pp. 537-566). Thousand Oaks, Ca: Sage Publications.
- Kala, D. (2020). Dimensions of a memorable food and restaurant experience in mountainous destinations: an interpretive structural modeling approach. *Enlightening tourism. A pathmaking journal*, 10(2), 206. https://doi.org/10.33776/et.v10i2.5111
- Kiatkawsin, K., & Han, H. (2019). What drives customers' willingness to pay price premiums for luxury gastronomic experiences at Michelin-starred restaurants? *International Journal* of Hospitality Management, 82, 209–219. https://doi.org/10.1016/j.ijhm.2019.04.024
- Kim, W. G., & Moon, Y. J. (2009). Customers' cognitive, emotional, and actionable response to the servicescape: A test of the moderating effect of the restaurant type. *International Journal of Hospitality Management*, 28(1), 144–156. https://doi.org/10.1016/j.ijhm.2008.06.010
- Kroeze, J.H. (2012). Interpretivism in IS a postmodernist (or postpositivist?) knowledge theory. Proceedings of the 18th Americas Conference on Information Systems (AMCIS 2012 Proceedings), Seattle, Washington, USA (pp. 9-11), 2012, Paper 7, ISBN 978-0-615-66346-3. Available: http://aisel.aisnet.org/amcis2012/proceedings/PerspectivesIS/7
- Lee, C. J., Wang, Y. C., & Cai, D. C. (2015). Physical Factors to Evaluate the Servicescape of Theme Restaurants. *Journal of Asian Architecture and Building Engineering*, 14(1), 97– 104. https://doi.org/10.3130/jaabe.14.97
- Lee, L., Lee, M. J., & Dewald, B. (2016). Measuring the customers' perception of tangible service quality in the restaurant industry: An emphasis on the upscale dining segment. *Journal of Foodservice Business Research*, 19(1), 21–38. https://doi.org/10.1080/15378020.201 6.1129216
- Lee, Y. J. (2015). Creating memorable experiences in a reuse heritage site. *Annals of Tourism Research*, 55, 155-170. https://doi.org/10.1016/j.annals.2015.09.009
- Liao, W.-L., & Fang, C.-Y. (2019). Applying an Extended Theory of Planned Behavior for Sustaining a Landscape Restaurant. Sustainability, 11(18), 5100. https://doi.org/10.3390/ su11185100
- Liu, P., & Tse, E. C.-Y. (2018). Exploring factors on customers' restaurant choice: an analysis of restaurant attributes. *British Food Journal*, 120(10), 2289–2303. https://doi.org/10.1108/ BFJ-10-2017-0561
- Liu, Y., & Jang, S. (Shawn). (2009). The effects of dining atmospherics: An extended Mehrabian–Russell model. *International Journal of Hospitality Management*, 28(4), 494–503. https://doi.org/10.1016/j.ijhm.2009.01.002

- Lougen, C. (2009). Sources: The Sage Encyclopedia of Qualitative Research Methods. *Reference & User Services Quarterly*, 49(1),112-123. https://doi.org/10.5860/rusq.49n1.101
- Mahalingam, S., Jain, B., & Sahay, M. (2016). Role of physical environment (dinescape factors) influencing customers' revisiting intention to restaurants. 2016 International Conference on Advances in Computing, Communications and Informatics (ICACCI), 1069–1072. https:// doi.org/10.1109/ICACCI.2016.7732186
- Marković, S., Dorčić, J., Rašan, D., Bucić, B., & Blažić, M. (2021a). Aesthetic guest experience in restaurant: a state-of-the-art review. In V. Bevanda (Ed.), *Economics & Management: How to Cope with Disrupted Times Selected Papers*. Paper presented at 5th International Scientific Conference EMAN 2021, Ljubljana, Slovenia (hybrid), March 18, 2021 (pp. 135 148). Belgrade: Association of Economists and Managers of the Balkans. https://doi.org/10.31410/EMAN.S.P.2021.135
- Marković, S., Dorčić, J., Rašan, D., Bucić, B., & Blažić, M. (2021b). Content analysis of Michelin restaurants online reviews in Croatia. In V. Bevanda (Ed.), *Economics & Management: How* to Cope with Disrupted Times. Paper presented at 5th International Scientific Conference EMAN 2021, Ljubljana, Slovenia (hybrid), March 18, 2021 (pp. 381 - 392). Belgrade: Association of Economists and Managers of the Balkans. https://doi.org/10.31410/EMAN.2021.381
- Marković, S., Dorčić, J., Rašan, D., Bucić, B., & Blažić, M. (2021c). The Importance of Aesthetic Experience in Fine Dining Restaurants. In V. Bevanda (Ed.), *Knowledge based sustainable development - Selected Papers*. Paper presented at 7th International Scientific Conference ERAZ 2021, Prague – Czech Republic (hybrid – online, virtually and in person), May 27, 2021 (pp. 147 - 154). Belgrade: Association of Economists and Managers of the Balkans. https://doi.org/10.31410/ERAZ.S.P.2021.147
- Moon, S., Bonn, M. A., & Cho, M. (2020). How can the solo dining experience be enhanced? Focusing on perceived territoriality. *International Journal of Hospitality Management*, 88, 102506, 1-10. https://doi.org/10.1016/j.ijhm.2020.102506
- Namkung, Y., & Jang, S. C. (Shawn). (2010). Effects of perceived service fairness on emotions, and behavioral intentions in restaurants. *European Journal of Marketing*, 44(9/10), 1233– 1259. https://doi.org/10.1108/03090561011062826
- Ouyang, Y., Behnke, C., Almanza, B., & Ghiselli, R. (2018). The Influence of Food Aromas on Restaurant Consumer Emotions, Perceptions, and Purchases. *Journal of Hospitality Marketing and Management*, 27(4), 405–423. https://doi.org/10.1080/19368623.2017.1374225
- Paakki, M., Aaltojärvi, I., Sandell, M., & Hopia, A. (2019). The importance of the visual aesthetics of colours in food at a workday lunch. *International Journal of Gastronomy and Food Science*, *16*, 100131, 1 36. https://doi.org/10.1016/j.ijgfs.2018.12.001
- Park, E., Kwon, J., Chae, B. (Kevin), & Kim, S.-B. (2021). What Are the Salient and Memorable Green-Restaurant Attributes? Capturing Customer Perceptions from User-Generated Content. SAGE Open, 11(3), 215824402110315, 1 – 13. https://doi.org/10.1177/21582440211031546
- Ponterotto, J. G. (2005). Qualitative research in counseling psychology: A primer on research paradigms and philosophy of science. In *Journal of Counseling Psychology*, 52(2), 126.-136. https://doi.org/10.1037/0022-0167.52.2.126
- Pritchard, A., Morgan, N., & Ateljevic, I. (2011). Hopeful tourism. A New Transformative Perspective. *Annals of Tourism Research*, *38*(3), 941–963. https://doi.org/10.1016/j.annals.2011.01.004
- Richardson, L., & St Pierre, E. (2000). *A method of inquiry*. Handbook of qualitative research, 923-948.
- Richardson, S., Lefrid, M., Jahani, S., Munyon, M. D., & Rasoolimanesh, S. M. (2019). Effect of dining experience on future intention in quick service restaurants. *British Food Journal*, *121*(11), 2620–2636. https://doi.org/10.1108/BFJ-09-2018-0617

- Ryu, K., Han, H., & Kim, T.-H. (2008). The relationships among overall quick-casual restaurant image, perceived value, customer satisfaction, and behavioral intentions. *International Journal of Hospitality Management*, 27(3), 459–469. https://doi.org/10.1016/j.ijhm.2007.11.001
- Ryu, K., Lee, H., & Gon Kim, W. (2012). The influence of the quality of the physical environment, food, and service on restaurant image, customer perceived value, customer satisfaction, and behavioral intentions. *International Journal of Contemporary Hospitality Management*, 24(2), 200–223. https://doi.org/10.1108/09596111211206141
- Ryu, K., & Shawn Jang, S. (2008). DINESCAPE: A scale for customers' perception of dining environments. *Journal of Foodservice Business Research*, 11(1), 2-22. https://doi. org/10.1080/15378020801926551
- Sarantakos, S. (2005). Social Research. (3rd ed.). In Macmillan Education.
- Scotland, J. (2012). Exploring the philosophical underpinnings of research: Relating ontology and epistemology to the methodology and methods of the scientific, interpretive, and critical research paradigms. *English Language Teaching*, 5(9), 9–16. https://doi.org/10.5539/elt.v5n9p9
- Sipe, L. J., & Testa, M. R. (2018). From Satisfied to Memorable: An Empirical Study of Service and Experience Dimensions on Guest Outcomes in the Hospitality Industry. *Journal of Hospitality Marketing & Management*, 27(2), 178–195. https://doi.org/10.1080/19368623.2017.13 06820
- Sthapit, E. (2017). Exploring tourists' memorable food experiences: a study of visitors to Santa's official hometown. *Anatolia*, 28(3), 404–421. https://doi.org/10.1080/13032917.2017.1328607
- Sthapit, E., Coudounaris, D. N., & Björk, P. (2019). Extending the memorable tourism experience construct: an investigation of memories of local food experiences. *Scandinavian Journal of Hospitality and Tourism*, 19(4–5), 333–353. https://doi.org/10.1080/15022250.2019.1689530
- Stone, M. J., Soulard, J., Migacz, S., & Wolf, E. (2018). Elements of Memorable Food, Drink, and Culinary Tourism Experiences. *Journal of Travel Research*, 57(8), 1121–1132. https://doi. org/10.1177/0047287517729758
- Sulaiman, S., & Haron, M. S. (2013). Foodscape and Customer's Future Behavioral Intentions in Casual Dining Restaurant. *Journal of Economics, Business and Management*, 94–97. https:// doi.org/10.7763/JOEBM.2013.V1.22
- Sunghyup Sean Hyun. (2010). Predictors of Relationship Quality and Loyalty in the Chain Restaurant Industry. *Cornell Hospitality Quarterly*, 51(2), 251–267. https://doi.org/10.1177/1938965510363264
- Tang, K. M., & Bougoure, U. (2011). Service quality: an investigation into Malaysia customers using DINESERV. *Journal of Services Marketing*, 12(6), 426–440.
- Taylor, P. C., & Medina, M. (2011). Educational Research Paradigms: From Positivism to Pluralism. College Research Journal, 1(1), 1-16.
- Taylor, P., & Settlemaier, E. (2003). Critical autobiographical research for science educators. Critical autobiographical research for science educators. Critical autobiographical research for science educators, 27(4), 233.-244. https://doi.org/10.14935/jssej.27.233
- Tkalac Verčić, A., Sinčić Čorić, D. & Pološki Vokić, N. (2010). Priručnik za metodologiju istraživačkog rada: Kako osmisliti, provesti i opisati znanstveno i stručno istraživanje. M.E.P.d.o.o.
- Tsaur, S.-H., & Lo, P.-C. (2020). Measuring memorable dining experiences and related emotions in fine dining restaurants. *Journal of Hospitality Marketing & Management*, 29(8), 887–910. https://doi.org/10.1080/19368623.2020.1748157
- Tsaur, S.-H., Luoh, H.-F., & Syue, S.-S. (2015). Positive emotions and behavioral intentions of customers in full-service restaurants: Does aesthetic labor matter? *International Journal of Hospitality Management*, 51, 115–126. https://doi.org/10.1016/j.ijhm.2015.08.015

- Tsoukas, H. (1989). The Validity of Idiographic Research Explanations. *Academy of Management Review*, *14*(4), 165 -181. https://doi.org/10.5465/amr.1989.4308386
- Wardono, P., Hibino, H., & Koyama, S. (2012). Effects of Interior Colors, Lighting and Decors on Perceived Sociability, Emotion and Behavior Related to Social Dining. *Procedia - Social and Behavioral Sciences*, 38, 362–372. https://doi.org/10.1016/j.sbspro.2012.03.358
- Watson, N. (2012). Postmodernism and lifestyles. *The Routledge Companion to Postmodernism, Third Edition* (pp. 62–72). Taylor and Francis.
- Wen, H., Leung, X., & Pongtornphurt, Y. (2020). Exploring the impact of background music on customers' perceptions of ethnic restaurants: The moderating role of dining companions. *Journal of Hospitality and Tourism Management*, 43, 71–79. https://doi.org/10.1016/j. jhtm.2020.02.007
- Wood, E. M. (1997). Modernity, postmodernity or capitalism? *Review of International Political Economy*, 4(3), 539–560. https://doi.org/10.1080/096922997347742
- Wu, C. H.-J., & Liang, R.-D. (2009). Effect of experiential value on customer satisfaction with service encounters in luxury-hotel restaurants. *International Journal of Hospitality Management*, 28(4), 586–593. https://doi.org/10.1016/j.ijhm.2009.03.008
- Wu, H. C., & Mohi, Z. (2015). Assessment of Service Quality in the Fast-Food Restaurant. Journal of Foodservice Business Research, 18(4), 358-388. https://doi.org/10.1080/15378020.201 5.1068673
- Yaris, A. (2019). Developing and validating a gastronomy experience scale. In O. Emir (Ed.), *City Tourism*. Paper presented at 4th International Tourism Congress 2019, October 16-19 (pp. 142-151). Eskişehir: Anadolu University Press.
- Yoon, B., & Chung, Y. (2018). Consumer Attitude and Visit Intention toward Food-Trucks: Targeting Millennials. *Journal of Foodservice Business Research*, 21(2), 187–199. https://doi.org/1 0.1080/15378020.2017.1368807
- Zhang, T., Chen, J., & Hu, B. (2019). Authenticity, Quality, and Loyalty: Local Food and Sustainable Tourism Experience. *Sustainability*, *11*(12), 1-18. https://doi.org/10.3390/su11123437